UNIVERSITY OF GRONINGEN

MASTER RESEARCH PROJECT

On Duality and Unity: Constructing Hybrid Soft Behaviour with the Double Copy and Transmutation Operators

Author: Jelmer Zijlstra

First Supervisor: Prof. Dr. D. Roest Second Supervisor: Prof. Dr. E. Bergshoeff

July, 2024



Abstract

Scattering amplitudes play a critical role in understanding fundamental interactions in quantum field theory. This thesis delves into the unifying relations for these amplitudes and applies the BCJ bootstrap method to construct kinematic numerators that respect the colour-kinematics duality. We began with a detailed analysis of Yang-Mills (gluon) and pion (NLSM) amplitudes, investigating their duality and behaviour under specific kinematic conditions. We then applied the BCJ bootstrap method to derive amplitudes for the gauged nonlinear sigma model, observing a hybrid soft degree where some amplitudes conform to $\sigma - 0$ while others exhibit pion contact terms with $\sigma = 1$.

In this work we have developed of a generalised transmutation (GT) operator, which allowed us to generate and analyze mixed amplitudes of scalars and pions interacting through gluons. The GT operator was validated by computing 6-point amplitudes and tracking Mandelstam variables, revealing correct results for Yang-Mills-Scalar (YMS) and Nonlinear Sigma Model (NLSM) amplitudes, and uncovering unexpected terms like the 4-point ϕ^4 amplitude.

We found that while the GT operator reconstructs some interactions observed in the BCJ bootstrap method, there are contributions to the $\sigma = 0$ amplitude that are not generated by the BCJ bootstrap. We therefore conclude that the BCJ bootstrap does not uniquely determine amplitudes based solely on the soft degree $\sigma = 0$.

Future work could examine how the GT operator acts on higher-derivative corrections in Yang-Mills theory in order to investigate their compatibility with CK-duality. We also suggest the possibility of applying GT to other theoretical frameworks, which could land on mixed theories such as Born-Infeld photons interacting with Maxwell theory photons through graviton exchange.

Contents

Introduction						
1	Scattering Amplitudes					
	1.1	Gluons - Yang-Mills Theory	12			
	1.2	Pions - Nonlinear Sigma Model	20			
	1.3	Factorisation - Recursion relations	23			
	1.4	Soft Theorems and Soft Factors	25			
	1.5	The Double Copy	32			
2	BC.	BCJ Bootstrap - Hybrid Soft Behaviour				
	2.1	A group Theory Approach to Numerators	40			
	2.2	Numerators for 4-point Interactions	45			
	2.3	Numerators for 5-point Interactions	48			
	2.4	Numerators for 6-point and Higher: BCJ Bootstrap	49			
	2.5	Colour x Kinematics: The Gauged NLSM	52			
	2.6	Key Findings and Implications	58			
3	Unifying Relations - Transmutation Operators					
	3.1	Conditions and Basis for Transmutation Operators	60			
	3.2	Trace and Insertion Operators	62			
	3.3	Longitudinal Operators	67			
	3.4	Web of theories connected through transmutation	71			
	3.5	Infrared properties of Transmutation	73			
	3.6	On-shell recursion, factorisation and the Double Copy	76			
	3.7	UV completion	78			
4	Generalised Transmutation of Yang-Mills					
	4.1	Defining the Generalised Transmutation Operator	81			
	4.2	Generalised transmutation at 4-point	84			
	4.3	Generalised Transmutation at 6-point	85			
5	Dis	cussion	89			

5]	Discu	ssion

	$5.1 \\ 5.2$	Partial 6-point Amplitude of BCJ BootstrapFactorization Comparison	89 92	
6	Con	clusion and Outlook	96	
Bi	bliog	raphy	99	
Appendix				
A	Dia	grammatic calculation of double color ordered BAS amplitudes	109	
в	KLI	f double copy	111	
С	Z-T	heory	113	
D	Tra	nsmutation operators	115	
	D.1	Kinematics	115	
	D.2	Gauge invariance	116	
	D.3	Dimensional Reduction: Pions from higher dimensional gluons	118	
	D.4	Gauge invariance of longitudinal operator	120	
	D.5	Transmutation of the subleading order soft factors	121	
\mathbf{E}	Mathematica computations of amplitudes		124	
	E.1	6-point YMS amplitude	124	

Acknowledgements

I would like to express my sincere gratitude to my supervisor, Diederik. First, for introducing me to the vast landscape of topics within the double copy framework. Although this thesis is extensive, the topics covered here represent only a tiny drop in the bucket of possible avenues to explore. Secondly, thank you for your enthusiasm, optimism, encouragement, and fun conversations over the past years.

A special 'thank you' also goes to Yang Li, to whom I could always turn with questions, even the seemingly dumb ones. Appreciation is also due to everyone from the Double Copy Journal Club (JCDC) and the Gravity/Cosmo journal club. I have truly enjoyed the interesting presentations and discussions during our Monday and Tuesday sessions.

Next, I would like to thank my office buddies Bram Alferink and Jelte Bottema. Working with you over the past year has been a lot of fun, and I have really enjoyed avoiding working on this thesis while drinking coffee and tea or grabbing snacks with you. Also, thanks to Cas Last for proofreading and spellchecking this thesis, even during his holiday.

Finally, I would like to thank Roos for always being supportive. Whenever this project got the better of me, you were there for me and I will always be grateful for that. I am excited to see what adventures are in store for us now that this project is finished.

Introduction

The study of quantum field theory has been an active area of research since the 1920s. One of the first fields to be successfully quantised was the electromagnetic field [1], which led to the development of the theory of quantum electrodynamics (QED). This theory was able to explain the existence of relativistic particles, namely photons, which appear as quantisations of the electromagnetic field. Such an explanation was not possible in the ordinary theory of quantum mechanics at the time.

With quantum field theory, a new method of calculating the *scattering matrix*, or S-matrix, of particle interactions was formulated. The S-matrix relates the initial and final state of particle processes, allowing for the calculation of interaction probabilities. The elements of this matrix are known as *scattering amplitudes*, which can be pertubatively computed using the famous Feynman diagram approach [2]. These diagrams are a diagrammatic representation of the scattering process, where each line and vertex can be associated to a mathematical expression. These associations are called the 'Feynman rules', and their specific form is based on the theory on is studying, as the Feynman rules can be derived from the interaction Lagrangian \mathcal{L}_{int} .

There are two types of theories that are fundamental to our understanding of particle interactions. The first is gauge theory, which is a core part of the Standard Model of particle physics. In this model, three of the four fundamental forces of the universe are described as interacting particles exchanging gauge bosons, which are referred to as the 'force carriers'. There are three kinds of gauge bosons, the first being the photon that mediates the electromagnetic interactions. The second are the W and Z bosons, which are responsible for mediating the weak interaction. Finally, there are gluons, which mediate the strong interaction between quarks that are described in quantum chromodynamics (QCD).

The different interactions above arise from different symmetry groups under which the gauge field transforms. The standard model interactions are defined by the gauge symmetry $U(1) \times SU(2) \times SU(3)[3]$, where each individual group is responsible for the electromagnetic, weak and strong interaction respectively. Associated to each symmetry group is a number of colours or flavours that dictate the interactions of the theory through the structure constants of the group f^{abc} .



Figure 1: Diagrams of the types of self-interactions of gluons (curly lines) and gravitons (double wavy lines). The self interaction of gravitons arise at every other in the pertubative expansion of the theory, while gluons are restricted to cubic and quartic self-interactions.

The other fundamental theory is that of gravity, which describes the interactions of the fourth fundamental force of the universe. Gravity is not incorporated in the standard model of particle physics due to the fact that it is not straightforward to correctly quantise the gravitational field. The best theory of gravity that we do have is general relativity, but a quantisation of this theory leads to divergences of the S-matrix for high energies (the so-called UV divergences)[4]. Nevertheless, it is possible to formulate these theories of quantum gravity [5] to study the interactions that arise from it in the hopes of ironing out these UV divergences [6]. Moreover, the amplitudes of the particles that arise from the quantisation of the gravitational field are called gravitons. It is also possible to use graviton scattering amplitudes to aid in the calculations of inspiraling binary black holes and neutron stars that produce gravitation waves [7, 8].

Gauge theory and gravity are fundamentally different theories. The first describe the interactions of elementary particles, while the second shapes the curvature of spacetime and dictates the evolution of the large scale structure of the universe. At a mathematical level these QFTs give rise to very different types of interactions. It can be shown that (non-Abelian) gauge theory allows for cubic and quartic self-interactions of gluons through the Yang-Mills Lagrangian

$$\mathcal{L}_{\rm YM} = -\frac{1}{4} F^a_{\mu\nu} F^{a\,\mu\nu}, \qquad \text{with} \qquad F^a_{\mu\nu} = \partial_\mu A^a_\nu - \partial_\nu A^a_\mu + g f^{abc} A^b_\mu A^c_\nu. \tag{1}$$

Gravitons on the other hand, arise from a pertubative expansion of the Einstein-Hilbert action

$$\mathcal{L} = -\frac{1}{16\pi G}\sqrt{-g}R, \qquad \text{with} \qquad g_{\mu\nu} = \eta_{\mu\nu} + h_{\mu\nu}, \qquad (2)$$

where the amplitudes are typically built from an expansion in $h_{\mu\nu}$ [9]. Due to this expansion, we find that there are self-interaction at every order. The self interactions of gluons and gravitons are displayed in Figure 1.

While in theory it is possible to calculate the scattering amplitudes of gravitons from their Lagrangian, it is in practice a computational nightmare. Not only do we have to consider

all the possible self-interactions, but already for each cubic graviton vertex we have to account for 171 terms [10, 11]. Regardless of this, a great number of simplifications occurs, which immensely reduces the expression for the resulting amplitude. This behaviour hints at hidden symmetries at the level of the amplitude that are not visible in the Lagrangian of the theory. These types of features that are hidden at the level of the Lagrangian, are the main motivation of the modern S-matrix programme, where scattering amplitudes are built from first principles such as symmetries, factorisation, and infrared behaviour.

There is a particularly interesting hidden connection in the amplitudes of gauge theory and gravity. The story starts in gauge theory, where the tree-level amplitudes $\mathcal{A}_m^{\text{tree}}$ can be formatted as sums over distinct cubic diagrams

$$\mathcal{A}_m^{\text{tree}} \sim \sum_{i \in \text{cubic}} \frac{c_i n_i}{D_i},\tag{3}$$

where each diagram is assigned a colour factor c_i , a kinematic numerator n_i and a denominator D_i . It was presented by Zvi Bern, John Joseph Carrasco and Henrik Johansson [12] that a general property of gauge theory amplitudes is that the kinematic numerators satisfy the same algebraic identity as the colour factors

$$c_i + c_j + c_k = 0,$$
 $n_i + n_j + n_k = 0,$ (4)

which is known as the *colour kinematics* (CK) *duality*. The CK-duality allows for a replacement of colour factors by another copy of kinematic numerators

$$\mathcal{A}_{m}^{\text{tree}} \sim \sum_{i \in \text{cubic}} \frac{c_{i} n_{i}}{D_{i}} \xrightarrow{c_{i} \to \tilde{n}_{i}} \sum_{i \in \text{cubic}} \frac{\tilde{n}_{i} n_{i}}{D_{i}} \sim \mathcal{M}_{m}^{\text{tree}}.$$
(5)

The objects $\mathcal{M}_4^{\text{tree}}$ that are constructed through this replacement turn out to be the treelevel graviton amplitudes. This procedure is known as the *double copy*, which lends its name to the fact that we are creating a copy of the kinematic numerators, and doubling it in the expression for the amplitude [13]. The relation between the two theories is often referred to as gravity being the 'square' of Yang-Mills, summarised as

$$Gravity = (Yang-Mills)^2.$$
(6)

Remarkably, CK-duality is not limited to gauge and gravity theories; it extends to a whole web of theories, ranging from effective scalar field theories to supersymmetric field theories.[14, 15].

The fact that there exist numerators that satisfy the same algebra as the colour factors is quite remarkable. The search for a *kinematic algebra*, analogous to the Lie algebra of the colour factors, is an active area of research [16, 17]. Equally important is the effort to understand this duality at the level of the Lagrangian [18].

Given a class of field theory, it is not immediately clear that such numerators even exist. If they do exist, finding expressions for the numerators n_i that satisfy CK-duality is a nontrivial task, as standard methods like Feynman rules generally do not automatically produce such numerators. A recent publication by Li, Roest and Ter Veldhuis [19] approaches the construction of numerators of scalar field theory amplitudes from a group theory perspective, categorising the representations of the symmetric group S_n that satisfy the CK-duality conditions and relating them to the representations that the kinematics live in.

This classification allows for the construction of 'kinematic building blocks', out of which numerators at any order in Mandelstam invariants can be bootstrapped. From these numerators it is possible to construct amplitudes of the gauged nonlinear sigma-model (gNLSM) that describes pions interacting with gluons. We will refer to this bootstrapping of numerators and amplitudes from the BCJ classifications as the 'BCJ bootstrap'. The full theory that is constructed obeys to a soft degree of $\sigma = 0$, which indicates that the amplitude scales as $A \sim p^0$ in infrared (IR) regime, where p is tuned to 0. However, there is a subsector of amplitudes that can be constructed from this theory that obey a soft degree of $\sigma = 1$.

This *hybrid soft behaviour* is interesting, as typically theories have a singular soft degree. In the case of the NLSM and other exceptional effective field theories, it is possible to uniquely define the theory based on the soft degree of the amplitudes. The soft degree is then strongly related to the symmetries of these theories. The uniqueness of a theory through its symmetries and soft theorems makes theories with a hybrid soft degree particularly intriguing to study [20].

The idea of a relationship between gauge theory and gravity amplitudes first emerged from string theory through the Kawai-Lewellen-Tye (KLT) relations [21]. These relations show how closed string (graviton) amplitudes can be expressed in terms of products of open string (gauge theory) amplitudes.

A different approach of relating the amplitudes within the double copy framework was proposed by Cheung, Shen and Wen [22]. These Unifying Relations are a set of differential operators, referred to as transmutation operators \mathcal{T} that act on the tree-level amplitudes of one theory (A_a) to produce (partial) amplitudes of a different theory (A_b)

$$\mathcal{T} \cdot A_a \to A_b. \tag{7}$$

This is a top-down approach that reduces the spin of the particles in the amplitudes to construct (mixed) amplitudes of a variety of theories. Of particular interest to this work are the amplitudes of scalar fields coupled to pion fields through gauge interactions, which can be constructed from a transmutation of Yang-Mills gluon amplitudes.

The aim of this thesis will be to use the framework of unifying relations to construct a new type of transmutation operator that lands on the amplitudes of the gNLSM constructed using the BCJ bootstrap. Through this construction, we aim to answer several questions on transmutations of amplitudes and the amplitudes of the BCJ bootstrap:

1. How can we construct a transmutation operator that generates amplitudes for comparison with those derived from the BCJ bootstrap?

2. What types of amplitudes can be derived from a generalised transmutation operator?

3. How can generalised transmutation operators inform us about the uniqueness of the theory developed via the BCJ bootstrap?

To answer these questions, we first need a solid understanding of the amplitudes we will be studying. We begin by analysing Yang-Mills amplitudes in section 1.1, where we will explain the details of the colour factors and how they can be used to decompose amplitudes. After the discussion on Yang-Mills amplitudes we turn our attention to pion amplitudes, described by the nonlinear sigma-model section 1.2. We conclude chapter 1 with a discussion on two topics from the modern amplitudes programme, where we will see that both of the amplitudes of Yang-Mills theory and the NLSM have interesting properties in certain kinematic regimes.

Following this, we will take a closer look at the BCJ double copy in section 1.5. Here we explore the objects that can be constructed using the CK-duality, such as graviton amplitudes and amplitudes of the so-called 'biadjoint scalar theory'. Furthermore, we will discuss how higher-derivative operators can be added to the Lagrangians of gluons and pions and what this means for the duality between colour and kinematics.

After we have discussed the details of scattering amplitudes and the double copy, we address the challenge of constructing duality-satisfying numerators in chapter 2. In this section, we elaborate on the approach of Li, Roest and Ter Veldhuis to construct amplitudes that contain the hybrid soft degree using the BCJ bootstrap.

Keeping in mind our goal of constructing these amplitudes using transmutation operators, we dedicate section chapter 3 to discuss the details of the unifying relations of Cheung and collaborators. We will see how we can construct operators that transmute gluons into pions scalars coupled to gluon, which will be generalised in chapter 4 with the aim of landing on the same amplitudes of the BCJ bootstrap. Finally, chapter 5 will be dedicated to comparing the amplitudes of these two methods and addressing the above research questions.

Chapter 1

Scattering Amplitudes

We have seen that gluon amplitudes play a fundamental role in the double copy formalism. The amplitudes are made up of both elements that are essential in the formalism: colour and kinematics. In the first section of this chapter, we will discuss the theory behind gluon amplitudes, which is known as Yang-Mills theory. In this discussion, we will take a closer look at the colour factors and algebra that dictate the theory and its amplitudes.

After discussing gluons, we will turn our attention to the theory and amplitudes of a different type of particle: pions. This theory, part of a class of particle models known as Nonlinear Sigma Models (NLSM), is also governed by symmetry groups. However, instead of focusing on the colour structure as we did with gluons, we will examine the flavour structure of pions. We will explore how these amplitudes exhibit interesting properties, such as a flavour-kinematics decomposition and exceptional behaviour in the infrared (IR) regime.

The properties of scattering amplitudes in the IR regime are then further studied in the following section on *soft theorems*. We find that by tuning the momenta of our external legs to vanish, we can not only recover general properties of physics such as conservation of charge and the equivalence principle, but also build up higher-point amplitudes from lower point amplitudes and heavily constrain the amplitudes of pions.

Throughout this entire work, when talking about scattering amplitudes, we will refer to tree-level amplitudes of massless, on-shell particles (obeying the EOM of the respective field with $p_{\mu}p^{\mu} = 0$), unless stated otherwise. Many of the properties of amplitudes that we derive do not easily generalise to loop-level amplitudes, and have sometimes only been proven to hold at low loop orderS.

1.1 Gluons - Yang-Mills Theory

1.1.1 Gauge Theories

When studying physics, the first field theory we are introduced to is that of electrodynamics. After we have familiarised ourselves with Maxwell's equations, we encounter the relativistic formulation of this theory that describes a four-potential A_{μ} whose components relate to the electric and magnetic field. We discover that different four-potentials can give rise to the same electromagnetic fields, which introduces us to the concept of gaugefreedom. The four-potential gives rise to the formulation of the electromagnetic tensor

$$F_{\mu\nu} = \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu}. \tag{1.1}$$

The Lagrangian for Maxwell's equations (in the absence of any sources) is then given by

$$\mathcal{L} = -\frac{1}{4} F_{\mu\nu} F^{\mu\nu}. \tag{1.2}$$

In a quantum field theory description, we can quantise the field A_{μ} after choosing a specific gauge [2]. The quantisation gives rise to a photon with two polarisation states, as required. The gauge invariance of electrodynamics is one example of a gauge symmetry.

The generalisation of such field theories is called Yang-Mills theory [23]. These theories describes massless, spin-1 vector fields A_{μ} in D dimensions that transform according to the adjoint representation of a gauge goup G. The above example of such a gauge group is U(1) which gives rise to the dynamics of the photon field. The group U(1) is an Abelian group, but we will be investigating non-Abelian gauge groups, such as SU(N). Examples of interactions that arise from non-Abelian groups are the electroweak interactions which are described by the gauge group $SU(2) \times U(1)$ and QCD, which is described by SU(3). This last example describes the theory of gluons that we are interested in, but we will keep our discussion generalised to the group SU(N). The Yang-Mills Lagrangian is denoted as

$$\mathcal{L}_{\rm YM} = -\frac{1}{4} F^{a}_{\mu\nu} F^{a\,\mu\nu}, \qquad \text{with} \qquad F^{a}_{\mu\nu} = \partial_{\mu} A^{a}_{\nu} - \partial_{\nu} A^{a}_{\mu} + g f^{abc} A^{b}_{\mu} A^{c}_{\nu}, \qquad (1.3)$$

where g denotes the coupling strength, the index $a = 1, 2, ..., \dim G$ denotes the colour index of the field A_{μ} and we use notation where summation over repeated indices is implied. The structure constants f^{abc} are derived from the defining Lie algebra of SU(N)

$$\left[t^a, t^b\right] = i f^{abc} t^c, \tag{1.4}$$

where t^a denote the generators of the Lie group.

The Lagrangian (1.3) is invariant under gauge transformations of the fields [24]

$$A^a_\mu \to A^a_\mu + \frac{1}{g} \partial_\mu \theta^a - f^{abc} \theta^b A^c_\mu.$$
(1.5)

To construct amplitudes from this theory we first have to remove the gauge redundancy through the process of gauge-fixing¹. We will choose to work in the *Feynman- 't Hooft gauge* $\xi = 1$. The scattering amplitude, sometimes referred to as the S-matrix, of the theory will also be invariant under gauge transformations.

As a consequence of this gauge invariance, for external on-shell particles the polarisation $e_{\mu}(k)$ of some gluon with momentum k can be decoupled from the amplitude. A gauge transformation $e_{\mu}(k) \rightarrow e_{\mu}(k) + k_{\mu}$ then leads to the vanishing of the amplitude

$$\mathcal{A}(k) = e_{\mu}(k)\mathcal{A}^{\mu}(k) \xrightarrow{e_{\mu}(k) \to e_{\mu}(k) + k_{\mu}} k_{\mu}\mathcal{A}^{\mu}(k) = 0.$$
(1.6)

This vanishing of the amplitude is referred to as the Ward identity [25].

1.1.2 Interactions and Amplitudes

Now that the Lagrangian of Yang-Mills theory has been properly defined, we look towards creating the amplitudes that arise from it. The type of interactions that arise from the Lagrangian are cubic and quartic interactions. The Feynman rules of these interactions are given by [2]

$$a, \mu = gf^{abc} \left[g^{\mu\nu} (k_1 - k_2)^{\rho} + g^{\nu\rho} (k_2 - k_3)^{\mu} + g^{\rho\mu} (k_3 - k_1^{\nu}) \right], \quad (1.7)$$

$$c, \rho = b, \nu$$

$$a, \mu = b, \nu$$

$$a, \mu = b, \nu$$

$$c, \rho = b, \nu$$

$$= -ig^2 \left\{ \begin{cases} f^{abe} f^{cde} \left(g^{\mu\rho} g^{\nu\rho} - g^{\mu\sigma} g^{\nu\rho} \right) \\ + f^{ace} f^{bde} \left(g^{\mu\nu} g^{\rho\sigma} - g^{\mu\sigma} g^{\nu\rho} \right) \\ + f^{ade} f^{bce} \left(g^{\mu\nu} g^{\rho\sigma} - g^{\mu\rho} g^{\nu\sigma} \right) \end{cases} \right\}. \quad (1.8)$$

With these Feynman rules, we can construct an m-point amplitude by considering the different possible topologies that can be constructed from the 3- and 4-point diagrams. For reasons that will be explained shortly, we will specifically be looking to create diagrams out of the 3-point vertex, which we call *cubic* diagrams.

At tree level, there are (2m - 5)!! possible cubic distinct diagrams. This number is found by considering that we can construct higher-point diagrams by attaching a new external

¹A thorough method of gauge-fixing is the Faddeev-Popov method. The treatment requires introducing new fields to the Lagrangian, whose excitations are called *Faddeev-Popov ghosts*. Luckily (or by design of the universe) the contribution of these ghosts exactly cancel other unwanted/unphysical contributions.[24]

leg to a lower-point diagram. Given an *n*-point diagram, there are (2n - 3) edges onto which we can attach a new leg. Therefore the number of cubic diagrams at a given order is given by

$$1 \times 3 \times 5 \times \dots \times (2n-3) = (2m-5)!!.$$
(1.9)

An example that we will consider shortly is the m = 4-point amplitude, which will have 3 distinct cubic diagrams, constructed out of the three possibilities of attaching a fourth leg to a 3-point diagram. The three distinct diagrams are shown in Figure 1.1.



Figure 1.1: Feynman diagrams that depict the three possible 4-point cubic diagrams. The momenta of the propagators are indicated by the Mandelstam invariants s, t and u.

The term 'distinct' means that each diagram has a unique propagator contribution D_i . Moreover, each diagram has its own colour factor c_i made out of contractions of structure constants and kinematic numerator n_i that contains information about the momenta p_i and the polarisations e_i of the external legs.

After assigning a colour factor, kinematic factor, and propagator to each distinct diagram, the total amplitude can then be organised as a summation over all diagrams given by

$$\mathcal{A}_{m}^{\text{tree}} = -ig^{m-2} \sum_{i=1}^{(2m-5)!!} \frac{c_{i}n_{i}}{D_{i}}, \qquad (1.10)$$

where the sum runs over the possible cubic diagrams. A useful example is the 4-point gluon amplitude. There are two topologies that contribute to the amplitude, namely the s, t and u exchange diagrams and a 4-point contact diagram. The full tree-level 4-point amplitude can be written as

$$i\mathcal{A}_4^{\text{tree}} = g^2 \left(\frac{n_s c_s}{s} + \frac{n_t c_t}{t} + \frac{n_u c_u}{u}\right),\tag{1.11}$$

where the Mandelstam variables indicate the exchange channels depicted in Figure 1.1 and which are defined as $s = (p_1 + p_2)^2$, $t = (p_2 + p_3)^2$ and $u = (p_1 + p_3)^2$. You may wonder what happened to the 4 point contact diagram of equation (1.8). Such diagrams are reformatted in terms of cubic vertices by multiplying and dividing the numerator of the contact diagram by one of the propagators, i.e. $\frac{s}{s}$. This already implies that the formulation of the kinematic numerators is not unique, as the contact vertex can be absorbed into each of the n_i .

We will return to the example of the 4-point gluon amplitude in our discussion of the BCJ double copy in subsection 1.5.1, but first we will need to deepen our understanding of the details of the colour algebra and structuring of the theory.

1.1.3 Colour factors and algebra

When we are discussing the duality between colour and kinematics, the colour factors c_i and their associated algebra are fundamental. To fully understand their significance, let's delve into some of the essential details and key concepts that will be referenced throughout this work.

We first start with the statement that for gauge theories the fields transform according to representations of a gauge-group G. Inherent to this Lie group is the (Lie) algebra of it's generators t^a and the structure constants f^{abc} . The structure constants are defined through the Lie product, which for us physicist is the commutator of the matrices of the generators

$$\left[t^a, t^b\right] = i f^{abc} t^c, \tag{1.12}$$

where the summation over the index c is implied. The elements of the Lie algebra satisfy a Jacobi identity

$$[t^{a}, [t^{b}, t^{c}]] + [t^{b}, [t^{c}, t^{a}]] + [t^{c}, [t^{a}, t^{b}]] = 0, \qquad (1.13)$$

which can be rewritten in terms of structure constants as

$$f^{ade} f^{bcd} + f^{bde} f^{cad} + f^{cde} f^{abd} = 0, (1.14)$$

$$(t^{a})_{i}^{k}(t^{b})_{k}^{j} - (t^{b})_{i}^{k}(t^{a})_{k}^{j} = if^{abc}(t^{c})_{i}^{j}.$$
(1.15)

It should be noted that the generators can be normalised in different ways. The above follows the normalisations of $\text{Tr}\left[t^a t^b\right] = \delta^{ab}/2$, but when calculating amplitudes this leads to cumbersome factors of 2 that can be avoided when renormalising the generators and structure constants as

$$T^a \equiv \sqrt{2}t^a. \qquad \qquad \tilde{f}^{abc} \equiv i\sqrt{2}f^{abc}. \qquad (1.16)$$

The identities of equation (1.14) are then redefined as

$$\tilde{f}^{ade}\tilde{f}^{bcd} + \tilde{f}^{bde}\tilde{f}^{cad} + \tilde{f}^{cde}f^{abd} = 0, \qquad (1.17)$$

$$(T^{a})_{i}^{k}(T^{b})_{k}^{j} - (T^{b})_{i}^{k}(T^{a})_{k}^{j} = \tilde{f}^{abc}(T^{c})_{i}^{j}.$$
(1.18)

Note that the summation over the index k equates to taking the trace of these products of matrices, $Tr(T^aT^b)$.

When we construct the amplitude, each diagram has an associated colour factor c_i which is constructed out of contractions of representations of structure constants f^{abc} that are associated to the vertices of the diagrams. In the next section, we will construct the colour factors explicitly and see how these allow us to remove some of the redundancy in the number of diagrams that we have to sum over in order to construct amplitudes.



Figure 1.2: Feynman diagram of m-point gluon amplitude. The colour factor can be deduced by following a path from leg 1 through leg m. [15]

1.1.4 Colour-Ordered Partial Amplitudes

The number of diagrams at a given multiplicity can be reduced by employing relations between diagrams such as the Jacobi identity. At *m*-point, the (2m - 5)!! diagrams each have their own colour factor c_i , but there are (2m - 5)!! - (m - 2)! independent Jacobi relations of the form (1.17).

Besides the Jacobi identities, through the definition of the algebra $[t^a, t^b] = if^{abc}t^c$ we see that the interchanging of a and b will result in a sign flip for the structure constant. Take for example the four point diagram g_s that indicates the s-channel exchange. This specific ordering of the external legs leads to a colour factor of $c(g_s) = \tilde{f}^{a_1 a_2 b} \tilde{f}^{ba_3 a_4}$. Flipping legs 1 and 2 still results in an s-channel diagram, g'_s , but with a new colour factor $c(g'_s) =$ $\tilde{f}^{a_2 a_1 b} \tilde{f}^{ba_3 a_4}$. We therefore realise that there is an anti-symmetry under the exchange of these two legs $c(g'_s) = -c(g_s)$.

For an adjoint-only theory, any colour factor can be written as a product of adjoint generator matrices $(\tilde{f}^a)_{bc} \equiv \tilde{f}^{bac}$. By following a path from leg 1 to leg *m* of a given diagram and denoting the corresponding (commutators of) \tilde{f}^a 's of the vertices we encounter along the path, we can denote the colour factor of any diagram. For example, the diagram in Figure 1.2 results in the colour factor

$$\left(\tilde{f}^{a_2}\left[\tilde{f}^{a_3},\tilde{f}^{a_4}\right]\left[\tilde{f}^{a_5}\left[\tilde{f}^{a_6},\tilde{f}^{a_7}\right]\right]\cdots\tilde{f}^{a_{m-1}}\right)_{a_1a_m},\tag{1.19}$$

where we have used the Lie-algebra identity $\tilde{f}^{abc}\tilde{f}^c = [\tilde{f}^a, \tilde{f}^b]$. The colour factor of (1.19) can be generalised to any permutation σ of external legs, giving

$$c_i = \sum_{\sigma \in S_{m-2}} b_{i\sigma} \left(\tilde{f}^{a_{\sigma(2)}} \tilde{f}^{a_{\sigma(3)}} \tilde{f}^{a_{\sigma(4)}} \cdots \tilde{f}^{a_{\sigma(m-1)}} \right)_{a_1 a_m}, \qquad (1.20)$$

where the prefactor $b_{i\sigma} \in \{0, \pm 1\}$ depends on the specific permutation and colour factor. There are (m-2)! permutations of σ , as indicated by the permutation group S_{m-2} .

Returning to the summation over colour factors to obtain the total amplitude in (1.10), we find that this transforms into a sum over (m-2)! ordered partial amplitudes

$$A_m^{\text{tree}}(1, \sigma(2), \sigma(3), \dots, \sigma(m-1), m),$$
 (1.21)

where leg 1 and m are fixed. These define a basis called the Kleiss-Kuijf (KK) basis. The summation over partial amplitudes and their corresponding colour trace is then given by

$$\mathcal{A}_{m}^{\text{tree}} = g^{m-2} \sum_{\sigma \in S_{m-2}} A_{m}^{\text{tree}}(1, \sigma(2), \sigma(3), \dots, \sigma(m-1), m) \left(\tilde{f}^{a_{\sigma(2)}} \tilde{f}^{a_{\sigma(3)}} \cdots \tilde{f}^{a_{\sigma(m-1)}} \right)_{a_{1}a_{m}},$$
(1.22)

and is referred to as the Del Duca-Dixon-Maltoni (DDM) colour decomposition [26].

The partial tree amplitudes $A_m^{\text{tree}}(1, 2, ..., m)$ in Yang-Mills theory have several useful properties. The following properties will be of relevance to our discussion of gluon amplitudes:

- 1. Gauge invariance: Individually each partial amplitude is invariant under gauge transformations.
- 2. Functions of kinematic variables e_i and p_i only. Specifically, they are functions of the Lorentz invariant dot products of e_i and p_i ; $(e_i \cdot e_j)$, $(e_i \cdot p_j)$, and $(p_i \cdot p_j)$. This will become a key aspect of the Unifying Relations between different theories discussed in chapter 3.
- 3. Cyclicity: partial amplitudes are invariant under cyclic permutations.

$$A_m^{\text{tree}}(1, 2, \dots, m) = A_m^{\text{tree}}(2, \dots, m, 1).$$
 (1.23)

4. **Reflectivity:** partial amplitudes exhibit a sign flip under reversal of the ordering

$$A_m^{\text{tree}}(m,\ldots,2,1) = (-1)^m A_m^{\text{tree}}(1,2,\ldots,m).$$
(1.24)

5. A **Photon-decoupling identity**² is satisfied by the partial amplitudes.

$$\sum_{\sigma \in cyclic} A_m^{\text{tree}}(1, \sigma(2), \dots, \sigma(m)) = 0, \qquad (1.25)$$

where we sum over cyclic permutations of all-but-one external leg.

6. Obey (fundamental) BCJ relations which take the form

$$\sum_{i=2}^{m-1} p_1 \cdot (p_2 + \ldots + p_i) A_m^{\text{tree}}(2, \ldots, i, 1, i+1, \ldots, m) = 0.$$
 (1.26)

For example, at four point such a relation is $tA_4(1324) - sA_4(1234) = 0$.

These properties describe linear relations between partial amplitudes, allowing us to reduce the summation over all possible diagrams to a summation over the independent partial amplitudes. After considering all permutations of the above BCJ relation, there are only (m-3)! independent partial tree amplitudes. The position of the three consecutive legs can be fixed in the cyclic ordering; for example $A_m^{\text{tree}}(1, \sigma(2), \ldots, \sigma(m))$ can be chosen as the independent BCJ basis.

²This can be seen by replacing one generator in the trace decomposition of the next section by a U(1) generator $T_{U(1)} = \mathbb{I}$. Gluons do not couple directly to photons because photons have no colour charge. Explicitly $\tilde{f}^{abU(1)} = \text{Tr}\left(\left[T^a, T^b\right] 1\right) = 0$, implying that the amplitude vanishes for one photon.

1.1.5 Trace basis decomposition

By relating the structure constants to traces of the generators,

$$\tilde{f}^{abc} \equiv i\sqrt{2}f^{abc} = \operatorname{Tr}\left(\left[T^a, T^b\right]T^c\right) = \operatorname{Tr}\left(T^aT^bT^c\right) - \operatorname{Tr}\left(T^bT^aT^c\right).$$
(1.27)

We can decompose the product of \tilde{f} 's that is present in the decomposition of the total amplitude into partial amplitudes as a summation over permutations of trace structures.

Let's continue with one of our four point colour factors, $c(g_s) = \tilde{f}^{a_1 a_2 b} \tilde{f}^{b a_3 a_4}$. We can use the SU(N) Fierz completeness relations,

$$(T^{a})_{i}^{j} (T^{a})_{k}^{l} = \delta_{i}^{l} \delta_{k}^{j} - \frac{1}{N} \delta_{i}^{j} \delta_{k}^{l}, \qquad (1.28)$$

to derive the trace decomposition of the colour factor. We can perform this calculation explicitly for the four-point colour factor, but for higher multiplicities it is convenient to employ an algorithm for the decomposition. For the colour factor $c(g_s)$ we find

$$\begin{split} \tilde{f}^{a_{1}a_{2}b}\tilde{f}^{ba_{3}a_{4}} &= \operatorname{Tr}\left[\left(T^{a_{1}}T^{a_{2}} - T^{a_{2}}T^{a_{1}}\right)T^{b}\right] \times \operatorname{Tr}\left[T^{b}\left(T^{a_{3}}T^{a_{4}} - T^{a_{4}}T^{a_{3}}\right)\right] \\ &= \left(T^{a_{1}}_{ij}T^{a_{2}}_{jk} - T^{a_{2}}_{ij}T^{a_{1}}_{jk}\right)T^{b}_{ki} \times T^{b}_{lm}\left(T^{a_{3}}_{mn}T^{a_{4}}_{nl} - T^{a_{4}}_{mn}T^{a_{3}}_{nl}\right) \\ &= \left(T^{a_{1}}_{ij}T^{a_{2}}_{jk} - T^{a_{2}}_{ij}T^{a_{1}}_{jk}\right)\left(\delta_{km}\delta_{li} - \frac{1}{N}\delta_{ki}\delta_{lm}\right)\left(T^{a_{3}}_{mn}T^{a_{4}}_{nl} - T^{a_{4}}_{mn}T^{a_{3}}_{nl}\right) \\ &= \left(T^{a_{1}}_{ij}T^{a_{2}}_{jk}T^{a_{3}}_{ni} - T^{a_{1}}_{ij}T^{a_{2}}_{jk}T^{a_{4}}_{kn}T^{a_{3}}_{ni} - T^{a_{2}}_{ij}T^{a_{1}}_{jk}T^{a_{3}}_{kn}T^{a_{4}}_{ni} + T^{a_{2}}_{ij}T^{a_{1}}_{jk}T^{a_{4}}_{kn}T^{a_{3}}_{ni}\right) \\ &= \operatorname{Tr}\left(T^{a_{1}}T^{a_{2}}T^{a_{3}}T^{a_{4}}\right) - \operatorname{Tr}\left(T^{a_{1}}T^{a_{2}}T^{a_{4}}T^{a_{3}}\right) \\ &- \operatorname{Tr}\left(T^{a_{1}}T^{a_{4}}T^{a_{3}}T^{a_{2}}\right) + \operatorname{Tr}\left(T^{a_{1}}T^{a_{3}}T^{a_{4}}T^{a_{2}}\right). \end{split}$$

Note that implicit summation over repeated indices is implied. The terms proportional to 1/N cancel out. Each trace factor occurs uniquely in every permutation, and will belong to the partial amplitude of its permutation. In other words, the partial tree amplitude $A_4^{\text{tree}}(1,2,3,4)$ is the kinematic coefficient of $\text{Tr}(T^{a_1}T^{a_2}T^{a_3}T^{a_4})$ and so forth for other permutations. The total amplitude is then denoted as

$$\mathcal{A}_{m}^{\text{tree}} = g^{2} \left(A_{4}(1,2,3,4) \operatorname{Tr} \left(T^{a_{1}} T^{a_{2}} T^{a_{3}} T^{a_{4}} \right) + \operatorname{perm}(2,3,4) \right).$$
(1.30)

Combining this decomposition with (1.11) we find the following expressions for the partial amplitudes in terms of the kinematic numerators

$$iA_4^{\text{tree}}(1,2,3,4) = \frac{n_s}{s} - \frac{n_t}{t},$$
(1.31)

$$iA_4^{\text{tree}}(1,3,2,4) = \frac{n_t}{t} - \frac{n_u}{u},$$
(1.32)

$$iA_4^{\text{tree}}(1,2,4,3) = \frac{n_u}{u} - \frac{n_s}{s},$$
(1.33)

From these relations we can confirm that the partial amplitudes indeed satisfy the fundamental BCJ relations of (1.26)

$$stA_4^{\text{tree}}(1,2,3,4) = utA_4^{\text{tree}}(1,3,2,4) = suA_4^{\text{tree}}(1,2,4,3).$$
 (1.34)

The trace-decomposition of 4-point can be generalised to m-point. The decomposition of any colour factor permutation can be written as

$$\left(\tilde{f}^{a_{\sigma(2)}}\tilde{f}^{a_{\sigma(3)}}\cdots\tilde{f}^{a_{\sigma(m-1)}}\right)_{a_{1}a_{m}} = \operatorname{Tr}\left(T^{a_{1}}T^{a_{2}}\cdots T^{a_{m}}\right) + (-1)^{m}\left(T^{m}\cdots T^{a_{2}}T^{a_{1}}\right) + \dots,$$
(1.35)

and the *trace basis decomposition* of the total amplitude can be denoted as

$$\mathcal{A}_{m}^{\text{tree}} = g^{m-2} \sum_{\sigma \in S_{m-1}} A_{m}^{\text{tree}}(1, \sigma(2), \sigma(3), \dots, \sigma(m-1), m) \operatorname{Tr} \left(T^{a_{1}} T^{a_{\sigma(2)}} T^{a_{\sigma(3)}} \cdots T^{a_{\sigma(m)}} \right).$$
(1.36)

This is a sum over (m-1)! terms, which is a larger basis than the (m-2)! basis of (1.22). The two are mapped into one another using the KK relations

$$A_m^{\text{tree}}(1, \{\alpha\}, m, \{\beta\}) = (-1)^{|\beta|} \sum_{\sigma \in \{\alpha\} \sqcup \sqcup \{\beta\}^T} A_m^{\text{tree}}(1, \sigma, m).$$
(1.37)

The notation appears cumbersome, but an example at five-point should clarify what is happening. For starters, $\{\alpha\}$ and $\{\beta\}$ are lists of the external legs. If we wish to denote the partial amplitudes related to for example $A_5(1, 2, 5, 4, 3)$, we construct the sets $\{\alpha\} = 2$ and $\{\beta\} = 3, 4$. The object $\{\beta\}^T$ represents reverse ordering of the list $\{\beta\}$, which is therefore $\{4, 3\}$. Next, $\{\alpha\} \sqcup \{\beta\}^T$ denotes the shuffle product, or ordered permutations, of these lists which in our case will be $\{\alpha\} \sqcup \lfloor \beta\}^T = \{2\} \sqcup \lfloor \{4, 3\}$ and gives rise to the permutations $\sigma = \{243\}, \{423\}, \{432\}$. Finally, the exponent $|\beta|$ denotes the number of elements in the list $\{\beta\}$. The KK relations for this ordering are then

$$A_5(1,2,5,4,3) = A_5(1,2,4,3,5) + A_5(1,4,2,3,4) + A_5(1,4,3,2,5).$$
(1.38)

These linear relations reduce the basis at five-point from 24 partial amplitudes down to six independent partial amplitudes, as desired.

1.2 Pions - Nonlinear Sigma Model

Now that we have discussed the detail of gauge theories, we will turn our attention to a different particle sector: scalars. Such theories typically offer a great introduction to the principles of QFT and scattering amplitudes due to their simple properties such as the absence of colour structure and being spin-0.

In this section we will investigate how we can extend and refine scalar theories to derive a more comprehensive and intriguing theory of pions. Pions we originally thought to be the mediator of the strong force, before the discovery of the gluons of section 1.1 [27]. However, it turned out that pions were only an *effective* description of these interactions at short distances. Their field theory description is therefore formulated in the form of an *effective field theory* (EFT).

Effective field theories describe the dynamics of a field theory in a certain energy regime such as the infrared (IR) or the ultraviolet (UV) regime. Typically, the 'effective' behaviour comes from additional derivative terms in the Lagrangian, which at the level of the amplitude induce higher-order momentum dependence, and therefore energy dependence.

A familiar example of an EFT are the Nambu-Goldstone bosons (NGSB's) which arrises due to spontaneous symmetry breaking, e.g. the Higgs boson [24]. The effective field theory of pions is often referred to as the *nonlinear sigma model* (NLSM) [28]. In this model, pions arise due to the spontaneous breaking of the chiral-flavour symmetry [29] in QCD. In reality they are "pseudo" NGSb's because they obtain small masses from the underlying quarks, but for our purposes we will treat them as massless scalar fields.

1.2.1 Lagrangian

Qualitatively, the NGSb arise from the symmetry breaking of a global group G into a subgroup H. Every broken generator of G gives rise to a NGSb. These NGSb's are then said to live in the coset space³ G/H. In the context of the NLSM we will be using the symmetry breaking pattern $U(N) \times U(N) \mapsto U(N)$ or more specifically we can discuss $SU(N_f) \times SU(N_f)$, where N_f denotes the number of quark flavours of the theories in the context of low-energy QCD [30]. The discussion is the same, keeping in mind that SU(N) has one fewer generator than U(N) which has N^2 generators.

The unitary matrix $U(\phi)$ can be used to represent the Goldstone field, which contains the scalar field ϕ that transforms according to the adjoint representation of G as $\phi = \phi^a T^a$. One of such parametrisations, along with the Lagrangian that describe it's dynamics is

³'If you want to show off your mastery of mathematical jargon you can say that the Nambu-Goldstone bosons live in the coset space G/H' - A. Zee [24]

given by

$$\mathcal{L} = \frac{F^2}{4} \operatorname{Tr} \left(\partial^{\mu} U \partial_{\mu} U^{\dagger} \right), \qquad \qquad U = \exp \left(\frac{i\phi}{F} \right), \qquad (1.39)$$

where F is the NLSM coupling strength, which can be seen as an expansion parameter. Expanding

The symmetry of the group is spontaneously broken by a non-linear shift of the field, $\phi \rightarrow \phi + a$ at first order in the field (not considering terms of $\mathcal{O}(\phi^2)$). It can be shown that the Lagrangian in (1.39) is invariant under this non-linear shift, which is where the name 'non-linear sigma model' comes from.

Being an effective field theory, the full Lagrangian in (1.39) can be written as a progressive expansion in F that contains only an even number of fields.

$$\mathcal{L} = \sum_{n=1}^{\infty} \mathcal{L}_{2n}.$$
 (1.40)

Each derivative of ϕ will contribute a factor of p at the level of the amplitude, as can be seen from a fourier transformation of the derivative in momentum-space. Every \mathcal{L}_{2n} contains 2n derivatives and is therefore of $\mathcal{O}(p^{2n})$. As an effective field theory, we can therefore consider only terms up to a certain order in p, or equivalently, a certain order in n.

Typically, we consider the $\mathcal{O}(p^2)$ Lagrangian \mathcal{L}_2 and it's expansion in terms of the scalar fields ϕ :

$$\mathcal{L}_{2}^{\text{NLSM}} = -\frac{1}{2} \partial_{\mu} \phi_{a} \partial^{\mu} \phi^{a} + \frac{1}{6F^{2}} f_{abe} f^{cde} \phi^{a} \partial_{\mu} \phi^{b} \phi^{c} \partial^{\mu} \phi^{d} + \dots$$
(1.41)

From this Lagrangian, we can construct the amplitudes of the NLSM, which has interesting properties in soft limit due to it's momentum scaling.

1.2.2 Amplitudes

Similarly to the colour ordering in YM theory, due to their flavour structure, the NLSM amplitudes can be decomposed using the familiar trace structure of Equation 1.36 as

$$\mathcal{A}_n^{\text{NLSM}} = \sum_{\sigma \in S_n/\mathbb{Z}_n} Tr\left(T^{a_1}T^{a_{\tau(1)}}\cdots T^{a_{\sigma(n)}}\right) A_n(\sigma_{a_1}, \sigma_{a_2}, \dots, \sigma_{a_n}).$$
(1.42)

The colour stripped amplitudes of the NLSM are cyclically symmetric and contain poles only in adjacent factorisation channels. This is a property we will explore in more detail for a different scalar theory, namely the bi-adjoint scalar in subsection 1.5.2.

The first non-vanishing amplitude of the NLSM can be found at n = 4. The 3-point amplitude vanishes, due to the fact that momentum conservation all Mandelstam invariants

 s_{ij} must vanish at 3-point. The four-point partial amplitude is generated by a single type of diagram, namely the 4-point contact diagram.

The 4-point partial amplitude with the trace ordering [1234] can be denoted as

$$A_4(1,2,3,4) = \frac{i}{2F^2}(s+t), \qquad (1.43)$$

where in most literature only the kinematic part of this amplitude is of interest, as the constant factors can be conveniently redefined.

It can be seen that even though the NLSM has two derivatives per vertex, the amplitudes of the NLSM are linear in all momenta and scale as p^1 , which would typically be observed if all fields in the Lagrangian contain a derivative. This is what is called an 'enhanced soft limit', which is defined by the *soft degree* that is higher than what we would expect from the number of derivatives per field [31].

The p^1 -scaling of the NLSM amplitudes was first found by Stephen Adler and it implies the vanishing of the amplitudes of the NLSM in the soft limit [32]. Specifically, the amplitude vanishes in the IR regime with a *soft degree* of $\sigma = 1$ which we will further investigate in section 1.4. This is widely known as the *Adler zero* condition, or *Adler's zero*:

$$\lim_{p \to 0} A_{\text{NLSM}} \propto p^1. \tag{1.44}$$

The Adler zero turns out to be an incredibly strong constraint on the amplitudes, and is sometimes regarded as a 'gauge symmetry' for such scalar EFTs [33]. One of the useful properties that can be derived from the Adler zero is the fact that the NLSM amplitudes can be unique formulated by imposing Adler's zero [34], similarly to how YM and gravity amplitudes can be uniquely formulated by imposing gauge invariance.

Moreover, higher-point NLSM amplitudes can be constructed from lower-point amplitudes through the process of *soft recursion* [35], which will be discussed in subsection 1.4.4.

1.3 Factorisation - Recursion relations

In a typical amplitude Feynman-diagram calculation of amplitudes, we construct the diagrams out of the basic vertices of the theory. At the level of diagrams, it is clear that higher-point diagrams can be built from lower-point diagrams. However, this is not a statement about amplitudes. In this section we will see that lower-point amplitudes *recursively* appear in higher-point amplitudes. There are different approaches to deriving this recursive behaviour of amplitudes. In this section we will describe the method of *on-shell recursion* [36].

On-shell recursion relies on the fact that tree-level amplitudes exhibit specific pole structures that are limited to forms such as $\frac{1}{s}$. Physically, this singularity can be interpreted as an intermediate state propagating over a physical distance in spacetime, hence the term 'propagator'. Imagine a scenario where the initial and final states are separated by a significant distance in spacetime, as illustrated for a four-particle diagram

In this limit, the intermediate state must be on-shell, and the initial and final vertices can be interpreted as separate, lower-point diagrams. This decomposition of higher-point amplitudes into lower-point amplitudes is known as 'factorisation'. We will see that the four point amplitude factorizes into a sum over products of 3-point amplitudes for the s, tand u factorisation channels.

1.3.1 On-Shell Recursion

A concise formulation of on-shell recursion was presented by Britto, Chachazo and Feng and Witten (BCFW)[37, 38]. The proposed BCFW recursion relations were derived by shifting two external legs by some complex factor z:

$$p_i \to p_i + zq,$$
 $p_j \to p_j - zq.$ (1.46)

This shift preserves the total momentum conservation as both momenta are oppositely shifted. By construction the vector q is defined such that $q^2 = p_i \cdot q = p_j \cdot q = 0$. Now that the momenta have been shifted onto the complex plane, the amplitude becomes dependent on the complex factor, implying $A \to A(z)$. We can then apply Cauchy's formula

$$f(a) = \frac{1}{2\pi i} \oint \frac{f(z)}{z - a} dz, \qquad (1.47)$$

to find an expression for the original unshifted amplitude by integrating around the pole at z = 0. We then apply Cauchy's residue theorem to find

$$A(z=0) = \frac{1}{2\pi i} \oint \frac{A(z)}{z} dz = -\sum_{I} \operatorname{Res}_{z=z_{I}} \left[\frac{A(z)}{z} \right] + B_{\infty}.$$
 (1.48)

Here I denotes a subset of external particles. Their sum of momenta is given by

$$P_I(z) = \sum_{i \in I} p_i(z).$$
(1.49)

The sum over z_I denotes a summation over the residues at all other poles. These poles are the kinematic singularities (propagators) of the amplitude and hence indicated by the external legs I that contribute to the propagator. At such a singularity $(z \rightarrow z_I)$, the propagator and the sub-amplitudes that connect to it are on-shell. The amplitude should therefore factorize into a product of lower-point amplitudes

$$\lim_{z \to z_I} P_I^2(z) A(z) = A_L(z_I) A_R(z_I).$$
(1.50)

The sum over all possible *factorisation channels* I then gives rise to the general formula for on-shell recursion, which denotes that the amplitude decomposes into the summation over all possible decompositions of lower-point amplitudes that have shifted momenta on both sides of the factorisation channel

$$A(z=0) = \sum_{I} A_{L}(z_{I}) \frac{1}{P_{I}^{2}} A_{R}(z_{i}) + B_{\infty}.$$
(1.51)

The boundary term that denotes the residue at $\hat{\infty}$ can cause problems for the on-shell constructibility of a theory. A theory is considered on-shell constructible if a momentum shift can be found such that $B_{\infty} = 0$ for every tree-level scattering amplitude. In [38] it was proven that A(z) vanishes at infinity for Yang-Mills, but the story is a bit more complicated for the NLSM [39]. For pions, it is then more convenient to use a different method, namely *soft recursion*. To explore this, we will have to investigate the IR behaviour of amplitudes, which will be done in the next section.

Now that we are equipped with shell-on-shell recursion relations, it is possible to start with a *seed* amplitude such as a 3-particle amplitude and build higher multiplicity amplitudes by imposing factorisation. For example, a 4-point amplitude can be seen to split up into the product of two 3-point amplitudes.

1.4 Soft Theorems and Soft Factors

In the discussion of recursion we have seen that shifting the external legs of the amplitude gives rise to the factorisation of the amplitude in the limit of the internal propagators becoming on-shell. It would also be interesting to probe what would happen if we tune one, or more, of the external legs to vanish. Such a limit is called the *soft limit*, and is explored by assigning a dimensionless parameter λ to the momentum as λp_{μ} , and sending $\lambda \to 0$. When probing this limit, the amplitude will behave according to a specific *soft degree* σ as

$$\lim_{p \to 0} A \propto p^{\sigma},\tag{1.52}$$

where $\sigma = -1, 0, 1, ...$

An amplitude is said to satisfy a *soft theorem* when the right-hand side of the equation features a universal structure, such as vanishing of the amplitude or as some factor multiplied by a lower-point amplitude [40]. It is possible to find an amplitude to satisfy a soft theorem as a byproduct of the action, but the inverse is also possible: impose the soft theorem in order to derive the S-matrix of the theory that satisfies this constraint.

We will start our discussion by exploring amplitudes that exhibit the soft degree $\sigma = -1$. It turns out that gauge theory and graviton amplitudes have this scaling. The soft behaviour of these theories has been famously studied by Steven Weinberg in the 1960s [41, 42], resulting in the *soft factors* of the amplitudes being named in his honour. We will examine both of these and see how, together with the other symmetries of the theories, we can find interesting soft theorems.

1.4.1 Soft factor of gauge theory

We will analyse the behaviour of gauge theory amplitudes by looking at scalar QED interactions, following the arguments of [3] to demonstrate the soft factors conceptually. The scalar QED derivation can be straightforwardly extended to QED interactions by denoting the electrons as spinors with the correct contractions of γ^{μ} . The (relevant) Feynman rules of scalar QED are

$$e^{-} p_{2} = -iQe(p_{1} + p_{2})^{\mu} \text{ and } e^{-} p_{2} = +iQe(p_{1} + p_{2})^{\mu}, \quad (1.53)$$

$$e^{-} p_{1} = -iQe(p_{1} + p_{2})^{\mu}, \quad (1.53)$$

where $Q_i e$ denotes the electric charge of the scalar particle (= ±1 for e^+/e^-).

There are two scenarios to consider. The first describes an outgoing photon attached to one of the incoming external legs (e^- or e^+), while the second will have the photon attached to one of the outgoing legs. This is depicted in figure Figure 1.3.



Figure 1.3: Attaching the soft photon to incoming or outgoing e^+/e^- .

In the first scenario, taking the external leg to be an e^- for now, the electron leg is attached to some scalar QED process, possibly involving many external legs and even loops. The total amplitude for the entire process is then denoted by $iM_0(p_i)$. Next, we attach an outgoing photon of momentum q and polarisation ϵ_{μ} to this e^- leg and describe the amplitude as $iM_i(p_i, q)$. Diagrammatically this is shown as

$$iM_0(p_i) = e^{-} \xrightarrow{p_i} \otimes \implies e^{-} \xrightarrow{p_i} q = iM_i(p_i, q).$$
(1.54)

Explicitly the amplitude now receives an additional contribution from the 3 point vertex and an additional propagator contribution

$$M_i(p_i,q) = (-iQ)\frac{i\left(p_i^{\mu} + (p_i^{\mu} - q^{\mu})\right)}{(p_i - q)^2 - m^2}\epsilon_{\mu}M_0(p_i - q).$$
(1.55)

After applying the on-shell conditions for the external legs $p_i^2 = m^2$ and $q^2 = q \cdot e = 0$, equation (1.55) simplifies to

$$M_i(p_i, q) \approx -Q \frac{p_i \cdot \epsilon}{p_i \cdot q} M_0(p_i), \qquad (1.56)$$

at leading order in the soft limit where the photon momentum is taken to be much smaller than the external electron/position momenta, $|q \cdot p_i| \ll |p_j \cdot p_k|$.

A similar derivation can be performed for incoming e^+ and outgoing e^-/e^+ . For the total amplitude we have to sum over all possible incoming/outgoing external electron/positron legs that the external photon can be attached to, giving

$$M \approx eM_0 \left[\sum_{i \in \text{incoming}} Q_i \frac{p_i \cdot \epsilon}{p_i \cdot q} - \sum_{i \in \text{outgoing}} Q_i \frac{p_i \cdot \epsilon}{p_i \cdot q} \right].$$
(1.57)

By the Ward identity (1.6) this amplitude should be invariant under the transformation $\epsilon_{\mu} \rightarrow \epsilon_{\mu} + \alpha q_{\mu}$, which implies that total charge should be conserved

$$\sum_{i \in \text{incoming}} Q_i = \sum_{i \in \text{outgoing}} Q_i.$$
(1.58)

Gauge invariance therefore requires the conservation of whatever coupling constant, like electric charge, governs the interaction of these particles at low energies for all amplitudes involving a massles spin-1 particle.

The pre-factor that is multiplied by the lower-point amplitudes of (1.57) is called the Weinberg soft factor [42]

$$S_{\text{QED}}^{\mu} = \sum_{i}^{n} Q_{i} \frac{p_{i}^{\mu}}{p_{i} \cdot q}.$$
 (1.59)

This soft factor is the leading order contribution to the amplitude in the soft limit and allows us to relate higher point amplitudes at higher points to lower point amplitudes.

For Yang-Mills theories the soft factor is given by

$$S_{\rm YM}^{(ijk)} = \frac{p_i e_j}{p_i p_j} - \frac{p_k e_j}{p_k p_j},$$
(1.60)

where we take leg i to be the soft leg. Note that the soft factor explicitly depends on the momenta of the adjacent particles i and k.

Subleading gluon soft factor

It is interesting that at leading order, the soft factor does not depend on the spin of the soft particle. This is not the case for the subleading soft factor, where explicit dependence on angular momentum appears. For gluons in Yang-Mills theory this subleading factor is given by [43, 44]

$$S_{\rm YM,sl}^{(ijk)} = \frac{p_j J_k e_j}{p_k p_j} - \frac{p_j J_i e_j}{p_i p_j} = -\frac{p_j J_i e_j}{p_i p_j} + (i \leftrightarrow k),$$
(1.61)

where particle j is taken to be the soft particle. Here J_i denotes the total angular momentum of the particle i and the notation $p_j J_i e_j = p_j^{\mu} (J_i)_{\mu\nu} e_j^{\nu}$ denotes the summation over the Lorentz indices. It can be seen that the factor depends on the angular momentum of the legs adjacent to the soft leg.

1.4.2 Soft gravitons

Although graviton amplitudes are not the main focus of this thesis, we would like to emphasise the interesting soft theorems that can be derived from the soft behaviour of gravity. Instead of describing the interaction of spin-1 particles, we can examine a soft spin-2 particle by substituting the polarisation ϵ^{μ} for $\epsilon_i^{\mu\nu}$. The deduction relies on the same arguments, however the coupling of the interaction is different than scalar QED, it is governed by the gravitational coupling κ_i . The soft factor is then given by [45]

$$S^{\mu\nu} = \sum_{i} \kappa_i \frac{p_i^{\mu} p_i^{\nu}}{p_i \cdot q}.$$
(1.62)

The argument based on gauge invariance has an analogue in gravity amplitudes, which are invariant under diffeomorphisms of the form $\epsilon_{\mu\nu} \rightarrow \epsilon_{\mu\nu} + \Lambda_{\mu}q_{\nu} + \Lambda_{\nu}q_{\mu}$. Just as gauge invariance ensures that electromagnetic or Yang-Mills fields remain consistent under local transformations, diffeomorphism invariance ensures that the equations governing gravity are consistent under arbitrary smooth changes of the spacetime coordinates. Applying this argument we can derive the following soft theorem from soft gravitons

$$\sum_{\rm in} \kappa_i p_i^{\mu} = \sum_{\rm out} \kappa_i p_i^{\mu}.$$
(1.63)

This equation is severely constrained by the conservation of momentum, $\sum_{in} p_i^{\mu} = \sum_{out} p_i^{\mu}$. There can be no more constraints on the momenta p_i , as this would give only a trivial solution $p_i = 0$. Therefore, the only possible amplitudes all have $\kappa_i = \kappa$. Hence, the soft theorem of massless spin-2 particles implies that gravity is universal, which is the foundation of the theory of general relativity.

Subleading graviton soft factor

As discussed, the soft factor is a pertubative expansion relating higher multiplicity amplitudes to lower-point amplitudes. For graviton amplitudes this expansion can be written as [45]

$$M_{n+1}(k_1, k_2, \dots, k_n, q) = \left(S^{(0)} + S^{(1)} + S^{(2)}\right) M_n(k_1, k_2, \dots, k_n) + \mathcal{O}\left(q^2\right).$$
(1.64)

With $S^{(0)}$ being the leading soft factor, $S^{(1)}$ the subleading factor and so forth. We have already seen the leading order $S^{(0)}$ in our analysis. The subleading soft factors require a more complicated analysis, as shown in [45] and are given by

$$S^{(1)} \equiv -i \sum_{i=1}^{n} \frac{e_{\mu\nu} k_i^{\mu} q_{\rho} J_i^{\nu\rho}}{k_i \cdot q}, \qquad (1.65)$$

$$S^{(2)} \equiv -\frac{1}{2} \sum_{i=1}^{n} \frac{e_{\mu\nu} \left(q_{\rho} J_{i}^{\mu\rho}\right) \left(q_{\sigma} J_{i}^{\nu\sigma}\right)}{k_{i} \cdot q}, \qquad (1.66)$$

With $e_{\mu\nu} = e_{\mu}e_{\nu}$ being the soft graviton polarisation tensor satisfying $e_{\mu\nu}q^{\nu} = 0$ and J_i being the total angular momentum of the *i*-th particle.

1.4.3 Higher-spin soft theorem

Can we go one step higher and examine the soft factor of some massless spin-3 particle? Certainly! The equation in (1.63) will transform into

$$\sum_{\rm in} \beta_i p_i^{\nu} p_i^{\mu} = \sum_{\rm out} \beta_i p_i^{\nu} p_i^{\mu}, \qquad (1.67)$$

where β_i corresponds to the coupling of a generic spin-3 form vertex.

Consider the $\mu = \nu = 0$ index of the momentum vectors, which are equal to the energy of the particles $p_i^0 = E_i$. The soft theorem on conservation changes to

$$\sum_{\text{in}} \beta_i E_i^2 = \sum_{\text{out}} \beta_i E_i^2, \qquad (1.68)$$

which tells us that the sum of 'charge' $\times E^2$ is conserved. However, this is too many constraints! The only possible solution would be if all $\beta_i = 0$, resulting in a boring non-interacting theory.

The soft theorem then tells us that there are no interacting theories of massless particles with s > 2 [42]. Theories of massive spin-3 particles, e.g. composite particles, are allowed and have been observed at collider experiments [46].

1.4.4 Soft recursion

The above exceptional scalar EFT are constructible via on-shell recursion. This can be shown through a *soft shift* of one of the external legs, by making use of the soft degree σ of the theory. The shift that we consider,

$$p_i \to p_i(1 - za_i), \tag{1.69}$$

conserves the on-shell kinematics only if the number of particles n > d + 1, where d is the dimension in which we are scattering. The dimensionality gives us a constraint on the number of particles in order to make a_i distinct for each leg. This constraint arises from the fact that if n = d, then the momenta p_i are linearly independent and total momentum conservation is not preserved. The case where n = d + 1 is not interesting, as the only solution would be to have all legs equally rescaled with $a_i = 1$ which does not probe any interesting features.

To probe the soft limit where the momentum p_i goes to zero we send $z \to \frac{1}{a_i}$. Given the soft degree σ , the amplitude then scales as

$$\lim_{z \to 1/a_i} A(z) \propto \left(z - \frac{1}{a_i}\right)^{\sigma}.$$
(1.70)

By construction, A(z) now contains multiple zeros of degree σ in the complex plane and we can apply Cauchy's residue theorem, as before in the discussion of on-shell recursion;

$$A(z=0) = \frac{1}{2\pi i} \oint \frac{A(z)}{z} dz = -\sum_{I} \underset{z=z_{I}}{\operatorname{Res}} \left[\frac{A(z)}{z} \right] + B_{\infty}.$$
 (1.71)

Next we construct a function F(z) which vanishes in the soft limit, but whose unshifted value is equal to 1, so that we can relate it to the unshifted amplitude A(0).

$$F(z) = \prod_{i}^{n} (1 - a_i z)^{\sigma}, \qquad F(0) = 1.$$
(1.72)

We will insert this function into the denominator of (1.71) which would in general introduce new poles into the expression, which are not the factorisation channels responsible for the original poles. However, by clever construction of (1.72) these poles are cancelled by the zeroes that arise from the soft degree of (1.70).

$$A(z=0) = \frac{A(z=0)}{F(0)} = \frac{1}{2\pi i} \oint \frac{A(z)}{zF(z)} dz = -\sum_{I} \operatorname{Res}_{z=z_{I}} \left[\frac{A(z)}{zF(z)} \right] + B_{\infty}.$$
 (1.73)

The recursion into a summation over factorisation channels of lower-point amplitude then occurs, in a similar manner as the BCFW recursion relations. There is still the crucial detail of the boundary term B_{∞} which occurs for the limit $z \to \infty$, which sends our momenta to the high-energy spectrum. In general, the UV behaviour of EFTs is poor, as they are nonrenormalisable theories. The only way for B_{∞} to vanish is for F(z) to grow faster

$$\lim_{z \to \infty} \frac{A(z)}{F(z)} = 0. \tag{1.74}$$

This will vanish precicely for the theories with an enhanced soft limit. First, as F(z) is a polynomial of at most degree $n\sigma$, it will scale as $z^{n\sigma}$ for large z. The amplitude A(z) scales at most as z^m , where m is the number of derivatives for the n particle vertex. Therefore B_{∞} vanishes for $\sigma > m/n$, proving that the exceptional EFTs are on-shell constructible.

As an example of soft recursion for EFTs, we will calculate the 6-point NLSM amplitude from the 4-point amplitudes [47]. First we note that odd-multiplicity pion amplitudes always vanish, therefore there is no 5-point amplitude to build from. We saw that the 4-point flavour-ordered amplitude $A_4^{\text{NLSM}}(1234)$ is given by

$$A_4(1,2,3,4) = s + t = s_{12} + s_{23}, (1.75)$$

up to a normalisation. We can identify three factorisation channels, namely s_{123} , s_{234} and s_{345} , which are depicted in Figure 1.4. The three factorisation channels are cyclic permutations of eachtother. The total 6-point, flavour-ordered amplitude is the sum over these three channels. We can calculate this amplitude to be

$$A_6^{\text{NLSM}}(123456) = \left[\frac{(s_{12} + s_{23})(s_{45} + s_{56})}{s_{123}} + \text{cyclic}\right] - (s_{12} + \text{cyclic}), \quad (1.76)$$



Figure 1.4: The three possible factorisation channels of the 6-point NLSM amplitude into 4-point amplitudes.

where the second term comes from the residue

$$\sum_{i=1}^{6} \operatorname{Res}_{z=z_{I}} \frac{s_{12} + \operatorname{cyclic}}{zF(z)} = -(s_{12} + \operatorname{cyclic}).$$
(1.77)

It can be clearly seen that the amplitude splits up into the left-side amplitude $A_L(123x)$ and the right-side amplitude $A_R(x456)$ over the exchange channel s_{123} , in agreement with the calculation from Feynman diagrams [47].

1.5 The Double Copy

Now that we have familiarised ourselves with the properties of colour factors and the decompositions of amplitudes, we are prepared to dive deeper into the concept of colour-kinematics duality.

In this chapter, we will investigate the BCJ double copy method, which is one of the methods used to construct graviton amplitudes from gluon amplitudes by utilising the duality between the colour and kinematic numerators. We will elaborate on the 4-point gluon amplitude, as an example that can be extended to higher-multiplicity gluon amplitudes.

During our investigation of the double copy, we will also encounter the 'single copy' and the scalar amplitudes it creates. We will also briefly touch upon a different approach to the BCJ double copy, namely the KLT double copy.

Finally, we will investigate how adding higher-derivative correction terms to the Lagrangians of the theories under consideration can break the duality between colour and kinematics of the amplitudes, and what needs to be done to fix this.

1.5.1 BCJ Duality between Colour and Kinematics

The Four-Point Gluon Amplitude

One example that beautifully demonstrates the BCJ duality is the 4-gluon amplitude described in Yang-Mills theory. The formulation of this amplitude is compact enough to manipulate by hand such that we are able to understand what is going on before we dive into m-point amplitudes. Moreover it is a familiar amplitude for those of us who have studied QFT before.

As we have seen in the previous chapter, the full tree-level amplitude at 4-point can be written as a sum over the three exchange channels s, t and u

$$i\mathcal{A}_4^{tree} = g^2 \left(\frac{n_s c_s}{s} + \frac{n_t c_t}{t} + \frac{n_u c_u}{u}\right). \tag{1.78}$$

These channels are shown diagrammatically in **figure**. The Mandelstam variables are defined as $s = (p_1 + p_2)^2$, $t = (p_2 + p_3)^2$ and $u = (p_1 + p_3)^2$. We have seen that the colour factor for the s-channel is constructed from the product of structure constants that correspond to the vertices of the diagram:

$$c_s = -2f^{a_1 a_2 b} f^{b a_3 a_4}, (1.79)$$

where the structure constants are normalised as $\left[t^a,t^b\right]=if^{abc}t^c$.

Through Feynman rule computations, we can also find the expression for the kinematic

numerator for the s-channel, n_s ,

$$n_{s} = -\frac{1}{2} \left\{ \left[(e_{1} \cdot e_{2})p_{1}^{\mu} + 2(e_{1} \cdot p_{2})e_{2}^{\mu} - (1 \leftrightarrow 2) \right] \left[(e_{3} \cdot e_{4})p_{3\mu} + 2(e_{3} \cdot p_{4})e_{4\mu} - (3 \leftrightarrow 4) \right] + s \left[(e_{1} \cdot e_{3})(e_{2} \cdot e_{4}) - (e_{1} \cdot e_{4})(e_{2} \cdot e_{3}) \right] \right\}, \quad (1.80)$$

where e_i denotes the polarisation of the external gluon legs.

The other combinations of colour and kinematic factors are obtained through cyclic permutations:

$$c_t n_t = c_s n_s |_{1 \to 2 \to 3 \to 1}$$
 $c_u n_u = c_s n_s |_{1 \to 3 \to 2 \to 1}.$ (1.81)

The contribution of the contact vertex is absorbed into the kinematic numerators as discussed in subsection 1.1.2.



Figure 1.5: Feynman diagrams that depict the three possible 4-point cubic diagrams. The momenta of the propagators are indicated by the Mandelstam invariants s, t and u.

The colour factors of the 4-gluon amplitude

The colour factors are fully encoded by the Lie-algebra associated to the gauge group under which the field transform. In the case of gluons, the Lie-algebra we are considering is su(3), of which the f_{abc} 's are the structure constants. These structure constant satisfy the Jacobi identity:

$$c_s + c_t + c_u = -2\left(f^{a_1a_2b}f^{ba_3a_4} + f^{a_2a_3b}f^{ba_1a_4} + f^{a_3a_1b}f^{ba_2a_4}\right) = 0.$$
(1.82)

This identity also appears as a requirement for the amplitude to be invariant under linearised gauge transformations. By utilising the Ward identity of (1.6), upon replacing for example $e_4 \rightarrow p_4$ we find

$$n_s|_{e_4 \to p_4} = -\frac{s}{2} \left[(e_1 \cdot e_2)((e_3 \cdot p_2) - (e_3 \cdot p_1)) + \text{cyclic}(1, 2, 3) \right] \equiv s\alpha(e, p).$$
(1.83)

Applying this transformation to the t and u channel, we find that the total amplitude transforms as

$$\frac{n_s c_s}{s} + \frac{n_t c_t}{t} + \frac{n_u c_u}{u} \Big|_{e_4 \to p_4} = (c_s + c_t + c_u)\alpha(e, p).$$
(1.84)

Therefore, in order for the amplitude to be invariant under gauge transformation, the Jacobi identity must be satisfied.

A kinematic Jacobi identity

If, we would decide to calculate the sum of the kinematic numerators n_i , we would see that under the on-shell conditions this sum gives rise to an identity analogous to the Jacobi identity for colour factors. This is also known as the 'kinematic Jacobi identity':

$$n_s + n_t + n_u = 0. (1.85)$$

This analogy between the colour factors and kinematic numerators is referred to as the 'duality between colour and kinematics' and the n_i that together satisfy this duality are referred to as BCJ numerators. In literature, the shorthand 'CK duality' or similar terms are often used to refer to this statement.

Gravitons from Gluons

Knowing that these factors satisfy the same algebraic relations, a natural question to ask would be "what would happen if we swap out one for the other?". It turns out that by swapping colour factors for kinematic factors in the Yang-Mills four-point amplitude, we obtain a new gauge-invariant object:

$$i\mathcal{A}_m^{\text{tree}} = g^2 \left(\frac{n_s c_s}{s} + \frac{n_t c_t}{t} + \frac{n_u c_u}{u}\right) \xrightarrow[g \to \kappa/2]{} \left(\frac{\kappa}{2}\right)^2 \left(\frac{n_s^2}{s} + \frac{n_t^2}{t} + \frac{n_u^2}{u}\right) \equiv i\mathcal{M}_4^{\text{tree}}.$$
 (1.86)

This new amplitude, $\mathcal{M}_4^{\text{tree}}$, effectively 'doubles' the kinematic numerators, which is where the term "double copy" originates. This expression has several key properties that indicate that this amplitude is in fact a valid graviton amplitude. First of all, the external states are described by symmetric polarisation tensors $e_{\mu\nu} = e_{\mu}e_{\nu}$. Secondly, the interactions involve two derivatives. Furthermore, the amplitude remains invariant under linearised diffeomorphism transformations

$$\delta h_{\mu\nu} = \partial_{\mu}\xi_{\nu} + \partial_{\nu}\xi_{\mu}u, \qquad (1.87)$$

which is the analog of gauge theory amplitudes being invariant under linearised gauge transformations. In momentum space, this implies that the amplitude should vanish upon the transformation $e^{\mu\nu} \rightarrow p^{\mu}e^{\nu} + p^{\nu}e^{\mu}$. Calculating this explicitally gives:

$$\frac{n_s^2}{s} + \frac{n_t^2}{t} + \frac{n_u^2}{u} \bigg|_{e_4^{\mu\nu} \to p_4^{\mu} e_4^{\nu} + p_4^{\nu} e_4^{\mu}} = 2(n_s + n_t + n_u)\alpha(e, p) = 0.$$
(1.88)

Finally, when we choose the initial gluon polarisation vectors to be circularly polarised, implying $e^2 = 0$, the resulting polarisations $e_{\mu\nu}$ are traceless. All of this brings us to the conclusion that this amplitude describes the scattering of four gravitons in Einsteins general relativity, up to an overall normalization factor.

The example at 4-point can be generalised to higher-multiplicity amplitudes. Here the crucial argument is that it is always possible to find and relate a triplet of diagrams (i, j, k) that satisfy Jacobi relations of the form

$$c_i + c_j + c_k = 0, (1.89)$$

and similarly for the kinematic numerators n_i [15]. The identification of these triplets follow from the investigation of diagrams discussed in section 1.1.

1.5.2 Zeroth Copy - Biadjoint Scalar (BAS) Theory

So far, we have seen that the duality between colour and kinematics allows us to replace the colour factors c_i of tree-level YM amplitudes by another kinematic numerator \tilde{n}_i to produce tree level graviton amplitudes. This double copy procedure is denoted as

$$i\mathcal{A}_m^{\text{tree}} \sim \sum_{\text{trivalent}} \frac{n_i c_i}{D_i} \xrightarrow{c_i \to \tilde{n}_i} \sum_{\text{trivalent}} \frac{n_i \tilde{n}_i}{D_i} \sim i\mathcal{M}_4^{\text{tree}}.$$
 (1.90)

By the same logic, we should be able to do the inverse; replace the kinematic numerator by a colour factor that obeys the same algebraic identities. This is known as the *zeroth copy* and the amplitudes that we land on are that of the exotic *biadjoint scalar* (BAS) theory

$$i\mathcal{A}_m^{\text{tree}} \sim \sum_{\text{trivalent}} \frac{n_i c_i}{D_i} \xrightarrow[\text{trivalent}]{} \sum_{\text{trivalent}} \frac{c_i \tilde{c}_i}{D_i} \sim i\mathcal{A}_m^{\text{BAS,tree}}.$$
 (1.91)

This theory consists of a scalar field with two charges, $\phi = \phi^{a\bar{a}}T^a\tilde{T}^{\bar{a}}$, that transforms according to adjoint representation of 2, possibly different gauge groups G and \tilde{G} whose generators are T^a and $\tilde{T}^{\bar{a}}$ [48]. The Lie algebras of these groups define the structure constants:

$$if^{abc}T^c = [T^a, T^b], \qquad \qquad if^{\overline{a}\overline{b}\overline{c}}\tilde{T}^{\overline{c}} = \operatorname{Tr}\left[\tilde{T}^{\overline{a}}, \tilde{T}^{\overline{b}}\right].$$
 (1.92)

The Lagrangian of this theory is denoted by

$$\mathcal{L} = \frac{1}{2} \partial_{\mu} \phi^{a\overline{a}} \partial^{\mu} \phi^{a\overline{a}} + \frac{\lambda}{3!} f^{abc} \tilde{f}^{\overline{a}\overline{b}\overline{c}} \phi^{a\overline{a}} \phi^{b\overline{b}} \phi^{c\overline{c}}, \qquad (1.93)$$

and results in ϕ^3 interactions. The principle of a bi-adjoint scalar that transforms according to two gauge group can be generalised beyond double colour ordering to a higher number of colour orderings. An extensive discussion of this topic can be found in [49].

At tree level, the amplitudes contain only (massless) propagators, as there are no kinematic factors in the decomposition of equation (1.91), nor can they arrise from the Lagrangian. The amplitudes of BAS theory can also be decomposed in a similar manner to (1.36). Due

to the dual colour charge, the amplitudes decompose into partial amplitudes which obey a dual trace structure, denoted as

$$\mathcal{A}_{m}^{\text{BAS}} = \sum_{\sigma} \sum_{\sigma'} \operatorname{Tr} \left[T^{a_{\sigma(1)}} \cdots T^{a_{\sigma(n)}} \right] \operatorname{Tr} \left[T^{\overline{a}_{\sigma'(1)}} \cdots T^{\overline{a}_{\sigma'(n)}} \right] A_{BAS}(\sigma_{1}, \cdots, \sigma_{n} | \sigma_{1}', \cdots, \sigma_{n}').$$
(1.94)

The two orderings σ and σ' are separated in the double colour-ordered partial amplitudes A_{BAS} by a vertical | and refer to the specific ordering of the trace structures that accompany this specific ordering.

A convenient method of calculating these partial amplitudes can be found in [50] and is briefly summarised in Appendix A.

1.5.3 Higher Derivative Corrections

We have extensively discussed the BCJ double copy, but there is another formulation of the double copy, namely the Kawai-Lewellen-Tye (KLT) formulae [21]. Similar to the BCJ formulation, it allows us to construct tree-level gravity amplitudes \mathcal{M}^{tree} by taking the (KLT) product of purely-adjoint gauge theory amplitudes. For 3- and 4-point they explicitly give the graviton amplitudes in the form of

$$\mathcal{M}_{3}^{tree}(1,2,3) = iA_{3}^{tree}(1,2,3)\tilde{A}_{3}^{tree}(1,2,3), \tag{1.95}$$

$$\mathcal{M}_{4}^{tree}(1,2,3,4) = -is_{12}A_{4}^{tree}(1,2,3,4)\tilde{A}_{4}^{tree}(1,2,3,4), \qquad (1.96)$$

where the amplitudes A^{tree} and \tilde{A}^{tree} are the colour-ordered partial amplitudes discussed in subsection 1.1.4. As the KLT double copy is not the main focus of this work, we leave a brief discussion of this formalism to Appendix B.

Originally the KLT relations were derived in the field of string theory as relations between colour-stripped disk amplitudes of open-strings. In these relations, there is an explicit dependence on the string tension α' . In the low energy limit, where α' goes to zero, the string-theory KLT kernel reduces to the field theory KLT kernel described. Interestingly the string tension α' can be used to formulate a pertubative expansion of higher-order corrections to the field theories. In the following sections we will explore constructing such higher-order order corrections by starting from field theory amplitudes instead of string theory amplitudes.

We now return to the discussion of QFTs, where we can improve the UV behaviour of a theory by adding higher-derivative corrections, in order to create an effective action, in the spirit of section 1.2. The ultimate goal is to remove UV divergences that occur in loop amplitudes. We can do a pertubative expansion in some parameter α' and investigate the properties of the amplitudes. Specifically we are interested whether or not the colour-kinematics duality is still present after adding such terms. We will see that in order to satisfy CK-duality for some order in α' at a specific multiplicity it will be necessary to include an operator at one order higher in α' [51].
CK-Duality of Higher Order Yang-Mills

For a pure-gluon Yang-Mills theory we need to consider that the higher order corrections that we include are gauge-invariant and conserve locality. The simplest operator, i.e. lowest order in derivatives, that can then be constructed for YM is a correction to the familiar F^2 operator of subsection 1.1.2 and is denoted by the F^3 operator

$$F^{3} \equiv \operatorname{Tr}\left(F^{\nu}_{\mu}F^{\rho}_{\nu}F^{\mu}_{\rho}\right) = \frac{1}{2}\operatorname{Tr}\left([T^{a},T^{b}]T^{c}\right)F^{a\nu}_{\mu}F^{b\rho}_{\nu}F^{c\mu}_{\rho} = \frac{1}{2}\tilde{f}^{abc}F^{a\nu}_{\mu}F^{b\rho}_{\nu}F^{c\mu}_{\rho}.$$
 (1.97)

The effective Lagrangian produced by this deformation then contains the first order in α' correction

$$\mathcal{L}_{\mathrm{YM}+F^{3}} = \frac{1}{4} \operatorname{Tr}\left(F^{2}\right) + \frac{\alpha'}{3} \operatorname{Tr}\left(F^{3}\right), \qquad (1.98)$$

From which gluon amplitudes can be constructed. Double copy consistent means that the amplitudes obey the CK duality *and* consistently factorize into the correct lower-point amplitudes.

The Lagrangian in (1.98) has been shown to produce amplitudes that satisfy the colourkinematics (CK) duality for specific multiplicities, but not universally. Specifically, at n = 3, the amplitudes satisfy CK duality at $\mathcal{O}(\alpha'^0)$ and $\mathcal{O}(\alpha'^1)$. Additionally, at n = 4, the operator satisfies CK duality at $\mathcal{O}(\alpha'^1)$. However, at the four-point level, there is also a contribution from an amplitude that arises from a combination of two cubic $\alpha' F^3$ vertices. To ensure that the amplitude maintains colour-kinematics duality at this order in α' , an additional term, $\operatorname{Tr}(F^4)$, is required [52].

It was shown by Carrasco and collaborators that this pattern continues [51]. Double copy consistency at any given multiplicity and order in α' requires the addition of an operator that is one order higher in α' . This results in an infinite tower of higher-order corrections. This is different from a scalar theory, such as the NLSM, which only requires a finite number of higher-derivative operators in order for a specific multiplicity to satisfy CK duality.

An Infinite Tower of Higher Derivative Corrections

Consider the five-point amplitude. Similar to the example at n = 4 we note that this amplitude can be constructed out of combination of 3- and 4-point amplitude factorisation.

$$\mathcal{A}_{5}(12345)|_{(k_{4}+k_{5})^{2}-cut} = \sum_{states} \mathcal{A}_{4}(123l^{s}) \mathcal{A}_{3}\left(-l^{\overline{s}}45\right).$$
(1.99)

This is schematically depicted in Figure 1.6. As explained before, the amplitude \mathcal{A}_3 comes from both F^2 and F^3 contributions through $A_3 \equiv A_3^{YM} + \alpha' A_3^{F^3}$. The five-point will then consist of a combination a four-point at $O(\alpha'^n)$ combined with the three point at $O(\alpha'^1)$ resulting in $O(\alpha'^{n+1})$ for the five-point. This contribution can only be colour dual by addition of specific 'four-field operator' of order $O(\alpha'^{n+1})$ that combines with the $Tr(F^2)$ term. This new $O(\alpha'^{n+1})$ term then also has to be contracted with the $\alpha' \operatorname{Tr}(F^3)$ term,



Figure 1.6: Contributions to the factorisation of the five-point tree-level amplitude at $\mathcal{O}(\alpha'^n)$. [51].

inducing a contribution at $O(\alpha'^{n+2})$. We can continue this pattern to construct a ladder of higher-order terms, and it has been proven up to order $O(\alpha'^5)$ by [51].

Given that the CK-duality holds for these parameters, we can apply the double copy procedure by replacing the colour factors with a set of kinematic numerators. As before, this result in gravity amplitudes, which in this case also arise from an effective action. Specifically, the amplitudes of bosonic closed strings at order $\mathcal{O}(\alpha')$ and $\mathcal{O}(\alpha'^2)$. At the level of the action these higher order contributions come partly from a R^2 and R^3 operator respectively. Through KLT relations the double copy procedure can be denoted as

$$\mathcal{A}_n^{F^3} \overset{\mathrm{KLT}}{\otimes} \mathcal{A}_n^{\mathrm{YM}} \mapsto \mathcal{M}_n^{R^2} \text{ at } \mathcal{O}(\alpha'),$$
 (1.100)

$$\mathcal{A}_{n}^{F^{3}} \overset{\mathrm{KLT}}{\otimes} \mathcal{A}_{n}^{F^{3}} \mapsto \mathcal{M}_{n}^{R^{2}} \text{ at } \mathcal{O}\left(\alpha^{\prime 2}\right), \qquad (1.101)$$

where the details are left to [53].

Resummation to $(DF)^2 + YM$

Carrasco and collaborators went back to the four-point amplitude and constructed a double-copy compatible (dcc) numerator of the s channel up to order $\mathcal{O}(\alpha'^5)$ in terms of kinematic building blocks σ that are permutation invariant such that

$$n_{s}^{\text{dcc}} = n_{s}^{\text{YM}} + \alpha' n_{s}^{\text{YM}+F^{3}} + \alpha'^{2} n_{s}^{(F^{3})^{2}+F^{4}} + \alpha'^{3} \left[a_{3} \left(n_{s}^{D^{2}F^{4}} + \sigma_{2} n_{s}^{\text{YM}+F^{3}} \right) + a_{3,\text{YM}} \sigma_{3} n_{s}^{\text{YM}} \right] + \alpha'^{4} \left[a_{4,1} \left(n_{s}^{(DF)^{4}} + \sigma_{2} n_{s}^{(F^{3})^{2}+F^{4}} \right) + a_{4,2} n_{s}^{(DF)^{4}} + a_{4,F^{3}} \sigma_{3} n_{s}^{\text{YM}+F^{3}} \right] + \mathcal{O}(\alpha'^{5}).$$

$$(1.102)$$

Here σ_2 and σ_3 denote quadratic and cubic kinematic building blocks consisting of Mandelstam invariants and the a_i 's are unfixed coefficients.

The question then arises whether the infinite tower resums into a known theory. It was shown that by setting $a_{3,\text{YM}} = 0$ that A_4 and A_5 precisely match the $O(\alpha'^4)$ expansion of

the 'B-amplitudes' in [54]. These amplitude correspond to a YM theory with an additional $(DF)^2$ operator deformed by a massive gauge theory whose mass is set by $1/\alpha'$ [55, 56] and which contains propagators that are constructed from the higher-order operators. This indicates that, by imposing double-copy consistency, the infinite tower of higher-order terms resums into $(DF)^2 + YM$.

Higher Derivative NLSM

In section 1.2, the NLSM Lagrangian we consider is the $\mathcal{O}(p^2)$ Lagrangian of (1.41). It is also possible to consider the terms with a greater amount of derivatives, such as the \mathcal{L}_4 or \mathcal{L}_6 which are of $\mathcal{O}(p^4)$ and $\mathcal{O}(p^6)$ respectively.

It was calculated that for 6-point amplitudes, up to $\mathcal{O}(p^{10})$, additional terms are also required in order to satisfy CK-duality due to the contact term not allowing for duality [57]. It is argued here that requiring DC consistency to arbitrary multiplicity involves adding an infinite chain of operators with fixed coefficients.

Another candidate for calculating the higher-derivative corrections for the NLSM amplitudes can be found in string theory [58]. Specifically, in a double-copy-like approach, it was found that it was possible to replace the Yang-Mills factors of Abelian open-string amplitudes by gauge theory colour factors. This replacement gives rise to Abelian Z-amplitudes [59]. A bief overview of these Z-theory amplitudes can be found in Appendix C.

In the field theory limit, the leading order α' contributions then give rise to NLSM amplitudes of the $\mathcal{O}(p^2)$ Lagrangian:

$$A_{\text{NLSM}}(1, 2, \dots, n) = \lim_{\alpha' \to 0} (\alpha')^{2-n} \sum_{\sigma \in S_{n-1}} Z_{1\sigma(2, 3, \dots, n)}(1, 2, \dots, n).$$
(1.103)

These amplitudes then denote the flavour-ordered NLSM amplitudes. Explicitly at n = 4 and n = 6 for example they give rise to the correct NLSM amplitudes:

$$A_{\rm NLSM}(1,2,3,4) = \pi^2 (s_{12} + s_{23}), \qquad (1.104)$$

$$A_{\rm NLSM}(1,2,\ldots,6) = \pi^2 \left[s_{12} - \frac{1}{2} \frac{(s_{12} + s_{23})(s_{45} + s_{56})}{s_{123}} + cyclic(1,2,3,4,5,6) \right], \quad (1.105)$$

up to a normalisation factor. It is also possible to calculate the subleading terms. These correspond to the higher-derivative corrections of the NLSM.

Chapter 2

BCJ Bootstrap - Hybrid Soft Behaviour

In the previous chapters, we explored the BCJ double copy and observed that the color factors exhibit interesting identities and properties derived from the Lie algebra of the theory under consideration. However, this raises the question: What about the kinematic numerators n_i ? In the 4-point example, we noted that there is no unique formulation of the numerators due to gauge invariance and the possibility of redistributing the contribution of the 4-point contact term into the numerators of the cubic exchange diagrams.

The aim of this chapter is to investigate whether there is a general formulation for the kinematic numerators, specifically for scalar theories. To achieve this, we will delve into the novel approach developed by Yang Li, Diederik Roest, and Tonnis ter Veldhuis, which classifies kinematic numerators that satisfy the BCJ colour-kinematics duality [19]. In this approach, they classify the irreducible representations (irreps) of the symmetric group S_n that meet the conditions of CK duality. With these irreps established, one can match them to the irreps of (products of) Mandelstam invariants out of which kinematic numerators are built. The constructed kinematic numerators will be used to construct amplitudes of the Gauged Nonlinear Sigma Model (GNLSM). The amplitudes of the full theory derived in this context will exhibit a hybrid soft behaviour, producing amplitudes that are characterised by a soft degree of $\sigma = 0$ and $\sigma = 1$.

2.1 A group Theory Approach to Numerators

As stated in subsection 1.5.1, we can formulate amplitudes as a sum over distinct cubic diagrams as

$$A_n = \sum_{\text{trivalent}} \frac{NN}{D},\tag{2.1}$$

where the numerator factors N and \tilde{N} can be referred to as BCJ numerators. These numerators satisfy the CK duality. In this notation, the numerator factors can be choosen

to be either a color or kinematic numerator. As an example we have extensively considered the colour factors consisting of a product of structure constants:

$$N_{abc...} = f_{ab}^{\ x_1} f_{x_1c}^{\ x_2} f_{x_2...} \cdots .$$
(2.2)

These colour numerators N are subject to the following constraints if they are to satisfy the CK duality [60]:

- 1. Anti-symmetric under the exchange of the first two indices $N_{abcd...} = -N_{bacd...}$.
- 2. Even and odd under reflections $N_{abcd...} = (-1)^n N_{...dcba}$.
- 3. The numerators satisfy Jacobi-like nested commutator identities

$$-N_{abcd...} = N_{bacd...} = N_{c[ab]d...} = N_{d[[ab]c]...}.$$
(2.3)

By satisfying the above properties, the colour numerators can then be associated with some of the representations of the symmetric group S_n . In fact, the factors then correspond to several irreducible representations (irreps) of S_n . In order to classify these specific irreps, Young diagrams can be used, which schematically depict the irrep as a collection of boxes with a specific number of rows and columns. Let us go over a brief overview of what these diagrams are and how they are used to characterise the symmetric group.

2.1.1 Young Diagrams and Representations of S_n

The symmetric group S_n is the group of all permutations of a set. The group can tell us how objects such as tensors transform under permutations of the tensor indices. Because the BCJ conditions on numerators discuss properties of numerators, that can be seen as tensors with indices a, b, \ldots , we can view them from a group theory perspective. Specifically, we are interested in the group's irreducible representations.

To classify the irreducible representation of S_n , we first define young diagrams by specifying a partition $\lambda = (\lambda_1, \lambda_2, \dots, \lambda_r)$ of the natural number n to be a sequence of positive integers with

$$\sum_{i=1}^{r} \lambda_i = n \qquad \text{and} \qquad \lambda_i \ge \lambda_{i+1}. \tag{2.4}$$

Young diagrams can then be used to depict partitions graphically, as a collection of n boxes arranged in r rows that are left-aligned. The *i*th row then consists of λ_i boxes. Left-aligned means that diagrams are structured such that the boxes within each row are non-increasing as you move downward, and similarly, non-increasing within each column as you move to the right. These diagrams are also known as "legitimate" diagrams.

Each legitimate diagram corresponds to a unique irreducible representation. Distinct diagrams represent inequivalent irreps, regardless of whether the dimensions of the irreps are identical. For n = 4, there are 5 distinct partitions:



Each partition for n is equal to the number of conjugacy classes¹ of S_n and is therefore equal to the number of distinct irreducible representations. We will use the shorthand notation [3, 2, 1] to describe a diagram with 3 boxes in the first row, 2 boxes in the second row, and 1 box in the third row:

The dimension of an irrep S^{λ} can be calculated using the hook length formula. The hook length h_i of a box *i* in a Young tableau is the number of boxes either below or to the right of that box in the Young diagram, including the box itself. The dimensionality is then given by

$$\dim S^{\lambda} = \frac{n!}{\prod_{i} h_{i}}.$$
(2.7)

If the diagram in (2.6) were to correspond to an irrep, its dimensionality would be dim $[3, 2, 1] = \frac{6!}{5\cdot 3\cdot 1\cdot 3\cdot 1\cdot 1}$.

There are two noteworthy 1-dimensional representations, namely the *singlet* and the *sign* representation. The singlet representation, whose Young-diagram [n] is displayed in Figure 2.1a, is commonly referred to as the 'trivial' representation of S_n due to the fact that it is a one-dimensional representation that sends every element of the group to the identity matrix of order 1.

The sign representation, sometimes referred to as the 'alternating' representation, is denoted by the Young diagram [1, 1, ..., 1] in Figure 2.1b. It sends all even number of permutations of two elements to 1 and all odd number of permutations to -1.

Returning to the discussion of the BCJ conditions on numerators, the irreps of S_n which satisfy these constraints have been classified by [19] and are denoted in Table 2.1. For convenience we display the irreps of S_4 , S_5 and S_6 that satisfy the constraints explicitly:



¹A conjugacy class characterises a cycle structure for permutations. For example, cyclic permutations in S_3 are denoted by the conjugacy class $(abc \rightarrow bca, abc \rightarrow cab)$.



Figure 2.1: Young diagrams of the trivial irrep [n], the sign irrep [1, 1, ..., 1] and the standard representation [n-1, 1].

n	Young diagram
4	[2,2]
5	[3, 1, 1]
6	[4, 2], [3, 1, 1, 1], [2, 2, 2]
7	[5, 1, 1], [4, 2, 1], [3, 3, 1], [3, 2, 1, 1], [2, 2, 1, 1, 1]
	[6, 2], [5, 2, 1], [5, 1, 1, 1], [4, 4], [4, 3, 1],
8	$2 \times [4, 2, 2], [4, 2, 1, 1], [4, 1, 1, 1, 1], 2 \times [3, 3, 2, 2],$
	[3, 2, 2, 1], [3, 2, 1, 1, 1], [2, 2, 2, 2], [2, 2, 1, 1, 1, 1]

Table 2.1: Irreps of S_n that obey the BCJ constraints. [61]

Now we know from a group theory perspective what the irreps are that correspond to objects that satisfy the BCJ constraints. From here, we would like to see how these relate to numerators that are made out of kinematic objects such as Mandelstam invariants.

2.1.2 Mandelstam Representations

We have seen that the kinematic numerators n_i of YM theory are made up of products of momenta and polarisations. For this discussion, we instead consider the numerators for (coloured) scalars. The reason is that for scalar field theories the only information of the external legs are the momentum vectors. There is no polarisation to consider and therefore the kinematic numerator can only consist out of Mandelstam invariants. These are then purely constructed through Lorentz invariant products of momenta $p_i p_j$. We define these Mandelstam invariants as

$$s_{ij\dots k} = (p_i + p_j + \dots + p_k)^2 = 2p_i p_j + 2p_i p_k + 2p_j p_k + \dots,$$
(2.9)

where contraction of the Lorentz indices of the momenta is implied $p_i p_j = (p_i)_{\mu} (p_j)^{\mu}$.

For the theories we consider in this section, the colour factors are all contained in either N or \tilde{N} , leaving the other numerator to solely depend on momentum Mandelstams. Similarly to the colour numerators, the kinematic numerators can be constructed to obey symmetries

of permutations of the external legs. The same language of the symmetric group can then be applied to the kinematic numerators.

For starters, due to momentum conservation, the sum of all momenta vanishes. This summation is invariant under permutation of the momenta and can be said to live in the singlet representation [n], indicating that anything we build with this irrep gives rise to vanishing contributions.

The set of all *n* external momenta, $\{p_1, \ldots, p_n\}$, lives in the *standard* representation of S_n denoted by [n-1, 1] and shown in Figure 2.1c. There are (n-1) independent momentum vectors due to conservation of momentum, which is consistent with the dimension of the representation dim $= \frac{n!}{n \cdot (n-2) \cdot (n-3) \cdots 1} = n-1$.

Mandelstam invariants are products of momenta $p_i p_j$. This product can be translated to representation language to the tensor product of standard [n-1, 1] representations



which produces the singlet irrep [n] in addition to another [n-1,1] irrep and the irrep [n-2,2]. The contribution of the singlet [n] vanishes due to momentum conservation, while the second leads to vanishing contributions due to the on-shell conditions $p_{\mu}p^{\mu} = 0$. We can therefore associate the Mandelstam invariants consisting of $p_i p_j$ to the irrep [n-2,2].

Now that we have identified the irrep corresponding to Mandelstam invariants, we note that the kinematic numerators N can be made out of any combinations of (different) powers pof these Mandelstam invariants. We therefore also need to find how the tensor product of [n-2, 2] irreps decomposes.

Before we construct the tensor product of Mandelstam irreps, we note that there are certain numerators that satisfy the BCJ conditions, but which result in vanishing amplitudes. These were classified as 'gauge solutions' and have in common the property that they can be written as the product of Mandelstam variables $s_{i...j}$ and a specific irrep of S_n . The shape of this gauge irrep depends on the multiplicity of the amplitude. The example is given that at n = 4 the gauge numerator is denoted as

$$N_{abcd} = s_{ab}G_{abcd},\tag{2.11}$$

where G_{abcd} is a fully anti-symmetric tensor which transforms according to the adjoint [1, 1, 1, 1] irrep of S_n . For n = 4 through n = 7 the irreps that give rise to a vanishing amplitude are denoted in Table 2.2.

2.1.3 Powers of Mandelstam Invariants

The construction of kinematic BCJ numerators has been translated into a representation theory problem by investigating the irreps that arise from tensor products of the [n-2, 2]

n	Gauge irreps
4	[1, 1, 1, 1]
5	[2, 2, 1]
6	[3, 2, 1], [3, 1, 1, 1]
7	[4,3], [4,2,1], [4,1,1,1], [3,2,2], [3,2,1,1], [3,1,1,1,1], [2,2,2,1]

Table 2.2: Irreps of S_n that give rise to a vanishing (gauge) contribution to the amplitude [62].

irreps. This product can be taken p times, corresponding to taking a Mandelstam variable to the power p:

$$s^{p} \cong \left(\bigsqcup^{n-2} \right)^{p} = \bigsqcup^{n-2} \otimes \cdots \otimes \bigsqcup^{n-2} \otimes \cdots \otimes \bigsqcup^{n-2}.$$
(2.12)

The specific calculation of these irrep products starts at n = 4 due to the fact that there exist no Mandelstams for n = 3 scalars. For n = 4 and n = 5 the decomposition of the product has been shown to obey a systematic classification, as we will see shortly. However, this classification does not prove useful for $n \ge 6$ due to intricate relations between invariants referred to as syzygys² [63, 64]. Nevertheless, by imposing that higher multiplicity amplitudes are bootstrappable from 4-point interactions it remains possible to characterize these amplitudes through powers of Mandelstam invariants.

2.2 Numerators for 4-point Interactions

2.2.1 Linear Order s^1 BCJ parameter

Out of the four external momenta p_i there are only 3 independent momenta. These momenta live in the [3, 1] representation of S_4 . The Mandelstam invariants then live in the [2, 2] irrep, which is the so-called "window" irrep. Upon comparison of this irrep to the irreps of S_4 that are compatible with the BCJ constraints in Table 2.1 it becomes clear that the [2, 2] irrep allows for BCJ compatible expression of Mandelstam invariants linear order s^1 . The combination of Mandelstams given by

$$N_4^{(1)} = N_{abcd}^{(1)} = s_{bc} - s_{ac} = t - u, (2.13)$$

is linear in Mandelstam variables and satisfies the same BCJ constraints as the colour factors. At the level of the amplitude, this kinematic numerator will give rise to exchange

²A mathematical description of syzygys goes beyond the scope of this work, but a heuristic argument of such relations would be that, as you increase the number of momentum vectors at some point not all of the vectors are linearly independent. The point at which this happens would depend on the number spacetime dimensions D that these momenta live in. When the momenta are no longer linearly independent there are relations between momenta that, in the context of Mandelstams s_{ij} could result in complicated relations between invariants.

diagrams, it is therefore called the 'exchange factor'. This can be seen by the the fact that the amplitude $A \sim N/D$ will have $D \sim s^1$ for any trivalent 4 point amplitude. Explicitly, the s-channel exchange diagram will give rise to an amplitude proportional to $\frac{t-u}{s}$. We will explore the construction of amplitudes from the numerators we find in this section in more detail in section 2.5.

2.2.2 Quadratic Order s^2 BCJ Parameter

At quadratic order in Mandelstams, s^2 , we need to take the symmetric product of 2 [2, 2] irreps. This product decomposes as



Here the [1,1,1,1] irrep is again the gauge factor which does not contribute to the fourpoint amplitude. The [4] irrep is the the trivial representation which does not satisfy the BCJ conditions. This leaves us with another contribution by the [2,2] window again. We already saw that this is compatible with the BCJ conditions, therefore we can denote

$$N_4^{(2)} = s_{ab}(s_{bc} - s_{ac}) = s(t - u)$$
(2.15)

as the quadratic order factor. Note that it is the same as the exchange factor of (2.13) multiplied by a propagor of the half ladder diagram. It therefore gives rise to contact interactions at the level of the amplitude as we will see in section 2.5.

2.2.3 Higher-Order Mandelstam Invariants: Series Expansions

At higher orders in p the [2, 2] window irrep will continuously appear as the only contributing irrep in the decomposition of the product of irreps. For this reason, a pattern occurs; the BCJ factor N can be written as the product of an invariant Mandelstam expression and one of the two building blocks $N_4^{(1)}$ and $N_4^{(2)}$.

It is not immediately clear how often the BCJ irrep will appear in the irrep decomposition. We can use the Molien series given by

$$M(x) = \frac{1}{n!} \sum_{g \in S_n} \det \left(1 - xg | V^* \right)^{-1}, \qquad (2.16)$$

to count the number of invariants for a specific order [65]. Here the term 1/n! comes from the cardinality of S_n , which is n!. We therefore sum over the determinants of the (1 - xg)for some complex representation V of S_n . The expression will result in a product of $factors(1 - x^{b_i})$ in the denominator due to the inverse determinant,

$$M(x) = \frac{1}{(1 - x^{b_1}) \cdots (1 - x^{b_r})},$$
(2.17)

which can be expanded using the properties of the geometric series to obtain a polynomial whose coefficients a_i correspond to the number of independent symmetric polynomials in n variables.

$$M(x) = \sum_{n=0}^{\infty} a_{i,n} x^n = a_0 + a_1 x + a_2 x^2 + \dots$$
 (2.18)

This then corresponds to the number of invariants of S_n at a specific order in irrep products.

Returning to our 4-point interactions, for S_4 the Molien series is denoted as

$$M_4(x) = \frac{1}{(1-x^2)(1-x^3)}.$$
(2.19)

From (2.19) we can already heuristically infer that there are two 'primary' invariants of order x^2 and x^3 out of which higher order invariants can be made. These are primary, because when M(x) is expanded through use of a geometric series (2.18), we will find new invariants at higher orders in primary invariants

$$M_4(x) = \frac{1}{(1-x^2)(1-x^3)} = (1+a_2x^2+a_4x^4+a_6x^6+\dots)(1+b_3x^3+b_6x^6+\dots)$$

$$(2.20)$$

$$= 1+a_2x^2+b_3x^3+a_4x^4+a_2b_3x^5+(a_6+b_6)x^6+\dots$$

$$(2.21)$$

The two primary invariants are referred to as the 'quadratic' and 'cubic' primary invariants, $I_4^{(2)}$ and $I_4^{(3)}$ respectively. In terms of Mandelstams, these are given by

$$I_4^{(2)} = s_{ab}s_{bc} + s_{ac}s_{bc} + s_{ab}s_{ac} = st + ut + su, (2.22)$$

$$I_4^{(3)} = s_{ab}s_{ad}s_{ac} = stu. (2.23)$$

Specifically, the number of times the window irrep [2, 2] occurs at any order in powers of Mandelstams is given by [64]

$$H_4^{BCJ}(x) = (x + x^2) M_4(x) = \frac{(x + x^2)}{(1 - x^2)(1 - x^3)},$$
(2.24)

which in turn implies that all BCJ invariants are written as $N_4^{(1)}$ or $N_4^{(2)}$ multiplied by any combination of primary invariants $I_4^{(2)}$ and $I_4^{(3)}$.

2.2.4 Gauge Parameters at 4-point

Similar to the BCJ window irrep factor, we can calculate the number of times the gauge parameter irrep [1, 1, 1, 1] occurs in the expansion. This is given by the Hilbert series

$$H_4^{\text{Gauge}}(x) = x^3 M_4(x) = \frac{x^3}{(1-x^2)(1-x^3)}.$$
 (2.25)

The coefficients of the expansion then count the number of gauge parameters at one order higher. In this Hilbert series we recognise the two primary invariants and a cubic secondary invariant. Explicitly, the higher order combination

$$2N_4^{(2)}I_4^{(2)} - 3N_4^{(1)}I_4^{(3)}, (2.26)$$

is of the order of s^4 . In terms of Mandelstams this term can be written as

$$2N_4^{(2)}I_4^{(2)} - 3N_4^{(1)}I_4^{(3)} = 2s(t-u)(st+ut+su) + 3(t-u)(stu)$$

= $s(t-u)(5tu+2s(t+u)).$ (2.27)

This term satisfies the BCJ conditions, but when we construct amplitudes from this numerator we will find that it vanishes.

2.3 Numerators for 5-point Interactions

The Mandelstam invariants live in the [3, 2] irrep. A quick glance at Table 2.1 reveals that there are therefore no BCJ compatible irreps at the linear order in mandelstam for interactions of 5 scalar legs. At quadratic order we find the decomposition of the product of Mandelstams [66]

$$[3,2] \otimes [3,2] \cong [5] + [4,1] + 2[3,2] + [3,1,1] + 2[2,2,1] + [2,1,1,1].$$
(2.28)

Therefore the matching BCJ irrep is 'hook' tableau[3, 1, 1], which is present in the decomposition. It turns out that the gauge parameter irrep [2, 2, 1] occurs twice in this decomposition. For higher order invariants the Hilbert series are given by

$$H_5^{\text{Inv}}(x) = \left(1 + x^6 + x^7 + x^8 + x^9 + x^{15}\right) / D_5(x)$$

$$U_{\text{BCI}}^{\text{BCI}}(x) = \left(-3 + 2 \cdot \frac{4}{3} + 4 \cdot \frac{5}{3} + 5 \cdot \frac{6}{3} + 6 \cdot \frac{7}{3} + 6 \cdot \frac{8}{3} + 5 \cdot \frac{9}{3} + 4 \cdot \frac{10}{3} + 2 \cdot \frac{11}{3} + \frac{12}{3}\right) / D_5(x)$$
(2.29)

$$H_5^{\text{Loss}}(x) = \left(x^5 + 2x^4 + 4x^5 + 5x^5 + 6x^7 + 6x^5 + 5x^5 + 4x^{10} + 2x^{11} + x^{12}\right)/D_5(x)$$
(2.30)

$$H_5^{\text{Gauge}}(x) = (x^2 + x^3 + 3x^4 + 3x^5 + 3x^6 + 4x^7 + 4x^8 + 3x^9 + 3x^{10} + 3x^{11} + x^{12} + x^{13})/D_5(x)$$
(2.31)

$$D_5(x) = (1 - x^2) (1 - x^3) (1 - x^4) (1 - x^5) (1 - x^6).$$
(2.32)

As before, the denominator terms $(1 - x^{b_i})$ indicate the presence of primary invariants and the numerator terms point to the presence of secondary invariants. Therefore at 5-point

there are primary invariant from quadratic order to quintic order for H_5^{Inv} . The sextic order invariant appears in both the denumerator and numerator, pointing to the syzygys at this order that were mentioned before.

Such relations also occur for H_5^{BCJ} , but it is stated that there are 29 independent [3, 1, 1] hooks which generate all BCJ compatible solutions. Other hooks that appear in the Molien series can be written as products of these independent hooks with primary and secondary inariants. For H_5^{Gauge} it is argued that due to the fact that the [2, 2, 1] irrep shows up multiple times in the decomposition of the product of BCJ irreps, the number of distinct irreps at one order higher is given by:

$$H_5^{\text{Gauge, dist}} = \left(x^3 + x^4 + 2x^5 + 3x^6 + 2x^7 + 3x^8 + 4x^9 + 3x^{10} + 2x^{11} + 2x^{12} + x^{13}\right) / D_5(x).$$
(2.33)

The resulting series for the number of physical BCJ parameters at every order is then given by

$$H_5^{\text{Phys}}(x) = \frac{(x^4 + 2x^5 + 2x^6 + 4x^7 + 3x^8)}{D_5(x)} + \frac{(x^9 + x^{10})}{(1 - x^2)(1 - x^4)(1 - x^5)(1 - x^6)}, \quad (2.34)$$

which presents 14 different hook structures, out of which 13 are independent as one of them can be written as the product of a secondary invariant. These 13 hook structures can be used as seed interactions for 5-point amplitudes.

2.4 Numerators for 6-point and Higher: BCJ Bootstrap

The Molien series methodology of 4-point and 5-point interactions can in theory be used to construct the independent BCJ structures for 6-point interactions, but in practice this is a cumbersome approach. For starters, there are 3 possible BCJ structures are 6-point and their Molien series contain terms up to $\mathcal{O}(x^{24})$ in the numerator. This would require explicit irrep product decompositions of these secondary invariants with products of 24 BCJ irreps. While the calculation of the irreps is cumbersome, the counting of the parameters was done by [19] and is shown in Table 2.3.

Alternatively, it is much more convenient to consider the 6-point amplitudes which can be constructed out of 4-point seed interactions similar to the approach of [67]. This utilises the factorization property of the amplitude discussed in section 1.3, which can be schematically depicted as

$$A_{6} = \sum_{s_{abc}} b \xrightarrow{q} f e \implies A = \sum_{s_{abc}} A_{L} \frac{1}{s_{abc}} A_{R}.$$
(2.35)

p	BCJ	Gauge	Phys	Inv	Bootstrap
1	1	0	1	0	0
2	3	0	3	1	1
3	9	1	8	2	2
4	23	5	18	4	3
5	54	16	38	6	8
6	121	42	79	13	24
7	246	95	151	19	53

Table 2.3: Overview of the classification of 6-point irrep counting for the BCJ, gauge, physical and bootstrap-compatible parameters up to $\mathcal{O}(s^7)$ [62].

Such a factorisation condition can then be imposed on the 6-point numerator, giving rise to the following condition for the BCJ numerator at order s^p

$$\lim_{s_{abc}\to 0} N_6^{(p)} = \sum_q c_q N_4^{(p-q)}(abcx) N_4^{(p+q)}(xdef),$$
(2.36)

where the summation runs over all possible factorization combinations as s^p can be written as a product of s^{p-q} with s^q , c_q is some coefficient specific to the channel and x denotes the internal propagator.

2.4.1 Quadratic Order Numerator

The first possible 6-point BCJ factor will consist of a product of two linear $N_4^{(1)}$ and is therefore of $\mathcal{O}(s^2)$ denoted by:

$$\lim_{s_{abc} \to 0} N_6^{(2)} = N_4^{(1)} N_4^{(1)} = (s_{ac} - s_{bc})(s_{de} - s_{df}).$$
(2.37)

The general numerator that satisfies this factorization is the above expression plus an expression of the form s_{abc} multiplied by a factor L_{abcdef} that is linear in all 15 possible Mandelstams. Later when we calculate our amplitudes we can express these into the basis of 9 independent Mandelstam invariants. We can make a generic ansatz for such a linear factor as

$$L_{abcdef} = c_1 s_{ab} + c_2 s_{ac} + \dots + c_{14} s_{df} + c_{15} s_{ef}.$$
(2.38)

We then require that L_{abcdef} satisfies the generalised Jacobi identity conditions of (2.3) which impose

$$-L_{abcdef} = L_{eabcdf} - L_{ebacdf} - L_{ecabdf} + L_{ecbadf}$$

$$(2.39)$$

$$-L_{edabcf} + L_{edbacf} + L_{edcabf} - L_{edcbaf}, \qquad (2.40)$$

$$-L_{abcdef} = L_{fedcba}.$$
(2.41)

This system of equations can be solved the coefficients c_i , which reduces the system to one overall coefficient, e.g. c_5 , for four of the 15 Mandelstams while the rest of the coefficients vanish. This leads to the reduced expression for the linear term

$$L_{abcdef} = -c_5(s_{ae} - s_{af} - s_{be} + s_{bf}).$$
(2.42)

Through this term, the quadratic 6-point amplitude is thereby uniquely fixed with the numerator

$$N_6^{(2)} = (s_{ac} - s_{bc})(s_{de} - s_{df}) + \frac{1}{2}(s_{ae} - s_{af} - s_{be} + s_{bf}).$$
(2.43)

2.4.2 Cubic Order Numerator

Next, we turn our attention to the numerator at $\mathcal{O}(s^3)$. While $N_6^{(2)}$ is unique, this is not the case for $N_6^{(3)}$ and $N_6^{(4)}$ as is shown in Table 2.3. In this case there are two possible combinations that contribute through factorization, explicitly

$$\lim_{s_{abc}\to 0} N_6^{(3)} = c_1 N_4^{(2)} N_4^{(1)} + c_2 N_4^{(1)} N_4^{(2)} = c_1 (s_{ac} - s_{bc}) s_{ab} (s_{de} - s_{df}) + c_2 (s_{ac} - s_{bc}) (s_{de} - s_{df}) s_{ef}.$$
(2.44)

The solution for $N_6^{(3)}$ will be the above factor plus s_{abc} multiplied by some factor that is a linear combination of quadratic Mandelstams denoted by Q_{abcdef} . We can construct a generic ansatz for Q_{abcdef} as a sum over all possible products of s_{ij}

$$Q_{abcdef} = c_{1,1}s_{ab}s_{ab} + c_{1,2}s_{ab}s_{ac} + \dots + c_{15,15}s_{ef}s_{ef}, \qquad (2.45)$$

where the coefficients are related to elements of our basis of Mandelstams $\{s_{ab}, s_{ac}, \ldots, s_{ef}\}$ are multiplied together.

After constraining the ansatz with the generalised Jacobi identities of (2.3) we find that there are a total of 3 unfixed parameters, $c_{5,9}$, $c_{6,6}$ and $c_{6,8}$. We can further constrain this equation by imposing a soft limit of $\sigma = 0$ or $\sigma = 1$ on the amplitude.

The final generalised numerator $N^{(3)}_{abcdef}$ with $\sigma = 0$ imposed is then given by the expression

$$N_{abcdef}^{(3)} = (s_{ac} - s_{bc})s_{ab}(s_{de} - s_{df}) + (s_{ac} - s_{bc})(s_{de} - s_{df})s_{ef} + s_{abc}Q_{abcdef}.$$
 (2.46)

These numerators can then be used to construct amplitudes, as we will see shortly in section 2.5.

2.4.3 Quartic and Higher Order Numerators

Finally at $\mathcal{O}(s^4)$ the factorization can be denoted as

$$\lim_{s_{abc} \to 0} N_6^{(4)} = c_1 N_4^{(3)} N_4^{(1)} + c_2 N_4^{(2)} N_4^{(2)} + c_3 N_4^{(1)} N_4^{(3)}.$$
(2.47)

We recall that the BCJ numerators can be constructed out of linear and quadratic invariant building blocks, therefore higher order numerators such as $N_4^{(3)}$ can be constructed from the building blocks $I_4^{(2)}$ and $I_4^{(3)}$. However it is stated that higher order combinations only factorize in the combinations $N_4^{(1)}(I_4^{(3)})^n$ and $N_4^{(2)}I$, where I is an invariant of arbitrary power. It has not been verified that this behaviour continues at higher order in Mandelstam. For this reason the terms with $N_4^{(3)} = N_4^{(1)}I_4^{(2)}$ do not contribute to the numerator of the quartic order. The resulting expression is therefore

$$\lim_{s_{abc} \to 0} N_6^{(4)} = N_4^{(2)} N_4^{(2)} = c_2 (s_{ac} - s_{bc}) s_{ab} (s_{de} - s_{df}) s_{ef}.$$
 (2.48)

We can use a similar method to finding the full expression of the numerator as for the cubic numerators; starting with an ansatz that is cubic in Mandelstam invariants and then constraining the coefficients of this ansatz using the BCJ conditions and imposing the soft degree of $\sigma = 1$ to fully determine the theory.

We have constructed the numerators $N_6^{(2)}$, $N_6^{(3)}$ and $N_6^{(4)}$, which contain the three possible combinations of $N_4^{(1)}$ and $N_4^{(2)}$. Due to the property that the higher-order numerators at 4-point are built up out of these building blocks, we can construct higher-order $N_6^{(p)}$ recursively.

2.5 Colour x Kinematics: The Gauged NLSM

Now that we have found expressions for kinematic numerators at 4- and 6-point that satisfy the BCJ constraints, we return to the starting point of the discussion. The aim is to use the constructed numerators to build amplitudes. Using the amplitude decomposition

$$A_n = \sum_{\text{trivalent}} \frac{NN}{D},\tag{2.49}$$

amplitudes for different theories can be considered by filling in N and \tilde{N} with colour and (or) kinematic numerators.

We will consider the single copy amplitude for a gauged scalar field, where one numerator describes a colour factor C and the other a kinematic numerator N. The scalar Gold-stone field ϕ^a transforms according the adjoint representation a of gauge group G, whose generators give rise to the colour factor C.

We will see that by imposing the soft limit $\sigma_{min} = 0$ and $\sigma_{max} = 1$ at 6-point, the resulting amplitudes describe amplitudes of the gauged NLSM pions interacting with gluons.

2.5.1 4 Point Interactions

Starting with the 4-point kinematic numerator in terms of (2.13) and (2.15) we see that the total amplitude of the theory separates into two terms of different powers in Mandelstam

invariants

$$A_4^{\text{GNLSM}} = \sum_{\text{trivalent}} \frac{C_4 N_4}{D} = \sum_{\text{trivalent}} \frac{C_4 N_4^{(1)}}{D} + \sum_{\text{trivalent}} \frac{C_4 N_4^{(2)}}{D}.$$
 (2.50)

We will treat these terms seperately, beginning with the term 'linear term' generated by $N_4^{(1)}$. This term generates the three possible 4-point gluon exchange channels:

$$\sum_{rivalent} \frac{C_4 N_4^{(1)}}{D} = C_4^s \frac{(t-u)}{s} + C_4^t \frac{(u-s)}{t} + C_4^u \frac{(s-t)}{u}, \qquad (2.51)$$

with $C_4^s = \tilde{f}^{abx} \tilde{f}^{xcd}$, $C_4^t = \tilde{f}^{bcx} \tilde{f}^{xad}$ and $C_4^u = \tilde{f}^{cax} \tilde{f}^{xbd}$ satisfying the usual Jacobi identity for colour factors. The Feynman diagrams depicting this exchange are shown in Figure 2.2.



Figure 2.2: The three distinct gluon exchange channels produced by $N_4^{(1)}$.

The derivative dependence of the amplitudes can be used to determine a Lagrangian that is able to produce these exchange diagrams. Firstly, the numerator scales as $N_4^{(1)} \sim s^1 \sim p^2$, which indicates the contribution of a $(\partial \phi)^2$ term. The interaction Lagrangian that is able to produce the diagrams of Figure 2.2 can then be denoted as

$$\mathcal{L}_{int}^{(1)} \sim f^2 \phi(\partial \phi) A_{\mu}.$$
 (2.52)

Since we are considering a theory of gauged scalars, we will later use the covariant derivative $(D_{\mu}\phi)_{a} = \partial_{\mu}\phi_{a} - igf^{abc}A^{b}_{\mu}\phi^{c}$ in order for the total Lagrangian to be gauge invariant.

Equation (2.51) can be rewritten by using t = -s - u. After this, we can set $p_a \to \lambda p_a$, recalling that $s = 2p_a p_b$ and $u = 2p_a p_c$ to find

$$-\frac{C_4^s\lambda(s+2u)}{\lambda s} + \frac{C_4^t\lambda(s-u)}{\lambda(s+u)} + \frac{C_4^u\lambda(2s+u)}{\lambda u},$$
(2.53)

where it is clear that the dependence on λ cancels. This indicates a soft degree of $\sigma = 0$. Identical analysis can be done for the other external legs.

Next, we turn our attention the quadratic numerator $N_4^{(2)}$. The contribution to the total amplitude in (2.50) generated by this numerator is given by

$$\sum_{i} \frac{C_4^i N_4^{(2)}}{D_i} = C_4^s \frac{s(t-u)}{s} + C_4^t \frac{t(u-s)}{t} + C_4^u \frac{u(s-t)}{u}$$
(2.54)

$$= C_4^s(t-u) + C_4^t(u-s) + C_4^u(s-t).$$
(2.55)

Remarkably, the exchange channels have been removed through the introduction of the addition order in Mandelstams, leaving us with only contact interactions. The Feynman diagram for such a contact term is denoted in Figure 2.3.



Figure 2.3: The contact term produced by $N_4^{(2)}$. The other two contact terms can be found through cyclic permutations of a, b and c.

The introduction of the additional factor of s_{ab} that cancels the gluon exchange in the amplitude indicates an extra derivative coupling at the Lagrangian level. From this, we can deduce that the contact interaction term can be associated with the interaction term

$$\mathcal{L}_{int}^{(2)} \sim f^2 \phi^2 (\partial \phi)^2, \qquad (2.56)$$

which corresponds to the a familiar 4 point interaction of the NLSM, with the addition of structure constants that indicate the colour structure similar to (1.41). Further confirmation of NLSM-like behaviour can be seen from from the fact that the amplitude contains the Adler zero property, as the amplitude generated by $N_4^{(2)}$ is linear in all four momenta p_i , indicating a soft degree of $\sigma = 1$.

Both scalar interactions can be combined into the total Lagrangian

$$\mathcal{L}_4^{\text{GNLSM}} = -\frac{1}{2} (D\phi)^2 - \frac{1}{4} F^2 + \frac{1}{6} f^2 \phi^2 (D\phi)^2, \qquad (2.57)$$

which includes the free Lagrangians of both the adjoint scalars and the gauge fields. This theory then satisfies gauge invariance, the nonlinear shift symmetry of the NLSM and produces CK-duality satisfying amplitudes at 4-point. The specification *at four point* will become apparent when we analyse the the 6-point amplitudes, where we will need to add additional terms in order to produce a theory that satisfies these same properties, which is familiar from Equation 1.5.3. The difference here is that only a finite number of correction terms are needed.

2.5.2 6-Point Interactions

We can go one step further and describe the amplitudes generated by the six-point numerators of section 2.4. The amplitude can then be constructed as

$$A_6^{\text{GNLSM}} = \sum_{\text{trivalent}} \frac{C_6 N_6}{D} = \sum_{\text{trivalent}} \frac{C_6 \left(N_6^{(2)} + N_6^{(3)} + N_6^{(4)} \right)}{D}.$$
 (2.58)

When we are considering cubic interactions, there are two types of diagram structures that can occur at 6-point. The first is the half-ladder diagrams we encountered in our discussion of gluon amplitudes and the second are 'snowflake' diagrams, both types of diagrams are displayed in Figure 2.4. There are a total of $(2 \cdot 6 - 5)!! = 105$ possible cubic diagrams at 6-point. Out of these, there are 90 half-ladder diagrams and 15 cubic diagrams [60]. The snowflake diagrams have distinctly different amplitude contributions. For starters, both



Figure 2.4: Half-ladder and snowflake diagram for a 6-point amplitude.

types of diagrams have denominators $D \sim s^3$, but they are made up of different interal propogators. The half-ladder diagrams will have propagators of the form

$$D_{\rm HL} = \frac{1}{s_{ab}s_{abc}s_{abcd}},\tag{2.59}$$

where it is possible to use momentum conservation to relate $s_{abcd} = s_{ef}$. The snowflake diagrams have denominators of the form

$$D_{\text{snowflake}} = \frac{1}{s_{ab}s_{cd}s_{ef}}.$$
(2.60)

Secondly, the colour factors for both diagram types are product of 4 structure constants, e.g. $C_6^{HL} = \tilde{f}^{abx} \tilde{f}^{xcy} \tilde{f}^{ydz} \tilde{f}^{zef}$ and $C_6^{snowflake} = \tilde{f}^{abx} \tilde{f}^{xyz} \tilde{f}^{ycd} \tilde{f}^{zef}$ for the diagrams in Figure 2.4.

Finally, we turn our attention to the numerators of the two diagrams. Initially, one might think that our assumption of the factorization of the numerator N_6 into a product of four-point numerators N_4N_4 does not allow for snow-flake diagrams, as these cannot be factorised in this way. However, the numerators of the snowflake diagrams are related to the numerators of the half-ladder diagrams through Jacobi identities by

$$N_{abcdef}^{\text{snowflake}} = N_{abcdef}^{\text{half-ladder}} - N_{abdcef}^{\text{half-ladder}}, \qquad (2.61)$$

The total amplitude can then be said to be a sum over half-ladder and snowflake diagrams

$$A_{6} = \sum_{\text{half-ladder}} \frac{N_{abcdef} C_{abcdef}^{\text{hl}}}{s_{ab} s_{abc} s_{ef}} + \sum_{\text{snow-flake}} \frac{(N_{abcdef} - N_{abdcef}) C^{\text{snowflake}}}{s_{ab} s_{cd} s_{ef}}.$$
 (2.62)

Now that we have determined how to construct amplitudes from the numerators, we will investigate the properties of these amplitudes. The quadratic numerator gives rise to amplitudes of the form

$$\sum_{\text{trivalent}} \frac{C_6 N_6^{(2)}}{D},\tag{2.63}$$

which are completely fixed by the BCJ conditions on the numerators. In terms of degree of Mandelstams these amplitudes scale as s^{-1} due to the cubic propagators of (2.59) and (2.60). Due to the pole structure of these amplitudes, we can identify them with gluon exchange diagrams similar to Figure 2.2.

The cubic numerators had unfixed constraints after requiring factorization, but these are fixed by imposing the soft degree $\sigma = 0$ on the amplitudes.

Finally, the quartic numerators $N_4^{(3)}$ are completely fixed by imposing the soft degree $\sigma = 1$, leading to the 6-point NLSM pion contact amplitude, whose diagram is shown in Figure 2.5.



Figure 2.5: The 6-piont GNLSM contact diagram produced by $N_6^{(4)}$.

As indicated before, we need to add addition terms to the Lagrangian in order to generate the above amplitudes. Requiring the conditions of gauge inariance, nonlinear shift symmetry and BCJ compatibility, the Lagrangian can be denoted as

$$\mathcal{L}_6 = \mathcal{L}_4 + \frac{1}{45} f^4 \phi^4 (D\phi)^2 - 2f^2 F^3 + \frac{1}{6} f^2 \phi^2 F^2.$$
(2.64)

The coefficients of these additional terms are tuned to satisfy the symmetry conditions.

The new terms include the higher derivative F^3 operator discussed in Equation 1.5.3, the six-pion contact term $\phi^4(\partial \phi)^2$ and a new type of interaction between gluons and pions through $f^2 \phi^2 F^2$. This term generate a 5-point interaction of the form $\phi^2(\partial A)\phi$ that does not contribute to the 6-point amplitude, a 6-point amplitude of the form $\phi^2 A^4$ that can not contribute to a 6-point amplitude with only scalar external legs at tree level, and a 4-point interaction of the form $\phi^2(\partial A)^2$. This last interaction can contribute to the amplitudes through diagrams of the form shown in Figure 2.6. However, these amplitudes do not conform to the requirement of factorising in 4-point amplitudes on $s_{abc} \to 0$.



Figure 2.6: The additional 6-point interaction type generated by the interaction $f^2\phi^2 F^2$ that contributes to A_6 .

2.5.3 Higher Multiplicities

Moving on to higher-point amplitudes, it is expected that the above pattern continues; the Lagrangian will consist of the lower point Lagrangians plus additional terms. This is still subject to current research by Yang Li and colleagues and could potentially be an interesting thesis topic for future students reading this work.

As we have seen at 4- and 6-point, the numerator with lowest order in Mandelstam generates the gluon exchange channel and the highest order in Mandelstam generates the pion contact term (after imposing the soft limit). For general (even) multiplicity n, it is expected that the numerator of $\mathcal{O}(s^{(n-2)/2})$ will generate the gluon exchange channels, while the term with the highest order in derivatives, which is $\mathcal{O}(s^{n-2})$ generates the pion contact term. The expected pattern of degrees of s^p is depicted in Table 2.4.



Table 2.4: Schematic depiction of the amplitudes generated at every order in Mandelstam s^p . The gluon exchange (GE) and pion contact (PC) terms follow the pattern of the soft degree of the amplitude. The gluon exchange amplitudes all obey $\sigma = 0$ while the pion contact terms obey the Adler zero of $\sigma = 1$. The amplitudes with ambiguous soft degree are in between these lines, and are denoted by x's.

We would like to emphasise the fact that the above Lagrangian describes a theory of gauged NLSM pions amplitudes with a *hybrid soft degree*. This is quite unusual compared

to 'regular' theories such as YM and the NLSM, where the entire theory has a singular fixed soft degree. Instead, the theory described by (2.64) has a minimum soft degree of $sigma_{min} = 0$, while a section of the theory has $\sigma = \sigma_{min} + 1$.

2.6 Key Findings and Implications

The initial goal of this chapter was to find a general expression for colour-kinematics (CK) duality satisfying numerators in scalar theories. We explored the existence of such numerators from a group theory perspective and successfully constructed the 4-point numerators, which served as the building blocks for higher-multiplicity numerators. In the final sections, we investigated the amplitudes arising from the single copy of these numerators.

These amplitudes can be interpreted as those of the Gauged Nonlinear Sigma Model (GNLSM) with lower derivative corrections, generated by a theory exhibiting a hybrid soft degree. The theory and the resulting amplitudes are highly non-trivial, and the correction terms required in the Lagrangian are also uncommon.

This raises an interesting question about the uniqueness of these amplitudes and whether alternative Lagrangians could yield similar results. Addressing this question is key to better understanding the underlying structures and seeing if there are other theories that can produce the amplitudes we discussed in this chapter.

To further our investigation, we will now turn our attention to a different method for generating CK-duality satisfying amplitudes: Unifying Relations. This approach enables the generation of amplitudes for various interacting fields, including biadjoint scalars, pions, and gluons. In the following chapter, we will thoroughly explore this subject to determine if and how we can construct amplitudes akin to those produced by the BCJ bootstrap method discussed here. This exploration will provide further insights into the versatility and applicability of the CK-duality framework in scalar theories.

Chapter 3

Unifying Relations - Transmutation Operators

The approach of this chapter is based on the concept of 'Transmutation Operators', which was initially proposed by Clifford Cheung, Chia-Hsien Shen and Congkao Wen in [22]. In this work, the authors proposed a set of differential operators which 'transmute' tree-level scattering amplitudes of one type of theory, (a), into amplitudes of a different theory, (b). In an abstract schematic sense we can depict this as follows:

$$\mathcal{T} \cdot A_a \to A_b. \tag{3.1}$$

With these operators, a web of theories related through transmutation of amplitudes can be created. At the heart of this unified web are the scattering amplitudes of *extended gravity*, which describes gravitons coupled to a dilaton and two-form¹. By applying different types of transmutation on these amplitudes, we can derive the amplitudes of several theories, including:

- Yang-Mills (YM): Gluon self interaction as describe in section 1.1.
- Maxwell: Photons described by Maxwell's equations.
- Born-Infeld (BI): An effective field theory describing a nonlinear generalisation of electromagnetism.
- Special Galileons (SG): Derivatively coupled scalar effective field theory characterised by an enhanced soft limit of $\sigma = 3$.
- Dirac-Born-Infeld (DBI): Derivatively coupled scalar effective field theory characterised by an enhanced soft limit of $\sigma = 2$.
- Nonlinear sigma-model (NLSM): Pion self interactions as described in section 1.2.

¹The tensor product of two polarisation $e^i_{\mu}e^j_{\nu}$ can be decomposed into three combinations representing a graviton polarisation $e^h_{\mu\nu}$, a B-field polarisation $e^B_{\mu\nu}$ and the dilaton polarisation $e^{\phi}_{\mu\nu}$. Respectively, these are the symmetric-traceless, the antisymmetric, and the scalar part of the polarisation tensor. The theory describing these field coupled together is sometimes referred to as $\mathcal{N} = 0$ supergravity.

• Biadjoint scalar (BAS) theory: Scalars with two colour indices that arises in the zeroth copy of YM and GR in subsection 1.5.2.

During the process of transmuting amplitude we can find 'hybrid' or 'mixed' amplitudes of all of the above theories, for example

- Einstein-Yang-Mills (EYM); describing gluons coupled to (extended) gravitons,
- Einstein-Maxwell (EM); photons coupled to (extended) gravitons
- Yang-Mills-Scalar (YMS): gluons coupled to scalars

We will extensively discuss the construction of transmutation operators, starting with a discussion on the necessary conditions and the basis of operators that arise from this. Using this basis, we create specific combinations of operators that take us to the amplitudes we desire, e.g. pion amplitudes from gluon amplitudes and amplitudes of scalars interacting with gluons. We will use the 3- and 4-point gluon amplitudes as an example of this process. Finally, we discuss the transmutation of important properties of the amplitudes, such as the soft factors, factorisation, and UV completion.

While the Unifying Relations are incredibly useful relations for building the partial amplitudes of a wide web of theories, the physical meaning of transmutation is still unclear. Two of the operators we will construct, namely the 'trace' and the 'longitudinal' operator, can also be seen as a *dimensional reduction*, where the kinematics of the external legs of e.g. gluons can be constructed as higher-dimensional objects, which reduce to *D*-dimensional pions by construction. A discussion on this dimensional reduction can be found in section D.3.

3.1 Conditions and Basis for Transmutation Operators

The main reasoning in the construction of transmutation operators is that amplitudes are functions of Lorentz-invariant products of the external, on-shell, kinematic data; $e_i e_j$, $p_i e_j$ and $p_i p_j$ [40]. In this formulation we use a shorthand notation for summations over Lorentz-indices $e_i e_j = e_i \cdot e_j$ where $i \neq j$. With the idea that amplitudes are functions of these kinematic products, we can construct operators that take derivatives with respect to these objects. The next step is to consider that it is not very useful to build operators that transmute amplitudes into nonphysical amplitudes which violate momentum conservation and gauge invariance. Therefore there are two conditions these operators need to satisfy:

- 1. Transmutation operators need to preserve the on-shell kinematics.
- 2. Transmutation operators need to preserve the gauge invariance.

These conditions can also be translated into differential operators. It can be shown that a vanishing commutator with respect to these differential operators then guarantees conditions 1 and 2. Here we will simply define these operators, and the basis of transmutation

operators that follow from them. An in-depth discussion of how these operators satisfy the claims of condition 1 and 2 and how to form the basis can be found in Appendix D.

The momentum conservation can be enforced by defining a *total momentum operator* \mathcal{P}_{v} :

$$P_{v} \equiv \sum_{i} p_{i} v = (p_{1} + p_{2} + \ldots + p_{n})v, \qquad (3.2)$$

where the summation runs over all external legs i and v denotes any momentum or polarisation vector. The requirement for momentum conservation can then be cast into the following requirement:

$$[P_v, \mathcal{T}] = 0. \tag{3.3}$$

The condition of gauge invariance is translated into a differential operator through the Ward identity of (1.6). The differential Ward operator can therefore be defined as:

$$W_i \equiv \sum_{v} p_i v \partial_{v e_i}.$$
(3.4)

The summation over v runs over all external momentum and polarisation vectors in the amplitude. The polarisation of leg i appears in the amplitude in the Lorentz invariants $e_i p_j$ and $e_i e_j$. What this operator essentially does is replace every $e_i \rightarrow p_i$ wherever it appears in the amplitude, turning $e_i p_j \rightarrow p_i p_j$ and $e_i e_j \rightarrow p_i e_j$.

Any gauge invariant amplitude satisfies the Ward identity and is therefore annihilated by the Ward operator:

$$W_i \cdot A = 0. \tag{3.5}$$

Requiring that gauge invariance be preserved for a transmuted amplitude then implies a vanishing result when the commutator $[\mathcal{T}, W_i]$ acts on an amplitude:

$$[W_i, \mathcal{T}] = 0. \tag{3.6}$$

Starting from the most general form that a transmutation can take, we can apply our condition for momentum conservation and gauge invariance to constrain the general form to form basic elements for the transmutation operators. The operators take the form of first-order differential equations which act on the objects $e_i e_j$, $e_i p_j$ and $p_i p_j$. The general form can be written as:

$$\mathcal{T} \equiv \sum_{i,j} A_{ij} \partial_{p_i p_j} + B_{ij} \partial_{p_i e_j} + C_{ij} \partial_{e_i e_j}.$$
(3.7)

Here, A_{ij} , B_{ij} and C_{ij} are general functions of external kinematic data.

We then impose the constraint of momentum conservation upon this ansatz. In terms of transmutation operators this leaves a set of commuting operators which will form the basic building blocks for more complicated operators. The operators are denoted as

$$T_{ij} \equiv \partial_{e_i e_j},\tag{3.8}$$

$$I_{ijk} \equiv \partial_{p_i e_j} - \partial_{p_k e_j},\tag{3.9}$$

$$T_{ijkl} \equiv \partial_{p_i p_j} - \partial_{p_k p_j} + \partial_{p_k p_l} - \partial_{p_i p_l}.$$
(3.10)

These operators have the following symmetry properties:

$$T_{ij} = T_{ji},\tag{3.11}$$

$$I_{ijk} = -I_{kji},\tag{3.12}$$

$$T_{ijkl} = -T_{kjil} = T_{klij} = -T_{ilkj}.$$
 (3.13)

Next, we turn our attention to the constraint of gauge invariance. We can see that T_{ij} is intrinsically gauge invariant, as $[T_{ij}, W_k] = \sum_v [\partial_{e_i e_j}, p_i v \partial_{v e_i}] = 0.$

Furthermore, we can see that I_{ijk} is not intrinsically gauge invariant. However, this does not pose a problem for as it can be shown to be *effectively* gauge invariant. The commutator is $[I_{ijk}, W_l] = \delta_{il}T_{ij} - \delta_{kl}T_{jk}$, which can be combined with supplementary operators that allow the vanishing of the amplitude.

The third operator strips of pairs of polarisation vectors and will therefore give rise to an object with double poles in momenta. The object generated after transmutation is therefore not a physical scattering amplitude. Regardless, this operator does appear in the calculation of the subleading soft factor of Born-Infeld theory.

In the next sections, we will see how these operators are the building blocks to connecting the amplitudes of various theories in a unified web.

3.2 Trace and Insertion Operators

3.2.1 Trace Operator $T_{ij} = \partial_{e_i e_j}$

First we will turn our attention to the effect of one T_{ij} operator on the amplitude: $T_{ij} \cdot A$. What this effectively does is strip the amplitude of the polarisations e_i and e_j except for the combination $e_i e_j$ which transforms as $e_i e_j \rightarrow 1$. Therefore, it reduces the spin of particles *i* and *j* by one. Moreover, these particles are placed within a new 'dual colour' trace structure, hence the name 'trace operator'. This trace structure will become more apparent with the next operator I_{ijk} .

The reduction of spin by one unit turns spin-2 particles into spin-1 particles, which in turn can be turned into spin-0 particles by applying another trace operator. For example, the trace operator transmutes gluons into biadjoint scalars:

$$T_{ij} \cdot A(\cdots, g_i, g_j, \cdots) = A(\cdots, \phi_i \phi_j, \cdots).$$
(3.14)

We should take a moment to unpack the notation used in this equation. The scalars ϕ_i and ϕ_j that have been created carry the original colour index from their origin as gluons, plus an additional 'dual colour index' obtained from the transmutation. This is denoted through the use of commas to indicate sets of particles which are ordered according to this dual colour. The gluons carry only the single colour index and are therefore all separated by commas.

Finally, the ellipses \cdots denote any 'spectator' particles present in the scattering process which we leave untouched. Using this operator, at the level of the amplitude, we can transmute gravitons into photons, gluons into (biadjoint) scalars and BI photons into DBI scalars.

3.2.2 Insertion Operator: $I_{ijk} = \partial_{p_i e_j} - \partial_{p_k e_j}$

As mentioned before, the insertion operator I_{ijk} is not intrinsically gauge invariant. Luckily we can combine it with the trace operator to create a combination which is effectively gauge invariant. We calculate

$$[T_{ik} \cdot I_{ijk}, W_l] = \delta_{il} T_{ik} \cdot T_{ij} - \delta_{kl} T_{ik} \cdot T_{jk}, \qquad (3.15)$$

which vanishes upon hitting the amplitude because of the multi-linearity of the amplitude in e_i and e_k . By multi-linearity we mean that the polarisation e_i only exists in the combination of one pair $e_i e_l$ for example.

This transmutation reduces the spin of the particle j by one unit and *inserts* this particle between the particle i and k in the dual-colour trace structure. At the level of colourordered amplitudes. This is denoted as

$$I_{ijk} \cdot A(\cdots \phi_i \phi_k \cdots, g_j, \cdots) = A(\cdots \phi_i \phi_j \phi_k \cdots, \cdots).$$
(3.16)

This operator then extends the dual colour ordering of the resulting amplitude. Repeated application of the insertion operator allows us to transmute all the remaining gluons into scalars and place them in the dual colour trace.

3.2.3 Combination of Trace and Insertion Operators $\mathcal{T}[\alpha]$

The repeated use of the insertion operator I_{ijk} after initial application of a trace operator T_{ik} can be denoted by use of the following *single-trace* operator. The name is earned by creating a new trace-ordering $[\alpha]$ in the resulting amplitude, this operator is defined as

$$T[\alpha] \equiv T_{\alpha_1 \alpha_n} \cdot \prod_{i=2}^{n-1} I_{\alpha_{i-1} \alpha_i \alpha_n}.$$
(3.17)

Here it is important to note when dealing with a product of transmutation operators such as $T[\alpha]$ we should read this as the operators acting from left to right: first we apply the

trace operator to particles α_1 and α_n , then we insert particle α_2 between α_1 and α_n followed by particle α_3 between α_2 and α_n and so forth. Applied to a Yang-Mills amplitude we will find the following resulting amplitude

$$\mathcal{T}[i_1\cdots i_n]\cdot A(g_{i_1},\cdots,g_{i_n},\cdots) = A(\phi_{i_1}\cdots \phi_{i_n},\cdots).$$
(3.18)

Because this is an amplitude of biadjoint scalars interacting with gluons (denoted by the \cdots) through gauge interactions, the theory this amplitude corresponds to is called 'gauged biadjoint scalar theory'.

On the other hand, if α is the set of all gluons in the initial amplitude then all gluons are transmuted and the amplitude is purely a biadjoint scalar amplitude.

The operator can also be applied to extended gravity amplitudes to form amplitudes of Einstein Yang-Mills theory which describes gravitons coupled to gluons [68]:

$$\mathcal{T}[i_1\cdots i_n]\cdot A(h_{i_1},\cdots,h_{i_n},\cdots) = A(g_{i_1}\cdots g_{i_n},\cdots).$$
(3.19)

It is essential to note that the amplitudes that are created are the partial amplitudes of subsection 1.1.4, which do not contain the associated trace factor. As an example, the fourparticle graviton amplitude does not have any colour structure and, therefore, does not have a decomposition into colour-stripped partial amplitudes. When we apply the single trace operators for a to this amplitude we create the colour-ordered amplitude A[1234]

$$T[1234] \cdot A_{\rm G}(h_{i_1}, h_{i_2}, h_{i_3}, h_{i_4}) = A_{\rm YM}[1234], \qquad (3.20)$$

which in the trace decomposition of the full amplitude $\mathcal{A}_4^{\text{YM}}$ would be accompanied by the factor $\text{Tr}[T^{a_1}T^{a_2}T^{a_3}T^{a_4}]$. During the process of transmutation we do not add any colour factors, but we only act on and construct the (kinematic) partial amplitudes.

3.2.4 Multi-Trace Amplitudes

Naturally, the single-trace operator can be extended to multi-trace transmutation operators which generate multiple colour-trace structures in the resulting amplitudes. These operators and amplitudes are formed through the products of the operators $T[\alpha_i]$.

We denote an amplitude which is colour ordered with ordering β as $A(\beta)$. Such an amplitude can be constructed through the single trace operator of (3.17):

$$T[\beta] \cdot A = A(\beta). \tag{3.21}$$

Next we can introduce a dual colour-ordering by acting with $T[\alpha]$ on the amplitude such that

$$T[\alpha] \cdot A(\beta) = A(\alpha|\beta). \tag{3.22}$$

Furthermore we can act with multiple different trace operators to find

$$T[\alpha_1] \cdots T[\alpha_m] \cdot A(\beta) = A(\alpha_1, \dots, \alpha_m | \beta).$$
(3.23)

A particularly interesting combination of trace operators consist of ordering all particles into pairs by acting with $\mathcal{T}[i_1j_1]\cdots \mathcal{T}[i_mj_m] = \prod_{i,j \in pairs} T_{ij}$ on an amplitude such as a gluon amplitude:

$$\mathcal{T}[i_1 j_1] \cdots \mathcal{T}[i_m j_m] \cdot A(g_{i_1}, g_{j_1}, \cdots, g_{i_m}, g_{j_m}, \cdots) = A(\phi_{i_1} \phi_{j_1}, \cdots, \phi_{i_m} \phi_{j_m}, \cdots).$$
(3.24)

Due to the absence of insertion operators in this case, the dual colour ordering trace reduces to products of δ_{ij} 's corresponding to the pairs [ij] through the structure constant identity $\text{Tr}(T^aT^b) = \delta^{ab}$. This Yang-Mills amplitude is transmuted into a Yang-Mills-Scalar amplitude, describing gluons coupled to scalars.

3.2.5 Examples: Transmutation of YM into BAS and YMS aAplitudes.

Three Point Gluon Amplitude

As an example of the above transmutations, let us consider the three point colour-ordered partial Yang-Mills amplitude:

$$A(g_1, g_2, g_3) = \frac{1}{2}e_1e_2(p_2e_3 - p_1e_3) + \text{cyclic}(123), \qquad (3.25)$$

where the colour ordering $\{123\}$ is implied for the gluon amplitude.

We can apply T_{ij} to the partial amplitude in order to turn a pair of gluons into scalars:

$$T[12] \cdot A(g_1, g_2, g_3) = T_{12} \cdot A(g_1, g_2, g_3) = \frac{1}{2} \partial_{e_1 e_2} (e_1 e_2 (p_2 e_3 - p_1 e_3) + \text{cyclic}(123)),$$

= $A(\phi_1 \phi_2, g_3) = \frac{1}{2} (p_2 e_3 - p_1 e_3),$ (3.26)

$$T[23] \cdot A(g_1, g_2, g_3) = A(\phi_2 \phi_3, g_1) = \frac{1}{2}(p_3 e_1 - p_2 e_1), \qquad (3.27)$$

$$T[31] \cdot A(g_1, g_2, g_3) = A(\phi_3 \phi_1, g_2) = \frac{1}{2}(p_1 e_2 - p_3 e_2).$$
(3.28)

The transmuted amplitudes $A(\phi_i \phi_j, g_k)$ correctly correspond to the three-point amplitude of YMS theory, as can be seen by the vertex rule of YMS theory:

$$\begin{array}{c}
2\\
p_2\\
p_1\\
p_1\\
1
\end{array} (p_2 - p_1)^{\mu} \propto g_{\rm YM}(p_2 - p_1)^{\mu}.$$
(3.29)

We can also transmute all three gluons and place them in a specific dual colour ordering of our choice to obtain a BAS amplitude:

$$T[123] \cdot A(g_1, g_2, g_3) = A(\phi_1 \phi_2 \phi_3) = 1, \tag{3.30}$$

$$T[321] \cdot A(g_1, g_2, g_3) = A(\phi_3 \phi_2 \phi_1) = -1.$$
(3.31)

Here, the anti-symmetry factor for the reverse ordering of the biadjoint scalar colour factors is correctly represented in the anti-symmetry of the BAS amplitude.

Four Point Gluon Amplitude

The four gluon amplitude with implied colour ordering $\{1234\}$ can be denoted as

$$A(g_1, g_2, g_3, g_4) = \frac{n_{[12][34]}}{p_1 p_2} + \frac{n_{[23][41]}}{p_2 p_3} - \frac{1}{2} [(e_1 e_2)(e_3 e_4) + (e_2 e_3)(e_4 e_1)] + (e_1 e_3)(e_2 e_4), \quad (3.32)$$

where $n_{[12][34]} = n_{1234} - n_{2134} - n_{1243} + n_{2143}$ and

$$n_{1234} = \frac{1}{4}(e_1e_2)(e_3e_4)(p_1p_3) - (e_2e_3)(p_3e_4)(p_2e_1) - \frac{1}{2}(e_1e_2)(p_2e_3)(p_1e_4) - \frac{1}{2}(e_3e_4)(p_4e_1)(p_3e_2).$$
(3.33)

Upon usage of the multi-trace operator which orders all gluons into pairs and transforms them into scalars we find the following three combinations of orderings into pairs for 4 gluons:

$$A(\phi_1\phi_2,\phi_3\phi_4) = T_{12} \cdot T_{34} \cdot A(g_1,g_2,g_3,g_4) = \frac{p_1p_3}{p_1p_2} = \frac{u}{s},$$
(3.34)

$$A(\phi_1\phi_3,\phi_2\phi_4) = T_{13} \cdot T_{24} \cdot A(g_1,g_2,g_3,g_4) = 1,$$
(3.35)

$$A(\phi_1\phi_4,\phi_2\phi_3) = T_{14} \cdot T_{23} \cdot A(g_1,g_2,g_3,g_4) = \frac{p_1p_3}{p_2p_3} = \frac{u}{t}.$$
(3.36)

Which correspond to the three scattering diagrams that contribute to the 4-point colourstripped YMS amplitude. This would be accompanied by the trace structure of the initial YM partial amplitude we started with, which is $\text{Tr}[T^{a_1}T^{a_2}T^{a_3}T^{a_4}]$

$$A(\phi_1\phi_2,\phi_3\phi_4) + A(\phi_1\phi_3,\phi_2\phi_4) + A(\phi_1\phi_4,\phi_2\phi_3) = -\left(\frac{t}{s} + \frac{s}{t} + 1\right) = \frac{u}{s} - \frac{s}{t}.$$
 (3.37)

When using the insertion operator we can transmute 3 and 4 gluons into scalars that are adjacent in both single and dual colour trace. These are the partial amplitudes of the BAS. for example:

$$T[123] \cdot A(g_1, g_2, g_3, g_4) = A(\phi_1 \phi_2 \phi_3, g_4) = \frac{p_3 e_4}{p_1 p_2} - \frac{p_1 e_4}{p_1 p_4},$$
(3.38)

$$T[1234] \cdot A(g_1, g_2, g_3, g_4) = A(\phi_1 \phi_2 \phi_3 \phi_4) = \frac{1}{p_1 p_2} + \frac{1}{p_2 p_3} = \frac{1}{s} + \frac{1}{t}.$$
 (3.39)

The latter will have the dual-colour trace structure of $\operatorname{Tr}[T^{a_1}T^{a_2}T^{a_3}T^{a_4}]\operatorname{Tr}[T^{\overline{a}_1}T^{\overline{a}_2}T^{\overline{a}_3}T^{\overline{a}_4}]$.



Figure 3.1: Feynman diagram depiction of the transmutation of the partial gluon amplitude (a) into the YMS amplitude (b) by use of the operator T_{12} , the YMS amplitude c which is created by use of the combination $T_{12}T_{34}$ and a pure BAS amplitude (d) created by T[1234]. The particles ϕ_i and g_i that are not separated by a ',' are adjacent in both the original and the dual-colour ordering.

3.3 Longitudinal Operators

3.3.1 Definition and gauge invariance of L_i and L_{ij}

Now that we have a firm grasp on the trace and insertion operator we can turn our attention to more exotic transmutation operators. We consider the *longitudinal operators* L_i and L_{ij} which Cheung et. al. define as:

$$L_i \equiv \sum_j p_i p_j \partial_{p_j e_i}$$
 and $L_{ij} \equiv -p_i p_j \partial_{e_i e_j}$. (3.40)

These operators replace polarisations by polarisations by momenta by acting on the object $p_j e_i$ and $e_i e_j$. Effective this number of derivatives per transmuted particle, converting the states to a longitudinal mode. The resulting particles are, for example, pions and (special) Galileons.

We will not be combining these operators with the insertion operator, and hence not placing particles in a specific dual-colour trace. We do, however, transmute pairs of polarisations $e_i e_j$ which will introduce the same δ^{ij} that were discussed for the YMS amplitudes.

The longitudinal operators (3.40) are not guaranteed to conserve the on-shell conditions and gauge invariance of the amplitude due to the new momentum pair $p_i p_j$ introduced. We will explain how to deal with the gauge invariance of these operators in section D.4.

After this discussion, we can extend the longitudinal operator that acts on pairs to a longitudinal operator \mathcal{L} that conserves the on-shell condition and the gauge-invariance of the amplitudes on which it acts. This operator acts on all the particles present in a partial amplitude and is defined as

$$\mathcal{L} \equiv \prod_{i} L_{i} = \sum_{\rho} \prod_{i,j \in pairs} L_{ij} + \cdots .$$
(3.41)

Here the summation i runs over all gluons in the amplitude and ρ denotes the partitions

into pairs. The ellipses denote terms that vanish upon acting on any amplitude as discussed in section D.4.

3.3.2 Remedying Vanishing Amplitudes

There is one caveat to the procedure of transmuting all the gluons in the initial amplitude: it gives rise to a vanishing amplitude. We will show this explicitly for 3- and 4- point gluon amplitudes in the example section of this chapter.

Luckily all our work is not for nothing and we can still connect the derivatively coupled theories to our web of transmutation amplitudes. The way around this problem is found in recursion. The recursion relations of the 'extended NLSM' [35] which describes mixed amplitudes of biadjoint scalars ϕ interacting with NLSM pions π give rise to the property that the amplitude described by 2 biadjoint scalars interacting with (n-2) pions are exactly equal to a pure *n*-point pion amplitude:

$$A_n(\phi_i, \phi_j, \pi_1, \dots, \pi_{n-2}) = A_n(\pi_i, \pi_j, \pi_1, \dots, \pi_{n-2}).$$
(3.42)

This property of the mixed amplitudes was also separately shown in [69], where it was deduced that this property had to be satisfied for any mixed theory of Nambu-Goldstone bosons and biadjoint scalars. Furthermore they deduce that no vertices containing only one biadjoint scalar $\phi^{a\bar{a}}$ can exist.

While we have used the example of transmuting gluons into pions for the discussion of longitudinal operators, the same operators can be applied to an amplitude of Born-Infeld photons. When transmuted, this amplitude becomes an amplitude describing derivatively coupled Special-Galileons. The equality of a pure SG amplitude and an amplitude of SG coupled to two NLSM pions is also proven in section 3.2 of [35].

More recently, several other relations for mixed amplitudes of coloured scalars with and NLSM amplitudes were found by Arkani-Hamed and collaborators in [70, 71, 72]. For starters, in their 2024 paper entitled 'NLSM \subset Tr (ϕ^3)' the authors employ kinematic shifts of planar propagators to show that mixed amplitudes with an odd number of pions vanishes. Moreover, mixed amplitudes with a single coloured scalar leg ϕ also vanish due to the Adler zero property of taking any of the pion momentum to zero. They also derive the property of amplitudes with two ϕ 's combined with an even number (n = 2m) of π 's gives rise to a (2m + 2)-point pure NLSM amplitude.

The identities of mixed amplitudes can be used to construct pure pion amplitudes from gluons amplitudes in the following way. First, we transmute two gluons into biadjoint scalars using the trace operator T_{ij} . Next, we apply \mathcal{L} on the obtained YMS amplitude to transmute the remaining gluons into NLSM pions. The final object will be a mixed amplitude with two scalars and (n-2) gluons. This procedure can be denoted as

$$\mathcal{L} \cdot T_{ij} \cdot A(g_1, \dots, g_n) = A(\pi_1, \dots, \pi_n).$$
(3.43)

As the specific pair of gluons i, j does not influence the outcome, we will use the shorthand notation $T\mathcal{L}$ for this operation. The resulting amplitude is a permutation invariant pion amplitude, therefore it is not important which pair of gluon we choose to transmute into biadjoint scalars. We can also choose to create more than two biadjoint scalar to more general mixed amplitudes of the extended NLSM. The same discussion can be applied to Born-Infeld amplitudes transmuted into (mixed) Special-Galileon amplitudes.

3.3.3 Examples: Transmuting Gluons into Pions

3-Point amplitudes

For starters, we recall that any odd-point NLSM amplitude vanishes. Especially at 3point , where there are no non-zero Mandelstam variables due to the on-shell conditions combined with momentum conservation. For these reasons, we expect the transmuted gluon amplitude to also vanish.

Starting again from the 3 point, colour-ordered gluon partial amplitude

$$A(g_1, g_2, g_3) = \frac{1}{2}e_1e_2(p_2e_3 - p_1e_3) + (\text{cyclic}).$$
(3.44)

First we check that transmutation of all gluons into pions directly gives rise to a vanishing amplitude. This can be shown explicitly through summation of the different orderings of the transmutation:

$$L_3 \cdot L_{12} \cdot A(g_1, g_2, g_3) = -\frac{1}{2} (p_3 p_1 \partial_{p_1 e_3} + p_3 p_2 \partial_{p_2 e_3}) (p_1 p_2 (p_2 e_3 - p_1 e_3))$$

= $\frac{1}{2} (p_1 p_2) (p_3 p_1) - \frac{1}{2} (p_1 p_2) (p_3 p_2),$ (3.45)

$$(1 \to 2 \to 3 \to 1) = \frac{1}{2}(p_2 p_3)(p_1 p_2) - \frac{1}{2}(p_2 p_3)(p_1 p_3),$$
 (3.46)

$$(1 \to 3 \to 2 \to 1) = \frac{1}{2}(p_3p_1)(p_2p_3) - \frac{1}{2}(p_3p_1)(p_2p_1),$$
 (3.47)

$$(L_3 \cdot L_{12} + L_2 \cdot L_{13} + L_1 \cdot L_{23}) \cdot A(g_1, g_2, g_3) = 0.$$
(3.48)

We note that this summation over the different pair arrangements of transmutation directly into pions is corresponds to a gauge transformation where all the polarisation vectors are set to their corresponding momentum vector, i.e. $e_i \rightarrow p_i$:

$$A(g_1, g_2, g_3) \xrightarrow{e_i \to p_i} \frac{1}{2} p_1 p_2 (p_2 p_3 - p_1 e_3) + \text{cyclic}$$

$$= \frac{1}{2} [(p_1 p_2) (p_2 p_3) - (p_1 p_2) (p_1 e_3) + (p_2 p_3) (p_3 p_1) - (p_2 p_3) (p_2 e_1) + (p_3 p_1) (p_1 p_2) - (p_3 p_1) (p_3 e_2)]$$

$$= 0. \qquad (3.49)$$

As stated before in section subsection 3.3.2, we get around this gauge transformation by first transmuting a pair of gluons into biadjoint scalars before applying the longitudinal operator L_i to transmute the third gluon into a pion

$$\mathcal{L} \cdot T[12] \cdot A(g_1, g_2, g_3) = \frac{1}{2} L_3 \cdot (p_2 e_3 - p_1 e_3)$$

$$= \frac{1}{2} \sum_j p_3 p_j \partial_{p_j e_3} (p_2 e_3 - p_1 e_3)$$

$$= \frac{1}{2} (p_3 p_2 - p_3 p_1) = p_3 p_2, \qquad (3.50)$$

where we have used that $p_1 = -p_2 - p_3$, $p_i^2 = 0$ and $p_3p_2 = \frac{1}{2}(p_2 + p_3)^2$. Depending on which initial pair of gluons is chosen to be transmuted into gluons we will find the identical result for the three-point pion amplitude:

$$\mathcal{L} \cdot T[12] \cdot A(g_1, g_2, g_3) = A(\phi_1 \phi_2, \pi_3) \mathcal{L} \cdot T[23] \cdot A(g_1, g_2, g_3) = A(\phi_2 \phi_3, \pi_1) \mathcal{L} \cdot T[31] \cdot A(g_1, g_2, g_3) = A(\phi_3 \phi_1, \pi_2)$$

$$= A(\pi_1, \pi_2, \pi_3) = 0.$$

$$(3.51)$$

This is in agreement with fact that odd-point pion amplitudes should vanish and in agreement with the findings of [72] that mixed amplitudes with an odd number of π 's should vanish.

4-Point amplitudes

We start by naively applying a transmutation of all four gluon legs into pions by using using the product of pairs of L_{ij} . For convenience, we can utilise our earlier result of the multi-trace amplitude, such as

$$T[12] \cdot T[34] \cdot A(g_1, g_2, g_3, g_4) = \frac{p_1 p_3}{p_1 p_2}, \qquad (3.52)$$

seeing as the longitudinal operator $L_{ij} = -p_i p_j T_{ij}$ is a linear combination of these trace operators.

In this naive calculation for the complete pion amplitude, we sum over all distinct orderings of 4 particles into pairs: $\mathcal{L} = L_{12}L_{34} + L_{13}L_{24} + L_{14}L_{23}$. Explicitly, these different arrangements are given by

$$L_{12}L_{34} \cdot A(g_1, g_2, g_3, g_4) = (-p_1 p_2)(-p_3 p_4)T[12] \cdot T[34] \cdot A(g_1, g_2, g_3, g_4)$$

= $(-p_1 p_2)(-p_3 p_4)\frac{p_1 p_3}{p_1 p_2} = (p_3 p_4)(p_1 p_3)$
= $-\frac{1}{4}u(u+t) = \frac{1}{4}us,$ (3.53)

$$L_{13}L_{24} \cdot A(g_1, g_2, g_3, g_4) = \frac{1}{4}uu, \qquad (3.54)$$

$$L_{14}L_{23} \cdot A(g_1, g_2, g_3, g_4) = \frac{1}{4}ut, \qquad (3.55)$$

where we have used the notation $s = s_{12} = (p_1 + p_2)^2 = 2p_1p_2$, $t = s_{23} = (p_2 + p_3)^2 = 2p_2p_3$ $u = s_{13} = (p_1 + p_3)^2 = 2p_1p_3$ for the Mandelstam invariants.

The summation over these three arrangements into pairs then results in

$$\mathcal{L} \cdot A(g_1, g_2, g_3, g_4) = \frac{1}{4}u(s+t+u) = 0,$$
 (3.56)

where we have used momentum conversation to show that $p_3p_4 = p_3(-p_1 - p_2 - p_3) = -p_1p_3 - p_2p_3$. In the last equality we use the fact that the Mandelstams sum to zero: s + t + u = 0. Therefore, this naive application of the transmutation operator on all gluon legs results in a vanishing amplitude, which we already conjectured due to the gauge invariance of the Yang-Mills amplitude.

To obtain the correct 4-point gluon amplitude we thus resort to utilising the property of the mixed BAS and NLSM amplitudes. We apply the operator $T\mathcal{L}$ to first transmute a pair of gluons into biadjoint scalars before transmuting the remaining gluons to pions. The final result will be a permutation invariant pure pion amplitude:

$$T_{12}L_{34} \cdot A_{YM} = (-p_3p_4)T[12] \cdot T[34] \cdot A(g_1, g_2, g_3, g_4) = p_1p_3 = \frac{u}{2}$$
$$= \frac{(-p_3p_4)u}{s} = \frac{(p_3p_1 + p_3p_2)u}{s} = \frac{(u+t)u}{2s} = \frac{-su}{2s} = -\frac{u}{2}.$$
(3.57)

For completeness we also explicitly calculate the other permutations²

$$T[13] \cdot \mathcal{L} \cdot A(g_1, g_2, g_3, g_4) = (-p_2 p_4) \cdot 1 = p_2 p_1 + p_2 p_3 = \frac{s+t}{2} = -\frac{u}{2},$$
(3.58)

$$T[14] \cdot \mathcal{L} \cdot A(g_1, g_2, g_3, g_4) = (-p_2 p_3) \frac{p_1 p_3}{p_2 p_3} = -\frac{u}{2}.$$
(3.59)

We can therefore conclude that this procedure generates the correct pion amplitude

$$\mathcal{L} \cdot T[12] \cdot A(g_1, g_2, g_3, g_4) = A(\phi_1 \phi_2, \pi_3, \pi_4) = p_1 p_3 \mathcal{L} \cdot T[13] \cdot A(g_1, g_2, g_3, g_4) = A(\phi_1 \phi_3, \pi_2, \pi_4) = p_1 p_3 \mathcal{L} \cdot T[14] \cdot A(g_1, g_2, g_3, g_4) = A(\phi_1 \phi_3, \pi_2, \pi_4) = p_1 p_3$$

$$= A(\pi_1, \pi_2, \pi_3, \pi_4).$$

$$(3.60)$$

The result is a permutation invariant pion amplitude, which is identical regardless of the choice of initial pair of biadjoint scalars that is created.

3.4 Web of theories connected through transmutation

In the above chapter, we have identified several building blocks with which we can transmute amplitudes:

 $^{^{2}}$ Recall the ordering of the transmutation operators, we should read this as the operators acting from left to right. First the trace operator acts on the amplitude followed by the longitudinal operator

- The single trace transmutation $T[i_1 \cdots i_n]$ which introduces a new colour ordering to the resulting partial amplitude.
- The multi trace transmutation $T[\alpha] \cdots T[\beta]$ which results in multiple colour traces in the obtained amplitude. Specifically the transmutation of all particles ordered into pairs with the operator combination $T[i_1j_1] \cdots T[i_nj_n]$ is interesting as it results in a special colour structure of products of δ_{ij} 's.
- The combination of a trace and longitudinal operators $T\mathcal{L}$, which adds derivatively coupled interactions to the amplitude.

There are therefore three *pure* objects (of interactions of only 1 type of particle) we can construct out of given a starting spin-1 or spin-2 amplitude. If we choose to not transmute all of the external legs, we land on a mixed theory of transmuted legs interacting with untransmuted legs. Starting from a extended gravity amplitudes we can find the following pure amplitudes:

$$A_{\rm YM} = T[\alpha] \cdot A_{\rm EG},\tag{3.61}$$

$$A_{\rm BI} = T\mathcal{L} \cdot A_{\rm EG},\tag{3.62}$$

$$A_{\rm EM} = T[i_1 j_1] \cdots T[i_n j_n] \cdot A_{\rm EG} \tag{3.63}$$

The hybrid theory amplitudes of external gravitons coupled to external gluons/photons/BI photons can be computed by leaving some external graviton legs untouched by transmutation. Applying transmutation to Yang-Mills amplitudes gives

$$A_{\rm BAS} = T[\alpha] \cdot A_{\rm YM},\tag{3.64}$$

$$A_{\rm NLSM} = T\mathcal{L} \cdot A_{\rm YM}, \tag{3.65}$$

$$A_{\rm YMS} = T[i_1 j_1] \cdots T[i_n j_n] \cdot A_{\rm YM}. \tag{3.66}$$

Finnaly we can find the following amplitudes from Born-Infeld amplitudes

$$A_{\rm NLSM} = T[\alpha] \cdot A_{\rm BI}, \tag{3.67}$$

$$A_{\rm SG} = T\mathcal{L} \cdot A_{\rm BI},\tag{3.68}$$

$$A_{\rm DBI} = T[i_1 j_1] \cdots T[i_n j_n] \cdot A_{\rm BI}.$$

$$(3.69)$$

The theories that are connected through transmutation are depicted in the tetrahedron of Figure 3.2. In this visualisation it is clear that transmutation is a type of 'top-down' process which creates lower spin partial amplitudes from the spin-2 extended gravity amplitudes.

The tetrahedron also reinstates that the order of applying transmutation operators does not matter, as these lead to the partial amplitudes of the same theory. This was already apparent from the fact that the operators commute. As an example, we can calculate the NLSM amplitudes from the extended gravity amplitudes through the combination $T[\alpha] \cdot T\mathcal{L}$ and $T\mathcal{L} \cdot T[\alpha]$ as indicated by the red and orange arrows in Figure 3.2.


Figure 3.2: Visualisation of Unifying Relations between amplitudes of different theories, based on the tetrahedron of [60] that displays the theories that are connected through the BCJ double copy of different combination of BCJ numerators.

3.5 Infrared properties of Transmutation

As discussed in subsection 3.5.1, the amplitudes of gluon and graviton scattering take special form when taking one of the external particles to be soft; $p_i \to 0$. The amplitude factorises as a soft factor multiplied by a lower point amplitude

$$A^{(n)} \xrightarrow{p_i \to 0} S^{(i)} \cdot A^{\prime(n-1)}. \tag{3.70}$$

The soft factor is a pertubative expansion and can be computed to different orders. At leading order in this expansion the gluon and graviton soft factors are simple factors [42]. However, at higher orders these factors will become more elaborate; depending on momenta and polarisations of adjacent legs and containing higher order differential operators with respect to these variables [45]. We are therefore interested whether transmutation preserves the soft factorisation of the amplitude at all, or perhaps only at leading order.

To investigate whether this behaviour persists, we apply transmutation on the soft factor relation by first factorising a general product of transmutation operators as $\mathcal{T}^{(n)} = \mathcal{T}^{(i)} \cdot \mathcal{T}^{(n-1)}$, where the operator $\mathcal{T}^{(i)}$ is the operator responsible for transmuting leg *i*, which will be taken to be soft and $\mathcal{T}^{(n-1)}$ transmutes the remaining (n-1) external legs. We can show that then show that upon transmutation we can find an expression for the transmuted soft factor $\tilde{S}^{(i)}$ which acts on the transmuted lower point amplitude denoted by $\tilde{A}^{(n-1)}$ by calculating

$$\mathcal{T}^{(n)} \cdot A^{(n)} \xrightarrow{p_i \to 0} \mathcal{T}^{(i)} \mathcal{T}^{(n-1)} S^{(i)} A^{(n-1)} \\
= \left[\mathcal{T}^{(i)} \mathcal{T}^{(n-1)}, S^{(i)} \right] A^{(n-1)} \\
= \mathcal{T}^{(i)} \left[\mathcal{T}^{(n-1)}, S^{(i)} \right] A^{(n-1)} + \left[\mathcal{T}^{(i)}, S^{(i)} \right] \mathcal{T}^{(n-1)} A^{(n-1)} \\
= \left[\mathcal{T}^{(i)}, S^{(i)} \right] \left(\mathcal{T}^{(n-1)} A^{(n-1)} \right) \\
= \left[\mathcal{T}^{(i)}, S^{(i)} \right] \tilde{A}^{(n-1)}.$$
(3.71)

This indicates that the transmuted soft factor can be denoted as

$$\tilde{S}^{(i)} \equiv \left[\mathcal{T}^{(i)}, S^{(i)}\right]. \tag{3.72}$$

The necessary arguments in this derivation are as follows. In the second line we can denote the commutator identity due to the fact that the commutator is defined as

$$\left[\mathcal{T}^{(i)}\mathcal{T}^{(n-1)}, S^{(i)}\right]A^{(n-1)} = \mathcal{T}^{(i)}\mathcal{T}^{(n-1)}S^{(i)}A^{(n-1)} - S^{(i)}\mathcal{T}^{(i)}\left(\mathcal{T}^{(n-1)}A^{(n-1)}\right), \qquad (3.73)$$

where the second term vanishes because $\mathcal{T}^{(i)}$ depends on derivatives with respect to e_i and p_i , whereas $\mathcal{T}^{(n-1)}A^{(n-1)}$ by construction is independent of the variables of leg *i*.

Next we argue that the commutator $[\mathcal{T}^{(n-1)}, S^{(i)}]$ vanishes for the trace operator T_{jk} and the insertion operator I_{jkl} due to the dependence of $\mathcal{T}^{(n-1)}$ on combination $e_j e_k$, $e_j p_k$ or $p_j p_k$ where $j, k \neq i$ while $S^{(i)}$ only depends on combinations $e_i e_j$, $e_i p_j$ or $p_i p_j$ at leading and subleading order as we will see shortly.

This condition also holds at leading order for the Longitudinal operator L_{jk} , however we will need to revise this statement for the subleading order soft factor due to the dependence of the soft factor on adjacent legs.

We will calculate explicitly the transmutation of the leading order soft factor for the theories that originate from gravity and Yang-Mills amplitudes, as these lead to the leading order soft factors YM, YMS and the NLSM. The transmutation of the subleading soft factors is left to section D.5 of the appendix.

These soft factors are interesting to us because we aim to create amplitudes with interactions of scalars, pions and gluons similar that may align with the amplitudes that have $\sigma = 0$ created by the cubic numerator $N_6^{(3)}$ of section 2.4.

3.5.1 Transmutation of the leading order soft factors

Extended Gravity

Recall that the leading-order soft graviton factor given by Weinberg is given by [45]

$$S_{G}^{(i)} = \sum_{j \neq i} \frac{p_{i} e_{j} p_{i} \tilde{e}_{j}}{p_{i} p_{j}}.$$
(3.74)

As stated in the definition of the web of related theories, we are able to transmute the extended gravity amplitudes into EM, YM and BI amplitudes through our three different transmutation operators T_{ij} , I_{ijk} and L_{ij} . By the fact that the graviton soft factor is independent of products of polarisation $e_i e_j$ it is straightforward that the soft factor of Einstein-Maxwell vanishes:

$$S_{\rm EM}^{(ij)} = \left[T_{ij}, S_G^{(i)}\right] = 0.$$
 (3.75)

The same argument holds for the computation of the leading order soft factor of BI amplitudes

$$S_{\rm BI}^{(i)} = \left[L_i, S_G^{(i)}\right] = 0.$$
(3.76)

This vanishing leading order soft factor is in agreement with the fact that Born-Infeld photons are derivatively coupled which should lead to vanishing amplitudes when this leg is taken to be soft. Finally, the soft factor of Yang-Mills amplitudes can be found through the commutator with the insertion operator,

$$S_{\rm YM}^{(ijk)} = \left[I_{ijk}, S_G^{(j)}\right] = \left[\partial_{p_i e_j} - \partial_{p_k e_j}, \sum_{j \neq i} \frac{p_i e_j p_i \tilde{e}_j}{p_i p_j}\right],\tag{3.77}$$

where in this case the leg j is taken to be soft with legs i and k adjacent to leg j. This then gives rise to the familiar leading order soft factor

$$S_{\rm YM}^{(ijk)} = \frac{p_i e_j}{p_i p_j} - \frac{p_k e_j}{p_k p_j},$$
(3.78)

which is consistent with the original soft factor by Weinberg [41].

Yang-Mills

Continuing with the leading order soft factor of Yang-Mills we can calculate commutators of (3.78) with the transmutation operators to find the soft factors of Yang-Mills-Scalar, biadjoint scalar and NLSM amplitudes.

When denoting leg j as the soft leg, the commutator with the trace operator results in the soft factor of YMS theory:

$$S_{\rm YMS}^{(ijk|jl)} = \left[T_{jl}, S_{\rm YM}^{(ijk)} \right] = 0.$$
(3.79)

the vertical bar indicates that particle j is in the same dual colour trace as particle l. Next the commutation with the insertion operator gives the soft factor of BAS theory

$$S_{BAS}^{(ijk|ljm)} = \left[T_{ljm}, S_{YM}^{(ijk)}\right] = \frac{\delta_{li} - \delta_{im}}{p_i p_j} - \frac{\delta_{lk} - \delta_{km}}{p_k p_j}.$$
(3.80)

Finally, at leading order the commutation with the longitudinal operator results in the vanishing of the NLSM soft factor

$$S_{\rm NLSM}^{(ijk)} = 0, \qquad (3.81)$$

as is to be expected from the Adler zero property exhibited by pion amplitudes.

3.6 On-shell recursion, factorisation and the Double Copy

It is further investigated by Cheung and collaborators whether the objects obtained after transmutation are in fact always scattering amplitudes which obey the appropriate properties of physical scattering amplitudes: on-shell construction, proper factorisation the Double Copy prescription. This section briefly touches upon the argumentation of these properties.

3.6.1 On-shell Constructibility

For starters, the discussion of the on-shell constructibility of the obtained amplitudes investigates whether higher-point amplitudes can be constructed recursively from lower-point amplitudes, as discussed in section 1.3.

By use of the BCFW recursion relation [38], it was shown through an -all-line shift that for amplitudes with more than four particles the deformed amplitude goes to zero in the case of large shifts z as a consequence of dimensional analysis.

It is further argued that for the insertion operator I_{ijk} , the resulting amplitude will have fewer derivatives as momenta are stripped off, and by using a power counting method this transmuted amplitude will be on-shell constructable.

This argumentation does not apply to the longitudinal operator L_{ij} , as it increases the number of derivatives of the amplitude. The on-shell constructibility of these theories is then saved if the resulting amplitude has suitable IR properties such as the Adler zero, which was shown in subsection 3.5.1.

3.6.2 Factorisation

In general, on the residue where the factorisation channel is propagator $1/p_I$ is tuned to 0, the the original amplitude will factorise as

$$A(\cdots) \sim \sum_{I} A_{L}(\cdots I_{L}) A(I_{R} \cdots).$$
(3.82)

We can investigate what happens to the factorisation of gluon amplitudes after transmutation. The summation over I runs over all possible internal gluon and scalar states. The discussion of proper factorisation is different for each transmutation operator and is discussed in [22]. Here we will simply state in what manner the amplitudes factorise depending on the initial amplitude.

The factorisation that will later be relevant to our discussion is that of the longitudinal operator and trace operators in pairs. The latter of these transmutes gluon amplitudes into YMS amplitudes, leaving the intermediate state unaltered (gluon exchange) and therefore needs no separate discussion.

The factorisation of the longitudinal operator $\mathcal{L} = \mathcal{L}_L \cdot \mathcal{L}_R$ knows two scenario's. In the first, the amplitude that is transmuted has scalars on both sides of the factorisation channel, requiring the internal particle to also be a scalar. The amplitude then factorises as

$$\mathcal{L} \cdot A(\cdots \phi_i \phi_l \cdots, g_j, g_k, \cdots) \sim \mathcal{L}_L \cdot A_L(\cdots \phi_i \phi_L, g_j, \cdots) \mathcal{L}_R \cdot A_R(\phi_R \phi_l \cdots, g_k, \cdots) \quad (3.83)$$
$$\sim A_L(\cdots \phi_i \phi_L, \pi_j, \cdots) A_R(\phi_R \phi_l \cdots, \pi_k, \cdots). \quad (3.84)$$

This is consistent with the factorisation of how the amplitude $A(\cdots \phi_i \phi_l \cdots, \pi_j, \pi_k, \cdots)$ would factorise. The residue of the factorisation can be schematically depicted at 6-point as



If the gluons are on the opposite side of the factorization channel from the all scalars, then the only possible internal state of the original amplitude is a gluon, and the transmuted amplitude factorises with an internal pion

$$\mathcal{L}_{L} \cdot A(\cdots \phi_{i}\phi_{l}\cdots,g_{j},g_{k},\cdots) \sim \mathcal{L}_{L} \cdot \mathcal{L}_{R} \cdot A_{L}(\cdots \phi_{i}\phi_{l}\cdots,g_{j},g_{L},\cdots)A_{R}(g_{R},g_{k},\cdots)$$
(3.86)
$$\sim A_{L}(\cdots \phi_{i}\phi_{l}\cdots,\pi_{j},\pi_{L},\cdots)A_{R}(\pi_{R},\pi_{k}\cdots).$$
(3.87)

We can display this schematically as



We will later use these factorisation properties to analyse the amplitudes that we construct from *generalised transmutation*, where we define a new type of transmutation operator with the aim of landing on the amplitudes of BCJ bootstrap of section 2.4.

3.6.3 Transmutation and The Double Copy

As we have seen in subsection 1.5.1, we can use the double copy to relate the amplitudes of different theories. Specifically, the BCJ double copy relates all the amplitudes in the tetrahedron of Figure 3.2, which therefore solidifies the unifying relations of between different

these theories. The transmutation of amplitudes can be shown to preserve the duality of the amplitudes through the use of the KLT relations.

The application of transmutation operators on the double copy amplitudes can be done in the KLT representation (Appendix B), due to the fact that the transmutation operators commute with the KLT relations.

In general, the KLT product allows for the crossing of amplitudes of different dualitysatisfying theories. In general we can then find the following amplitudes from the KLT product of YM amplitudes with other theories:

$$GR = YM \overset{KLT}{\otimes} YM, \qquad (3.89)$$

$$EM = YMS \overset{KLT}{\otimes} YM, \qquad (3.90)$$

$$EYM = gBAS \bigotimes_{KLT}^{KLT} YM, \qquad (3.91)$$

$$BI = NLSM \overset{RL1}{\otimes} YM. \tag{3.92}$$

Furthermore, the double copy of NLSM amplitudes results in the following amplitudes:

TZT (D

$$BI = YM \overset{KLT}{\otimes} NLSM, \qquad (3.93)$$

$$DBI = YMS \overset{KL1}{\otimes} NLSM, \qquad (3.94)$$

$$BI + NLSM = gBAS \bigotimes_{KLT}^{KLT} NLSM, \qquad (3.95)$$

$$SG = NLSM \bigotimes^{RL1} NLSM.$$
 (3.96)

Due to the fact that transmutation operators commute with the KLT relations, transmutation can be applied to either of the amplitudes in the product. For example, the amplitudes of BI theory can be constructed through by applying the longitudinal operator to extended graviton amplitudes. Applying this operator to either of the amplitudes in the double copy KLT product we find

$$A_{\rm BI} = T\mathcal{L} \cdot A_{\rm EG} = A_{\rm YM} \bigotimes_{\rm KIT}^{\rm KLT} T\mathcal{L} \cdot A_{\rm YM}$$
(3.97)

$$= A_{\rm YM} \overset{\rm KLI}{\otimes} \tilde{A}_{\rm NLSM}, \qquad (3.98)$$

which agrees with (3.93). The same argument can be applied for the different types of transmutation operators and different combinations of amplitudes in the KLT product.

3.7 UV completion

The process of transmutation extends to the UV completion of theories within the interconnected web of theories related by transmutation. However, it's important to note that the resulting amplitudes might not always be consistent since the consistency check of section 3.6 assumed a massless spectrum for the intermediate particles.

As discussed in Equation 1.5.3, the UV completion of Yang-Mills is obtained through dimensionful α' corrections in the Lagrangian. Because the YM $+\alpha'$ amplitude is gauge invariant, after transmutation, the object will remain gauge invariant. The single trace operator produces BAS amplitudes from YM, therefore we can speculate that the amplitudes created by

$$T[i_1\cdots i_n]\cdot A_{\mathrm{YM}+\alpha'}(g_{i_1},\cdots,g_{i_n}) = A_{\mathrm{BAS}+\alpha'}(\phi_{i_1}\cdots\phi_{i_n}), \qquad (3.99)$$

where $A_{BS+\alpha'}(\phi_{i_1}\cdots\phi_{i_n})$ are amplitude obtained through some theoretical UV completion of BS theory with α' corrections.

It would be interesting if we could relate these amplitudes to some known theory or description. An example candidate for the UV complete YM theory is said to be open superstring theory [73]. The details of this theory go beyond the scope of this work, but we can nevertheless relate the amplitudes as

$$A_{\rm YM+\alpha'} \sim A_{\rm open}.$$
 (3.100)

Applying transmutation on this open superstring amplitude is then argued to result in

$$T[i_1\cdots i_n]\cdot A_{\text{open}}(g_{i_1},\cdots,g_{i_n}) = A_Z(\phi_{i_1}\cdots\phi_{i_n}), \qquad (3.101)$$

where A_Z denotes an amplitude of biadjoint scalars described by the framework of Ztheory. A short description of these amplitudes is found in Appendix C, which is based on [58, 59, 74].

A more explicit application of transmutation of amplitudes of UV-completed theories can be done in the KLT representation. In this framework, the amplitudes of open superstrings can be described as the product of amplitudes in Z-theory and YM theory through the KLT relations:

$$A_{\rm open} = A_Z \overset{\rm KLT}{\otimes} A_{\rm YM}. \tag{3.102}$$

Here \otimes denotes the KLT product of the amplitudes. The single trace operator can then be applied to the product (with which it commutes), giving

$$T[i_1 \cdots i_n] \cdot A_{\text{open}} = T[i_1 \cdots i_n] \cdot (A_Z \otimes A_{\text{YM}})$$
(3.103)

$$= A_Z \otimes T[i_1 \cdots i_n] \cdot A_{\rm YM} \tag{3.104}$$

$$= A_Z \otimes A_{BAS}. \tag{3.105}$$

Since BAS amplitudes function as an 'identity' element in the KLT framework, the single trace operator therefore produces

$$T[i_1 \cdots i_n] \cdot A_{\text{open}} = A_Z. \tag{3.106}$$

Similarly, applying the longitudinal transmutation procedure result in the KLT product of Z-theory amplitudes and NLSM amplitudes:

$$L \cdot T[i_1 i_n] \cdot A_{\text{open}} = L \cdot T[i_1 i_n] \cdot (A_Z \otimes A_{\text{YM}})$$
(3.107)

$$= A_Z \otimes (L \cdot T[i_1 i_n] \cdot A_{\rm YM}) \tag{3.108}$$

$$= A_Z \otimes A_{\text{NLSM}}.$$
 (3.109)

The amplitudes $A_Z \otimes A_{\text{NLSM}}$ describe a scalar amplitude which corresponds to the amplitudes of the NLSM at low energies. It does not correspond to Abelian Z-theory [59].

2

Finally we note that the original paper on transmutation relations [22] does not comment on the UV completion of YMS amplitudes obtained through the pair-wise application of trace operators such as $T[i_1i_2] \cdots T[i_{n-1}i_n]$. Presumably the reason for this is that the UV completion of YM+ ϕ amplitudes is not as well-defined as the above theories. More information on higher derivative corrections to these theories can be found in e.g. [75] and [76]. The transmutation of open-string amplitudes could therefore be an interesting approach to investigating the UV completion of such theories.

Chapter 4

Generalised Transmutation of Yang-Mills

In chapter 2, we demonstrated the generation of hybrid amplitudes through the use of BCJ numerators, which specifically led to interactions involving adjoint scalars mediated by gauge interactions. On the other hand, chapter 3 introduced hybrid scalar amplitudes derived from the application of transmutation operators. This chapter aims to explore the intersection of these methodologies to address several key questions: Do these approaches yield the same amplitudes? Are the resulting amplitudes governed by interactions originating from analogous Lagrangians? Regardless of whether the methods converge or diverge, understanding the underlying reasons for their (dis)connection is of significant interest.

4.1 Defining the Generalised Transmutation Operator

The proposed amplitudes of the NLSM gauged described in section 2.5 are found by defining the kinematic numerator N through the powers of Mandelstam invariants s^p . The numerators at different orders give rise to scalars interacting with gluons and derivatively coupled scalars, therefore giving rise to YMS and NLSM amplitudes. In 3.2.4 and 3.2.5 it was discussed that the multi-trace amplitudes that transmute gluons into pairs of scalars through the differential operator T_{ij} also give rise to YMS amplitudes. Furthermore, the longitudinal operator combined with a trace operator $T \cdot \mathcal{L}$ was shown to transmute the colour-ordered gluon amplitudes into NLSM amplitudes in 3.3.

Therefore, a combination of these two transmutations would be expected to land on mixed theories describing scalars coupled to gluons and possibly gluons depending on whether or not all the external legs are transmuted and whether gluons will show up as intermediate particles. More importantly, we will construct the new operator to keep track of the order of Mandelstam invariants, allowing us to construct an amplitude with $\sigma = 0$. We would like to know whether the amplitude constructed through this method coincides with the amplitude generated by $N_6^{(3)}$.

To achieve this, the polarisations of the gluons need to be stripped off through the trace operator in pairs. When this is applied to all legs, we get a pure YMS amplitude. In order to transmute create derivatively coupled scalar modes, some pairs of polarisations are replaced by pairs of momenta $p_i p_j$. Each pair that is transformed contributes an additional factor of Mandelstams s^1 to the numerator of the amplitude.

The proposed generalised transmutation (GT) operator, designed to lead us to the aforementioned hybrid amplitudes, is as follows. Consider the operator $\mathbb{T}^{(1...n)}$, which transmutes the external leg 1 through n. When this operator acts on an n-point Yang-Mills amplitude A_n^{YM} , it transforms all external legs and outputs a non-trivial scalar amplitude:

$$\mathbb{T}^{(1\dots n)} \cdot A_n^{\mathrm{YM}} = A_n^{\mathrm{scalar}}.$$
(4.1)

This operator will be constructed out of a new type of longitudinal operator that strips off pairs of polarisations $(e_i e_j)$ and replaces them by $(1 + s_{ij})$ is denoted by

$$\mathbb{L}_{ij} \equiv (1 + \tau p_i p_j) \partial_{e_i e_j} = T_{ij} + \tau L_{ij}. \tag{4.2}$$

The dimensionless parameter τ will keep track of number of pairs of gluons that have been transformed into pions. Where we use the original definitions $T_{ij} = \partial_{e_i e_j}$ for the trace operator and $L_{ij} = p_i p_j \partial_{e_i e_j}$ for the longitudinal operator modulo a sign difference to split the operator into its component parts.

Similarly to the process of the original longitudinal operator, \mathbb{L}_{ij} is applied for all arrangements ρ of the particle legs into distinct pairs. The full transmutation is therefore defined as

$$\mathbb{T}^{(1\dots n)} \equiv \sum_{\rho} \prod_{i,j \in pairs} \mathbb{L}_{ij} = \sum_{\rho} \mathbb{L}_{\rho(1)\rho(2)} \dots \rho_{(n-1)\rho(n)}$$
(4.3)

When applied to a gluon amplitude, this transmutation spits out two new amplitudes

$$\mathbb{L}_{ij} \cdot A_n^{\mathrm{YM}}(g_1, \dots, g_i, g_j, \dots, g_n) = A(\phi_i \phi_j, g_1, \dots, g_n) + \tau A(g_1, \dots, \pi_i, \dots, \pi_j, \dots, g_n).$$
(4.4)

Because it contains the product of $\frac{n}{2} \mathbb{L}_{ij}$ operators, the total transmutation $\mathbb{T}^{(1...n)}$ which transmutes all external legs contains contributions starting at τ^0 which will be the pure YMS amplitudes of subsection 3.2.4 up to order $\tau^{n/2}$. At this order the longitudinal operator L_{ij} has been applied to all legs. However, in subsection 3.3.2 we saw that this $\mathcal{O}(\tau^{n/2})$ transmutation will result in a vanishing amplitude and that the actual NLSM pion amplitude is constructed by first setting one pair $(e_i e_j)$ to 1. Fortunately, this is exactly what happens at $\mathcal{O}(\tau^{n/2-1})$.

The amplitudes in between these orders will therefore be non-trivial scalar amplitudes of pions interacting with scalars. Due to the vanishing amplitudes, the first encounter with these mixed amplitudes will be at 6-point.

The spectrum of contributions can then be depicted as

$$\mathbb{L}_{12}\mathbb{L}_{34}\cdots\mathbb{L}_{(n-1)(n)}\cdot A_n^{\mathrm{YM}} = \tau^0 A\left(\phi^n\right) \tag{4.5}$$

$$+\tau^{1}A\left(\phi^{n-2},\pi^{2}\right) \tag{4.6}$$

$$+ \tau^2 A\left(\phi^{n-4}, \pi^4\right)$$
 (4.7)

$$+\dots$$
 (4.8)

$$+\tau^{n/2-1}A\left(\pi^{n}\right) \tag{4.9}$$

$$+\tau^{n/2}\cdot 0, \tag{4.10}$$

where we have used superscript notation to indicate the number of scalars and pions. In the following sections we will perform explicit calculations of the transmutation applied to different multiplicities.

Pair ordering and computational analysis

Although at four points there exist only three distinct arrangements of pairs of polarisation, at *n*-point the number of distinct combinations of divisions of *n* numbers into pairs is $\frac{n!}{(\frac{n}{2})!2^{(n/2)}}$. The order in τ ranges from τ^0 to $\tau^{n/2}$, where the latter again results in a vanishing amplitude. This results in 15 arrangements at 6-point, 105 arrangements at 8-point and 945 at 10-point. The pattern is that if we go from *n* point to n + 2 point, the number of arrangements at 12-point will be 945 * 11 = 10395 distinct arrangements.

In the next section, we will explicitly apply the GT operator analytically for only one specific arrangement of the 4-point amplitude. The other arrangements can be calculated analogously and will be performed using a Mathematica algorithm. First of all Mathematica will be used to calculate the distinct arrangements into pairs. After this, an algorithm can be written that applies the GT-operator. Computationally this can be done either through the application of derivatives, but it was quickly seen that taking these nested derivatives with Mathematica was computationally heavier than simply setting pairs of polarisation of $e_i e_j \rightarrow 1$ using a simple replacement rule and thereafter multiplying by the pairs of momenta $p_i p_j$ in the needed arrangement. Due to the linearity of the amplitudes in polarisation pairs, there is no ambiguity as to whether this replacement is identical to (4.1).

As input we will use colour-stripped partial amplitudes of pure Yang-Mills provided by [77]. Along with this publication, the authors Song He, Linghui Hou, Jintian Tian, and Yong Zhang provided a Mathematica package that contains an algorithm that makes use of the calculation of the so-called Cayley functions [78] in the CHY formalism of amplitude calculations. This algorithm is able to compute all tree-level amplitudes of YM, GR and BI. Through use of the double copy and dimensional reduction (in the CHY formalism [79]) the algorithm can also calculate all tree-level amplitudes of the NLSM, SG, YMS, EYM

and DBI theories. This package also allows for the generation of kinematic numerators that satisfy Jacobi identities by default.

4.2 Generalised transmutation at 4-point

We begin our investigation with the tree-level colour-ordered Yang-Mills amplitude $A_4^{\text{YM}}[1234]$ given by (3.32). There are 3 distinct arrangements of 4 particles into pairs, namely (12)(34), (13)(24) and (14)(23). The GT-operator therefore leads to the summation of three ordered contributions.

$$\mathbb{T}^{(1234)} \cdot A(g_1, g_2, g_3, g_4) = \sum_{\rho} \prod_{i,j \in pairs} L_{ij} \cdot A(g_1, g_2, g_3, g_4)$$
(4.11)

$$= (\mathbb{L}_{12}\mathbb{L}_{34} + \mathbb{L}_{13}\mathbb{L}_{24} + \mathbb{L}_{14}\mathbb{L}_{23}) \cdot A(g_1, g_2, g_3, g_4).$$
(4.12)

Let us first consider only one of the arrangements, namely the first ordering into pairs $\mathbb{L}_{12}\mathbb{L}_{34}$. This produces three amplitudes, two of which are non-vanishing:

$$\mathbb{L}_{12}\mathbb{L}_{34} \cdot A(g_1, g_2, g_3, g_4) = (T_{12} - \tau L_{12})(T_{34} - \tau L_{34})A(g_1, g_2, g_3, g_4)
= (T_{12} \cdot T_{34} + \tau T_{12} \cdot L_{34} + \tau L_{12} \cdot T_{34} + \tau^2 L_{12} \cdot L_{34}) \cdot A(g_1, g_2, g_3, g_4)
= A(\phi_1 \phi_2, \phi_3 \phi_4) + \tau A(\phi_1 \phi_2, \pi_3, \pi_4) + \tau A(\pi_1, \pi_2, \phi_3 \phi_4)
+ \tau^2 L_{12} \cdot L_{34} \cdot A(g_1, g_2, g_3, g_4).$$
(4.13)

The calculations of these 4 point transmutations were already determined in (3.52), (3.53) and (3.57):

$$\tau^0 T_{12} T_{34} \cdot A_{\rm YM} = A(\phi_1 \phi_2, \phi_3 \phi_4) = \frac{p_1 p_3}{p_1 p_2} = \frac{u}{s}, \tag{4.14}$$

$$\tau T_{12}L_{34} \cdot A_{\rm YM} = A(\phi_1\phi_2, \pi_3, \pi_4) = A_{\rm NLSM}(\pi_1, \pi_2, \pi_3, \pi_4) = p_1p_3 = u/2, \tag{4.15}$$

$$\tau^2 L_{12} L_{34} \cdot A_{\rm YM} = us/4. \tag{4.16}$$

It was explicitly shown in (3.53) that the sum of all τ^2 contributions transforms all polarisations into momenta, which is a gauge transformation and therefore this contribution vanishes. In the same section we also calculated the other arrangements of T_{ij} . The final transmutation at the different orders in τ are therefore

$$\mathcal{O}(\tau^{0}): \quad \tau^{0}[A(\phi_{1}\phi_{2},\phi_{3}\phi_{4}) + A(\phi_{1}\phi_{3},\phi_{2}\phi_{4}) + A(\phi_{1}\phi_{4},\phi_{2}\phi_{3})] = \tau^{0}\left[\frac{u}{s} - \frac{s}{t}\right], \quad (4.17)$$

$$\mathcal{O}(\tau^{1}): \qquad 3\tau^{1} \cdot A_{NLSM}(\pi_{1}, \pi_{2}, \pi_{3}, \pi_{4}) = \tau^{1}[3u], \qquad (4.18)$$

$$\mathcal{O}(\tau^2)$$
: $3\tau^2 \cdot 0 = \tau^2[0].$ (4.19)

These results were verified computationally using Mathematica, up to an overal minus sign, due to the difference in definition between the original longitudinal operator $L_{ij} = -p_i p_j T_{ij}$ and the GT-operator which has \mathbb{L}_{ij} . The YMS partial amplitude at $\mathcal{O}(\tau^0)$ is also the correct YMS amplitude, as it agrees with the YMS amplitude from the provided package up to a total sign difference.

Soft limit of generalised transmutation at 4-point

We are interested in whether the amplitudes adhere to familiar behaviour in the limit where one of the external legs is taken soft. Based on the results of (4.17), the soft behaviour of the amplitude at different orders in τ is denoted as follows:

 $\mathcal{O}(\tau^0)$: The contributions at this order for 4-point are purely Yang-Mills scalar amplitudes. All polarisations pair have been set to 1, resulting in the soft $\sigma = 0$. Similar calculations can be applied to the other arrangements in (3.52) and have been confirmed to adhere to the same soft limit using Mathematica. $\mathcal{O}(\tau^1)$: The amplitudes at this order are the pure 4-point pion amplitudes. Explicitly the soft limit at this order is $\lim_{p_i\to 0} A_{\text{NLSM}}(\pi_1, \pi_2, \pi_3, \pi_4) \sim p^1$, which corresponds to the Adler zero as expected for such amplitudes. Finally, the $\mathcal{O}(\tau^2)$ amplitudes will individually have a soft limit of one order higher, $\sim p^2$. However, when all terms are collected the total amplitude at this order vanishes.

As expected, the GT-operator at four point holds relatively few suprises. The non-trivial hybrid amplitudes we are interested in will first be encountered in the six-point calculation.

4.3 Generalised Transmutation at 6-point

Transmutation of a 6-point Yang-Mills partial amplitude will lead to more interesting results as this is where the mixed theories with ambiguous soft behaviour start to appear. This is due to the fact that there are three pairs of polarisations in each arrangements to consider when splitting into pairs. The distinct arrangements are calculated to be

We start by consider one of such arrangements into pairs, the ordering (12)(34)(56). The contribution to the transmutation of this ordering is $\mathbb{L}_{12}\mathbb{L}_{34}\mathbb{L}_{56}$. When applied to the 6-point gluon amplitude we find contributions of $\mathcal{O}(\tau^0)$ to $\mathcal{O}(\tau^3)$, where the latter gives rise to a vanishing amplitude per the gauge transformation argument when all contributions of this order in τ are summed. We have explicitly verified that this vanishes for n = 6 by computational analysis.

$$\mathbb{L}_{12}\mathbb{L}_{34}\mathbb{L}_{56} \cdot A_6^{\text{YM}} = (T_{12}T_{34}T_{56} - \tau T_{12}T_{34}L_{56} - \tau T_{12}L_{34}T_{56} - \tau L_{12}T_{34}T_{56} + \tau^2 T_{12}L_{34}L_{56} + \tau^2 L_{12}T_{34}L_{56} + \tau^2 L_{12}L_{34}T_{56} - \tau^3 L_{12}L_{34}L_{56}) \cdot A_6^{\text{YM}} \quad (4.21)$$

At the amplitude level, the GT-operator then creates the following amplitude contributions.

$$T_{12}T_{34}T_{56} \cdot A_6^{\rm YM} = A_{\rm YMS}(\phi_1\phi_2, \phi_3\phi_4, \phi_5\phi_6) \tag{4.22}$$

$$\tau T_{ij}T_{kl}L_{mn} \cdot A_6^{\rm YM} = \tau (A(\pi_1\pi_2, \phi_3\phi_4, \phi_5, \phi_6) + A(\phi_1\phi_2, \pi_3\pi_4, \phi_5, \phi_6) + A(\phi_1\phi_2, \phi_3\phi_4, \pi_5, \pi_6))$$
(4.23)

$$\tau^{2}T_{12}L_{34}L_{56} \cdot A_{6}^{\text{YM}} = \tau^{2}(A(\phi_{1}\phi_{2},\pi_{3},\pi_{4},\pi_{5},\pi_{6}) = A_{NLSM}(\pi_{1},\pi_{2},\pi_{3},\pi_{4},\pi_{5},\pi_{6}))$$
(4.24)
$$\tau^{3}L_{12}L_{34}L_{56} \cdot A_{6}^{\text{YM}} + (\text{ permutations}) \implies 0$$
(4.25)

For the complete amplitude we must take into account that the complete transmutation sums over different pair arrangements of these transmutations.

4.3.1 τ^0 : Yang-Mills-Scalar

Starting from $\mathcal{O}(\tau^0)$, we find the 6 point YMS amplitude. An example calculation of one such an arrangement is given by the arrangement (12)(34)(56) which results in the contribution

$$4A_{\rm YMS}(\phi_1\phi_2,\phi_3\phi_4,\phi_5\phi_6) = \frac{s_{2,3}}{s_{1,2}s_{1,2,3}} + \frac{s_{4,5}s_{2,3}}{s_{1,2}s_{1,2,3}s_{1,2,3,4}} + \frac{1}{s_{1,2,3}} + \frac{-s_{2,4}s_{3,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{1,2}s_{3,4}s_{3,4,5}} \\ + \frac{s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{1,2}s_{4,5}s_{3,4,5}} + \frac{-s_{1,4}s_{2,5} - s_{2,4}s_{2,5} - s_{2,4}s_{3,5} + s_{2,3}s_{4,5}}{s_{1,2}s_{3,4}s_{1,2,3,4}} + \frac{s_{4,5}}{s_{1,2}s_{3,4}s_{1,2,3,4}} \\ + \frac{-s_{2,4}s_{2,5} - s_{2,4}s_{3,5} + s_{2,3}s_{4,5}}{s_{3,4}s_{2,3,4}s_{1,2,3,4}} + \frac{1}{s_{2,3,4,5}} + \frac{s_{4,5}}{s_{2,3,4}s_{2,3,4,5}} + \frac{-s_{2,4}s_{2,5} - s_{2,4}s_{3,5} + s_{2,3}s_{4,5}}{s_{3,4}s_{2,3,4}s_{2,3,4,5}} \\ + \frac{-s_{2,4}s_{3,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{3,4}s_{3,4,5}s_{2,3,4,5}} + \frac{s_{4,5}}{s_{2,3}s_{4,5}s_{2,3,4,5}} + \frac{s_{4,5}s_{4,5} + s_{2,5}s_{4,5}}{s_{4,5}s_{3,4,5}s_{2,3,4,5}} \\ + \frac{-s_{2,4}s_{3,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{3,4}s_{3,4,5}s_{2,3,4,5}} + \frac{s_{4,5}s_{4,5} + s_{2,5}s_{4,5}}{s_{4,5}s_{3,4,5}s_{2,3,4,5}} \\ + \frac{-s_{2,4}s_{3,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{3,4}s_{3,4,5}s_{2,3,4,5}} + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{3,4,5}s_{2,3,4,5}} \\ + \frac{-s_{2,4}s_{3,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{4,5}s_{3,4,5}s_{2,3,4,5}} + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{3,4,5}s_{2,3,4,5}} \\ + \frac{-s_{2,4}s_{3,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{4,5}s_{3,4,5}s_{2,3,4,5}} \\ + \frac{-s_{4,5}s_{4,5} + s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{3,4,5}s_{2,3,4,5}} + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}s_{4,5}} \\ + \frac{-s_{4,5}s_{4,5} + s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}s_{4,5}} \\ + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}}{s_{4,5}s_{4,5}s_{4,5}s_{4,5}} + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}s_{4,5}} \\ + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}} + \frac{s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}} + \frac{s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}} \\ + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}} + \frac{s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}} + \frac{s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}} \\ + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s$$

up to a factor of 4 due to the replacement of the momentum factor $p_i p_j$ that was present in numerator of the gluon amplitude to $p_i p_j \rightarrow s_{ij}/2$. The complete amplitude is the summation over all 15 arrangements and is given in Appendix E.

4.3.2 Mixed Amplitudes

We then turn our attention to the $\mathcal{O}(\tau^1)$ contribution, supposedly describing pions coupled to scalars. The arrangement (12)(34)(56) contributes the following types of amplitudes to the full amplitude

$$\tau[A(\phi_1\phi_2,\phi_3\phi_4,\pi_5,\pi_6) + A(\phi_1\phi_2,\pi_3,\pi_4,\phi_5\phi_6) + A(\pi_1,\pi_2,\phi_2\phi_3,\phi_5\phi_6)], \tag{4.27}$$

and similarly for the other 14 arrangements of the 6 legs. These are calculated starting from the YMS amplitude of every arrangement. For every arrangement of legs, there are three possible mandelstams s_{ij} with which the YMS can be multiplied. This results in a total of 45 contributions at this order in τ . This calculation requires an algorithm in order to correctly keep track of the combinatorics. The full expression is too large to display here. We can however analyse the contributions that appear in the amplitude, e.g. by collecting terms that share a propagator.

The propagators for 6-point diagrams are products of the Mandelstams that enter the cubic vertices. As an example we consider the half-ladder and snowflake diagrams of Figure 4.1. The half-ladder diagram will have a propagator of the form $D_{\rm hl} = s_{ab}s_{abcc}s_{abcd}$ and the



Figure 4.1: Half-ladder and snowflake diagram for a 6-point amplitude.

snowflake diagram will have a propagator contribution of the form $D_{\text{snowflake}} = s_{ab}s_{cd}s_{ef}$. The terms that share the propagator D_{hl} can be collected, note that we have to be mindful that we also have to account for terms that contain $D = s_{1234}s_{123}$, as these are the result of a cancellation of the term $s_{12}/(s_{12}s_{1234}s_{123})$. Collected these give

$$\frac{s_{45}\left(s_{12}^2 + s_{23}\left(s_{34} + s_{56}\right) + s_{12}\left(s_{23} - 2s_{13} - s_{24} + s_{34}\right)\right)}{s_{12}s_{12}s_{123}s_{1234}}.$$
(4.28)

For the snowflake diagrams we have to consider that by construction of the initial YM amplitudes there are Mandelstams present of the form s_{i6} . However, the package that provides the YM amplitudes will have replaced the momentum p_6 using momentum conservation $p_6 = -(p_1 + p_2 + p_3 + p_4 + p_5)$. We have to use a similar replacement fo the propagators of the snowflake diagrams. These will be altered to $D_{\text{snowflake}} = s_{ab}s_{cd}s_{abcd}$. As an example, the terms that share the propagator $D = s_{12}s_{34}s_{1234}$ are

$$-\frac{(s_{14}s_{25}+s_{24}(s_{25}+s_{35})-s_{23}s_{45})s_{56}}{s_{12}s_{34}s_{1234}}=-\frac{s_{14}s_{25}+s_{24}(s_{25}+s_{35})-s_{23}s_{45}}{s_{12}s_{34}}.$$
 (4.29)

In order to compare all the terms of the the 6 point partial amplitude to the amplitude obtained from the BCJ bootstrap, we will compare the factorization channels of the partial amplitude. This comparison will be done in section 5.1.

4.3.3 Mandelstam Order of 6-point Amplitude

The counting parameter τ allows us to keep track of how many orders of Mandelstam invariants we introduce in the generalised transmutation process. Here we summarise the order of Mandelstams at different orders in τ :

 $\mathcal{O}(\tau^0)$: The contributions at this order for 6-point are purely Yang-Mills scalar amplitudes. All polarisation pairs have been set to 1 according to the arrangements of (4.3). The resulting soft limit $\lim_{p_i\to 0} A_{\text{YMS}}(\phi_1\phi_2,\phi_3\phi_4,\phi_5\phi_6) \sim s^{-1}$. This is in agreement with theory, where every vertex contributes $\propto p$ and every propagator contributes $\propto s^{-1}$. The half-ladder diagrams contain 4 vertices and 3 propagators, together this makes 1/s scaling of the amplitude.

 $\mathcal{O}(\tau^1)$: The amplitudes of this order are created from the above YMS amplitude by multiplying with one Mandelstam s_{ij} . By this reasoning and by inspection of the amplitude the resulting scaling in Mandelstams can be seen to be combination of s^3/s^3 , leading to a scaling of s^0 .

 $\mathcal{O}(\tau^2)$: These amplitudes gain an additional mandelstam s_{ij} to the above, therefore the scaling is of one order higher, s^1 . we have explicitly checked that the amplitude generated by \mathbb{T} at this order give rise to the correct NLSM amplitude, up to a rescaling factor of $-\frac{16}{5}$ that arises from the fact that the same permutation invariant amplitude arises multiple times in the amplitude and some factors of 2 from the definitions of $s_{ij} = 2p_ip_j$.

 $\mathcal{O}(\tau^3)$: The composite amplitudes of each arrangement will have a scaling of s^2 . We have verified explicitly that the summation over the GT of all arrangements results in a vanishing amplitude.

Chapter 5

Discussion

5.1 Partial 6-point Amplitude of BCJ Bootstrap

We wish to compare the amplitude obtained from the GT-operator with the amplitude that is obtained from the BCJ bootstrap of chapter 2. The latter however, contains colour factors in the amplitude, while the former does not. In order to make our comparison, we recall that we applied the GT-operator on a colour-ordered partial Yang-Mills amplitude $A_6^{\rm YM}$ [123456]. Each partial amplitude is the kinematic weight of a specific trace colour factor according to the trace decomposition

$$\mathcal{A}_{m}^{\text{tree}} = g^{m-2} \sum_{\sigma \in S_{m-1}} A_{m}^{\text{tree}}(1, \sigma(2), \sigma(3), \dots, \sigma(m-1), m) \operatorname{Tr} \left(T^{a_{1}} T^{a_{\sigma(2)}} T^{a_{\sigma(3)}} \cdots T^{a_{\sigma(m)}} \right).$$
(5.1)

Due to this decomposition we therefore know that in the full amplitude, this partial amplitude $A_6^{\text{YM}}[123456]$ was initially accompanied by the colour trace $\text{Tr} [T^{a_1}T^{a_2}T^{a_3}T^{a_4}T^{a_5}T^{a_6}]$. We can therefore compare the resulting partial amplitude $\mathbb{T}^{(123456)} \cdot A_6^{\text{YM}}[123456]$ to the partial amplitude from the BCJ bootstrap that has the same colour trace $A_6^{\text{BCJ}}[123456]$.

In order to to calculate this partial amplitude, we first need to calculate an expression for the BCJ numerators in the latter method. To do this, we only need to find one numerator, N_{abcdef} , and the other numerators will be permutations of the indices. The numerator N_{abcdef} can be found by decomposing the total amplitude into partial amplitudes in the, by now familiar, way

$$\mathcal{A}_6 = \frac{c_{123456}N_{123456}}{D_{123456}} + \frac{c_{123465}N_{123465}}{D_{123465}} + \dots$$
(5.2)

However we know that c_{123456} is not the only colour factor that will contain trace ordering [123456] in it's decomposition. Therefore we first decompose the total amplitude in terms

of the partial amplitude in the trace decomposition of the colour factors. :

$$\mathcal{A}_{6} = \operatorname{Tr}\left[T^{a_{1}}T^{a_{2}}T^{a_{3}}T^{a_{4}}T^{a_{5}}T^{a_{6}}\right] \left[\frac{N_{123456}}{D_{123456}} + \ldots + \frac{N_{123465}}{D_{123465}} + \ldots\right] + \ldots$$
(5.3)

To calculate which diagrams contribute to the partial amplitude, we start from the possible 6-point colour factor for a half ladder diagram.

$$c_1 = \tilde{f}^{a_1 a_2 b} \tilde{f}^{b a_3 c} \tilde{f}^{c a_4 a_d} \tilde{f}^{d a_5 a_6}.$$
(5.4)

We have written a Mathematica algorithm that performs the process of (1.29) at n point. This decomposes the colour factor c_1 into following 16 trace contributions:

$$c_{1} = \operatorname{Tr} \left[T^{a_{1}} T^{a_{2}} T^{a_{3}} T^{a_{4}} T^{a_{5}} T^{a_{6}} \right] - \operatorname{Tr} \left[T^{a_{1}} T^{a_{2}} T^{a_{3}} T^{a_{4}} T^{a_{6}} T^{a_{5}} \right] - \operatorname{Tr} \left[T^{a_{4}} T^{a_{1}} T^{a_{2}} T^{a_{3}} T^{a_{5}} T^{a_{6}} \right] \\ + \operatorname{Tr} \left[T^{a_{4}} T^{a_{1}} T^{a_{2}} T^{a_{3}} T^{a_{6}} T^{a_{5}} \right] + \operatorname{Tr} \left[T^{a_{3}} T^{a_{1}} T^{a_{2}} T^{a_{4}} T^{a_{5}} T^{a_{6}} \right] - \operatorname{Tr} \left[T^{a_{3}} T^{a_{1}} T^{a_{2}} T^{a_{4}} T^{a_{6}} T^{a_{5}} \right] \\ - \operatorname{Tr} \left[T^{a_{4}} T^{a_{3}} T^{a_{1}} T^{a_{2}} T^{a_{5}} T^{a_{6}} \right] + \operatorname{Tr} \left[T^{a_{4}} T^{a_{3}} T^{a_{1}} T^{a_{2}} T^{a_{6}} T^{a_{5}} \right] - \operatorname{Tr} \left[T^{a_{2}} T^{a_{1}} T^{a_{3}} T^{a_{4}} T^{a_{5}} T^{a_{6}} \right] \\ + \operatorname{Tr} \left[T^{a_{2}} T^{a_{1}} T^{a_{3}} T^{a_{4}} T^{a_{6}} T^{a_{5}} \right] + \operatorname{Tr} \left[T^{a_{4}} T^{a_{2}} T^{a_{1}} T^{a_{3}} T^{a_{5}} T^{a_{6}} \right] - \operatorname{Tr} \left[T^{a_{4}} T^{a_{2}} T^{a_{1}} T^{a_{3}} T^{a_{6}} T^{a_{5}} \right] \\ - \operatorname{Tr} \left[T^{a_{3}} T^{a_{2}} T^{a_{1}} T^{a_{4}} T^{a_{5}} T^{a_{6}} \right] + \operatorname{Tr} \left[T^{a_{3}} T^{a_{2}} T^{a_{1}} T^{a_{4}} T^{a_{6}} T^{a_{5}} \right] - \operatorname{Tr} \left[T^{a_{4}} T^{a_{3}} T^{a_{2}} T^{a_{1}} T^{a_{5}} T^{a_{6}} \right] \\ + \operatorname{Tr} \left[T^{a_{4}} T^{a_{3}} T^{a_{2}} T^{a_{1}} T^{a_{6}} T^{a_{5}} \right] . \tag{5.5}$$

We can perform permutations of the indices of the structure constants of c_1 to get the trace decomposition of all other possible colour factor. We then construct the total amplitude in terms of the unspecified numerators and denominators and extract the coefficient of the terms that contain $Tr \left[T^{a_1}T^{a_2}T^{a_3}T^{a_4}T^{a_5}T^{a_6}\right]$ and it's cyclic permutations and reflections in order to obtain all the combination of N/D that contribute to the partial amplitude. A similar analysis needs to be done for the colour factors of the snowflake diagrams. Finally, we would need to consider that these diagrams are not all distinct, but linked by Jacobi relations. Therefore, there is a redundancy in the above calculation.

An alternative method is to use FeynCalc to calculate the full amplitude, which already takes care of the redundancy and output the amplitude in terms of the master colour factors. After this, we need to calculate the trace decomposition of these colour factors to calculate the partial amplitude ¹.

¹Kind regards to Yang Li for providing us with this partial amplitude.

The partial amplitude we are interested in is then denoted as

$$A_{6}[123456] = \frac{N_{abcdef}}{s_{ab}s_{ef}(s_{ab} + s_{ac} + s_{bc})} + \frac{N_{abcdef} - N_{abdcef}}{s_{ab}s_{cd}s_{ef}} + \frac{N_{afbecd}}{s_{af}s_{cd}(s_{cd} + s_{ce} + s_{de})} \\ + \frac{N_{afebcd}}{s_{af}s_{cd}(s_{bc} + s_{bd} + s_{cd})} + \frac{N_{cdefab}}{s_{ab}s_{cd}(s_{cd} + s_{ce} + s_{de})} + \frac{N_{defabc}}{s_{bc}s_{de}(s_{ab} + s_{ac} + s_{bc})} \\ + \frac{N_{efabcd}}{s_{cd}s_{ef}(s_{bc} + s_{bd} + s_{cd})} - \frac{N_{bcdeaf}}{s_{af}s_{bc}(s_{bc} + s_{bd} + s_{cd})} - \frac{N_{afbcde} - N_{afcbde}}{s_{af}s_{bc}s_{de}} \\ - \frac{N_{abcfde}}{s_{ab}s_{de}(s_{ab} + s_{ac} + s_{bc})} - \frac{N_{abfcde}}{s_{ab}s_{de}(s_{cd} + s_{ce} + s_{de})} - \frac{N_{afbcde}}{s_{af}s_{bc}(s_{cd} + s_{ce} + s_{de})}$$

$$- \frac{N_{efdabc}}{s_{bc}s_{ef}(s_{ab} + s_{ac} + s_{bc})} - \frac{N_{efadbc}}{s_{bc}s_{ef}(s_{bc} + s_{bd} + s_{cd})},$$

$$(5.6)$$

where the numerators of the snowflake diagram are clearly taken into account.

The cubic 6-point numerator $N_6^{(3)}$ is obtained through the methodology of subsection 2.4.2, where the general expression for $N_6^{(3)}$ is given by the expression

$$N_{abcdef}^{(3)} = (s_{ac} - s_{bc})s_{ab}(s_{de} - s_{df}) + (s_{ac} - s_{bc})(s_{de} - s_{df})s_{ef} + s_{abc}Q_{abcdef},$$
(5.7)

where Q_{abcdef} is the ansatz of a term that is quadratic in Mandelstams. As discussed, after imposing the BCJ conditions there are 3 unfixed parameters. The resulting Q_{abcdef} is used to construct the numerator $N_{abcdef}^{(3)}$ with which te partial amplitude is calculated. We scale the momentum of leg *a* through scaling of the Mandelstams $s_{ai} \mapsto xs_{ai}$. The resulting amplitude will have contain terms that are of $\mathcal{O}(x^{-1})$. We impose the soft limit of $\sigma = 0$ by requiring that the coefficients of these terms are tuned such that the $\mathcal{O}(x^{-1})$ contribution vanishes. This leaves an amplitude with leading soft behaviour of $\mathcal{O}(x^0)$. This imposing of the soft limit fixes all of the remaining coefficients.

Using these numerators we can construct the partial amplitude of (5.6) by summing over all the permutations of the indices *abcdef*. We should check that this amplitude correctly factorises into the desired 4-point partial amplitudes on the imposed factorisation channel $s_{abc} \rightarrow 0$. On this factorisation channel, the amplitude factorises as

$$A_6[123456] \to A_4^{(1)}[123x] \frac{1}{s_{abc}} A_4^{(2)}[x456] + A_4^{(2)}[123x] \frac{1}{s_{abc}} A_4^{(1)}[x456], \tag{5.8}$$

where $A^{(p)}[ijkl]$ denotes the 4-point partial amplitude constructed from the numerator $N_4^{(p)}$. These partial amplitudes are given by Equation 1.31 as

$$A_4^{(p)}[1234] = \frac{N_s^{(p)}}{s} - \frac{N_t^{(p)}}{t}.$$
(5.9)

We have checked that the residue of the partial amplitude on the factorisation channel reduces to (5.8) up to a minus-sign. Computationally it is beneficial to calculate the residue by setting $s_{abc} \rightarrow \lambda$ and doing a series expansion around $\lambda = 0$. Once this is done, the coefficient of the $\mathcal{O}(\lambda^{-1})$ term is equal to the residue.

5.2 Factorization Comparison

The amplitude obtained from the generalised transmutation has been denoted in terms of the 9 independent Mandelstam basis $\{s_{a,b}, s_{a,c}, s_{a,d}, s_{a,e}, s_{b,c}, s_{b,d}, s_{b,e}, s_{c,d}, s_{c,e}\}$. The residue of this amplitude on the factorisation channel $s_{abc} = s_{ab} + s_{ac} + s_{bc}$ is given by

$$\lim_{s_{abc} \to 0} s_{abc} A_{6}^{GT} = -\frac{3(s_{ab} + s_{bc})(s_{ae} + s_{be} + s_{ce})}{s_{ad} + s_{bd} + s_{cd}} - \frac{s_{cd}(s_{ab} + s_{bc})(s_{ae} + s_{be} + s_{ce})}{s_{ab}(s_{ad} + s_{bd} + s_{cd})} - \frac{s_{ad}(s_{ab} + s_{bc})(s_{ae} + s_{be} + s_{ce})}{s_{bc}(s_{ad} + s_{bd} + s_{cd})} + \frac{3(s_{ab} + s_{bc})(s_{ae} + s_{be} + s_{ce})}{s_{ad} + s_{ae} + s_{bd} + s_{be} + s_{cd} + s_{ce}} + \frac{(s_{ab} + s_{bc})(s_{ae} + s_{be} + s_{ce})(s_{ab} - s_{cd} - s_{ce})}{s_{ab}(s_{ad} + s_{ae} + s_{bd} + s_{be} + s_{cd} + s_{ce})} - \frac{(s_{ab} + s_{bc})(s_{ae} + s_{be} + s_{cd} + s_{ce})}{s_{bc}(s_{ad} + s_{ae} + s_{bd} + s_{be} + s_{cd} + s_{ce})} + \frac{s_{bd}(s_{ae} + s_{be} + s_{cd} + s_{ce})}{s_{ad} + s_{bd} + s_{cd}} + \frac{(s_{ae} + s_{be} + s_{ce})(s_{ab} + s_{bc} + s_{bd} + s_{be})}{s_{ad} + s_{ae} + s_{bd} + s_{be} + s_{cd} + s_{ce}}} - \frac{s_{ce}(s_{ab} + s_{bc})}{s_{ab}} - \frac{s_{ce}(s_{ab} + s_{bc})}{s_{ab}}}{s_{bc}} + 3(s_{ae} + s_{be} + s_{ce}) - 3(s_{ab} + s_{bc}) + s_{be}. \quad (5.10)$$

The amplitude from the BCJ bootstrap $A_6^{\rm BC}$ was confirmed to factorise as (5.8). Explicitly the residue is denoted as

$$\lim_{s_{abc} \to 0} s_{abc} A_6^{\text{BCJ}} = -\frac{(s_{ab} + 2s_{bc})(s_{ad} - s_{ae} + s_{bd} - s_{be} + s_{cd} - s_{ce})}{s_{ab}} \\
-\frac{(2s_{ab} + s_{bc})(s_{ad} - s_{ae} + s_{bd} - s_{be} + s_{cd} - s_{ce})}{s_{bc}} + \frac{(2s_{ab} + s_{bc})(s_{ad} - s_{ae} + s_{bd} - s_{be} + s_{cd} - s_{ce})}{s_{ad} + s_{ae} + s_{bd} + s_{be} + s_{cd} + s_{ce}} \\
+\frac{(s_{ab} + 2s_{bc})(s_{ad} - s_{ae} + s_{bd} - s_{be} + s_{cd} - s_{ce})}{s_{ad} + s_{ae} + s_{bd} + s_{be} + s_{cd} + s_{ce}} + \frac{(2s_{ab} + s_{bc})(s_{ad} + 2s_{ae} + s_{bd} + 2s_{be} + s_{cd} + 2s_{ce})}{s_{bc}} \\
+\frac{(s_{ab} + 2s_{bc})(s_{ad} + 2s_{ae} + s_{bd} + 2s_{be} + s_{cd} + 2s_{ce})}{s_{ab}} + \frac{(2s_{ab} + s_{bc})(s_{ad} + 2s_{ae} + s_{bd} + 2s_{be} + s_{cd} + 2s_{ce})}{s_{ad} + s_{bd} + s_{cd}} \\
+\frac{(s_{ab} + 2s_{bc})(s_{ad} + 2s_{ae} + s_{bd} + 2s_{be} + s_{cd} + 2s_{ce})}{s_{ad} + s_{bd} + s_{cd}}}, \quad (5.11)$$

where it can be seen that there are multiple contributions with the same denominator, but which cannot be expressed together as this would not fit on the page. In total there are 4 terms, which corresponds to the four terms obtained from the product of (5.9). We can distinctly identify the four-point

On the factorsiation pole s_{abc} these residues differ. There are several key aspects that can be identified. The first being that there are more exchange channels present in $A_6^{\rm GT}$ that are not present in $A_6^{\rm BCJ}$. check factorisation into YMS partial amplitude times pion amplitude We also note that there are some similarities, which become clearer once we take further take the residue on the channel $s_{ab} \rightarrow 0$,

$$\lim_{s_{ab}\to 0} \lim_{s_{abc}\to 0} A_6^{\text{GT}} = -\frac{s_{bc}s_{cd}\left(s_{ae} + s_{be} + s_{ce}\right)}{4\left(s_{ad} + s_{bd} + s_{cd}\right)} - \frac{\left(s_{cd} + s_{ce}\right)s_{bc}\left(s_{ae} + s_{be} + s_{ce}\right)}{4\left(s_{ad} + s_{ae} + s_{bd} + s_{be} + s_{cd} + s_{ce}\right)} - \frac{3}{4}s_{bc}\left(s_{ae} + s_{be} + s_{ce}\right) - \frac{1}{4}s_{bc}s_{ce},$$
(5.12)

$$\lim_{s_{ab} \to 0} \lim_{s_{abc} \to 0} A_6^{\text{BCJ}} = 6s_{bc} \left(s_{ae} + s_{be} + s_{ce} \right).$$
(5.13)

In this limit, it becomes apparent that both partial amplitudes have a common contribution $s_{bc} (s_{ae} + s_{be} + s_{ce})$ up to a factor. The BCJ-bootstrap amlitude has no remaining exchange channels, but the GT amplitude clearly does.

It would be interesting to investigate what kind of diagrams contribute to these factorisation limits. We first note that we can recognise two partial amplitudes in the factorisation procedure. The first is the 4-point Yang-Mills-Scalar gluon exchange channel

$$A_4^{\rm YMS}(\phi_a \phi_b, \phi_c \phi_x) = \frac{(s_{ab} + s_{bc})}{s_{ab}},$$
(5.14)

which is can be found in both (5.10) and (5.11). The diagram corresponding to this amplitude is shown in Figure ?? Specifying leg x is not necessary due to momentum conservation. We will therefore identify leg x as an internal leg.

The second type of contribution we encounter in both of the residues is the 4-point NLSM amplitude consisting of the legs d, e and f and an 'internal' leg x

$$A_4^{\text{NLSM}}(\pi_d, \pi_e, \pi_f, \pi_x) = (s_{de} + s_{ef}) = (s_{ae} + s_{be} + s_{ce}).$$
(5.15)

Note that this type of amplitude is equal to an amplitude of 2 scalars interacting with 2 pions of the form $A(\phi, \phi, \pi, \pi)$ that is shown in Figure 5.1c.

The only factor that remains to be identified is the factors of s_{bc} in (5.13). We find that this factor is precisely the residue of the YMS amplitude in (5.14) on the channel $s_{ab} \rightarrow 0$

$$\lim_{s_{ab} \to 0} s_{ab} A_4^{\text{YMS}}(\phi_a \phi_b, \phi_c \phi_x) = s_{bc}.$$
 (5.16)

We can therefore conclude that a diagram such as the diagram displayed in Figure 5.1a is responsible for the residue in (5.13) and for a section of the residue of (5.12). It is not suprising that these are the factorisation structures that occur. After all, one of the key assumptions that we imposed on the numerator $N_6^{(3)}$ is that it factorises into the product of two four-point amplitudes. The 4-point numerators that are combined to make the cubic order numerator are $N_4^{(1)}$ and $N_4^{(2)}$, which individually created the 4-point gluon exchange channel and the 4-point NLSM contact term respectively. This exactly matches the factorisation we have found.



Figure 5.1: Figure (a) shows the type of diagram that can contribute to the residue of both A_6^{GT} and A_6^{BCJ} . On the factorisation limit where $s_{abc} \rightarrow 0$ this diagram splits into the product of the 4-point YMS amplitude shown in (b) and the 4-point NLSM amplitude shown in (c).

The remaining contributions to the residue of A_6^{GT} are interesting to study as well, as these are apparently not produced through the BCJ bootstrap method. We can see that after imposing the factorisation of s_{abc} there are a few terms that do not have any remaining propagation channels. These contributions can also be seen to of the form of the 4-point NLSM amplitudes in (5.15) multiplied by a constant, which could indicate a ϕ^4 scalar contact term. Such contributions could then potentially arise from the factorisation of the diagram in Figure 5.2.



Figure 5.2: Diagram that factorises into a quartic ϕ^4 contact term amplitude and a 4-point NLSM contact term.

It is interesting that such contact terms would arise through transmutation for multiple reasons. The first being that such contact terms do not arise during the transmutation at 4-point partial amplitudes that we discussed in subsection 3.2.5.

The second reason is that such ϕ^4 contact terms are not discussed in the original publication by Cheung and collaborators [22], presumably because they do not occur in typical YMS formulations [18, 80]. However, these are explicitly mentioned to occur when discussing transmutation operators in the CHY representation [81]. It should be noted that these contact terms also deviate from the 'extended NLSM' described in [35] and [69], where mixed amplitudes of pions and scalar are obtained by extending the U(N) NLSM to include a cubic biadjoint scalar self-interaction. It should be noted that these theories do not include gauge interactions. 8

It is not a suprise that these quartic contact terms do not arise from the BCJ bootstrap method. The numerator in this framework that gives rise to a contact diagram at four point is $N_4^{(2)}$, but this gives rise to a NLSM $\phi^2(\partial \phi)^2$ contact term, not a ϕ^4 term. The only other 4-point amplitudes that we can construct in this framework are the gluon exchange amplitudes.

There is another combination of factorisation channels that is interesting to check. We will again investigate the limit where two internal propagators go on-shell, but this time we will set $s_{ab} \rightarrow 0$ and $s_{ef} \rightarrow 0$. The residue of both approaches on these limits is

$$\lim_{s_{ab}\to 0} \lim_{s_{ef}\to 0} A_6^{\text{GT}} = \frac{s_{b,c}s_{c,d} \left(s_{a,e} + s_{b,e} + s_{c,e}\right)}{s_{a,d} + s_{b,d} + s_{c,d}} - s_{b,c} \left(s_{a,e} + s_{b,e} + s_{c,e}\right),$$

$$= s_{a,d}s_{b,c} - s_{b,d}s_{b,c} - s_{b,d}s_{b,c}, \qquad (5.17)$$

$$\lim_{s_{ab} \to 0} \lim_{s_{ef} \to 0} A_6^{\text{BCJ}} = \frac{3\left(s_{a,d} + s_{b,d}\right)\left(s_{a,e}\left(s_{b,c} + s_{b,d}\right) + s_{b,e}\left(s_{b,c} + s_{b,d} + s_{c,d}\right)\right)}{s_{c,d}}.$$
 (5.18)

We find that there on this limit, both amplitudes still allow for an exchange channel. As the only exchange channel left in the BCJ amplitude is of the form s_{ij} , we conclude that this residue is derived from a snowflake diagram that has $D \sim \frac{1}{s_{ab}s_{cd}s_{ef}}$.

For the amplitude obtained through generalised transmutation, there is also a contribution to the residue with no exchange channels left, identical to the contributions to the residue of (5.11) and a new type of residue that could arise from a diagram factorised on two 4-point YMS exchange channels in a snowflake diagram. The remaining amplitude should contribute no additional momenta. This is possible if the contributions from the pions cancel the remaining exchange pole, similar to how the longitudinal operator turns an amplitude with an exchange channel of the form $A(\phi, \phi, g, g)$ into a contact diagram of the form $A(\pi, \pi, \pi, \pi)$. A proposal for such a diagram is shown in Figure 5.3. This interaction would require a Lagrangian that includes the interaction $(D\phi)^2 F^2$, hinting at a higherderivative correction to the gNLSM theory that differs from the correction considered in section 2.4. It would be interesting to study this in more detail, but we leave this for future work.



Figure 5.3: Proposed diagram before (left) and after (right) longitudinal transformation, that gives rise to the $s_{ij}s_{kl}$ residues on the factorisation limits $s_{ab} \to 0$ and $s_{ef} \to 0$

Chapter 6

Conclusion and Outlook

In conclusion, this thesis provides an in-depth investigation into the Unifying Relations for scattering amplitudes and the recent BCJ bootstrap that can be used to construct kinematic numerators that conform to the duality between colour and kinematics. Prior to this, we performed an extensive analysis of gluon (Yang-Mills) amplitudes and pion (NLSM) amplitudes, exploring the duality between colour and kinematics, as well as examining the behaviour of these amplitudes in special kinematic limits.

We have extensively discussed the construction of amplitudes of the gauged nonlinear sigma-model using the BCJ bootstrap method. This theory appeared to have a hybrid soft degree, where a subsector of the amplitudes generated from the theory obey a soft degree of $\sigma = 0$, and another subsector generates pion contact terms that obey the Adler zero of $\sigma = 1$.

Using the BCJ bootstrap, we have constructed the $\sigma = 0$ amplitude that occurs at 6point. To perform this derivation, we have imposed the BCJ conditions on an ansatz for numerators composed of Mandelstam variables in a specific order. The coefficients of the ansatz were fixed by imposing the soft degree of $\sigma = 0$. The amplitudes that are constructed describe adjoint Goldstone scalars ϕ^a interacting with gluons.

We aimed to answer the question

How can we construct a transmutation operator that generates amplitudes for comparison with those derived from the BCJ bootstrap?

This was achieved by generalising the transmutation relations proposed by Cheung et al. such that we were able to generate mixed amplitudes of scalars and pions that exchange gluons through transmutation of Yang-Mills amplitudes. This generalised transmutation (GT) operator was constructed such that we were able to keep track of the order of Mandelstam variables that were introduced to the transmuted amplitudes, allowing us to determine whether we had achieved the $\sigma = 0$ soft degree that we aimed to construct. Using this GT-operator, explicit calculations of the transmutation of 6-point Yang-Mills partial amplitude were performed using computational tools. At different orders in Mandelstam invariants, tracked by the parameter τ , these gave rise to the correct expressions for Yang-Mills-Scalar (YMS) at $\mathcal{O}(\tau^0)$ and Nonlinear Sigma Model (NLSM) amplitudes at $\mathcal{O}(\tau^2)$. The vanishing of the amplitude at $\mathcal{O}(\tau^3)$, when the longitudinal operator \mathcal{L} is applied to all external gluons, i.e., when all pairs of polarisations are set to pairs of momenta, was also verified computationally at 6-point.

This leaves the $\mathcal{O}(\tau^1)$ amplitude that contains the $\sigma = 0$ soft degree to be analysed, which brings us to our second research question

What types of amplitudes can be derived from a generalised transmutation? operator?

The answer to this question can be found by studying the factorisation of the resulting amplitude. In the limit where the internal exchange particles are taken to be on-shell we found the GT-amplitudes to mostly adhere to familiar lower point amplitudes. Examples of these lower-point amplitudes are the 4-point YMS amplitude, where two pairs of scalars exchange a gluon and the 4-point NLSM contact term. Beyond this, we unexpectedly encountered the 4-point ϕ^4 , which does not typically occur in YMS theories but which can occur in the CHY formalism. We also interpreted that a $(\partial \phi)^2 A^2$ contact term is necessary for the factorisation of 6-point snowflake-diagrams.

How can generalized transmutation operators inform us about the uniqueness of the theory developed via the BCJ bootstrap? operator?

In generality the generalised transmutation operator fails to land on the same partial amplitude as the amplitude obtained from the BCJ bootstrap method. However, there are common structures in the amplitudes of the two different methods. This indicates that generalised transmutation is able to reconstruct some of the interactions that occur in the BCJ bootstrap in the $\sigma = 0$ sector, namely the set of amplitudes that factorise into the 4-point YMS exchange channel and the 4-point pion contact term. The snowflake diagrams of the 6-point BCJ bootstrap-amplitude were also identified, but cannot be matched to amplitudes obtained from GT due to the cancellation of one of the gluon exchange channels when two external gluons are transmuted into pions.

This leads us to conclude that the amplitudes of the BCJ bootstrap are not uniquely fixed by their soft degree of $\sigma = 0$, as opposed to amplitudes such as the NLSM and SG that are fully determined through its soft behaviour.

It would be interesting to study how the amplitudes obtained from GT compare to the higher-derivative corrections to Yang-Mills discussed in Equation 1.5.3. Investigating the behaviour of generalised transmutation on these higher-derivative corrections could also provide valuable insights into CK-duality compatibility between numerators of different theories. It would also be intriguing to explore the transmutation of these amplitudes and see if this contributes additional terms to the $\sigma = 0$ GT amplitude. Finally, comparing

the obtained amplitudes to the field theory limit of the amplitudes of Z-theory [59] could strengthen the approach.

On a broader scale, it would be valuable to extend the investigation of transmutation operators. First, extending the transmutation calculations to include higher-order Mandelstam variables, i.e. a transmutation of $(e_i e_j) \mapsto (1 + s + s^2 + ...)$ instead of purely (1 + s)would be an interesting avenue to explore. Additionally, applying generalised transmutation techniques to the amplitudes of the other theories in Figure 3.2 presents an intriguing possibility for further study. Such transmutations could lead to mixed amplitudes of Born-Infeld photons interacting with Maxwell photons that interact through graviton exchange through GT of graviton amplitudes.

Bibliography

- Paul Dirac. The quantum theory of the emission and absorption of radiation. Proceedings of the Royal Society of London. Series A, Containing Papers of a Mathematical and Physical Character, 114(767):243-265, March 1927. ISSN 0950-1207, 2053-9150. doi: 10.1098/rspa.1927.0039. URL https://royalsocietypublishing.org/doi/10. 1098/rspa.1927.0039.
- Michael E. Peskin and Daniel V. Schroeder. An Introduction to quantum field theory. Addison-Wesley, Reading, USA, 1995. ISBN 978-0-201-50397-5 978-0-429-50355-9 978-0-429-49417-8. doi: 10.1201/9780429503559.
- D. [3] Matthew Schwartz. Field Theory the Stan-Quantum and December 2013.URL https://www.cambridge.org/ dard Model, highereducation/books/quantum-field-theory-and-the-standard-model/ A4CD66B998F2C696DCC75B984A7D5799. ISBN: 9781108985031 Publisher: Cambridge University Press.
- [4] Karen Crowther and Niels Linnemann. Renormalizability, fundamentality and a final theory: The role of UV-completion in the search for quantum gravity, July 2017. URL http://arxiv.org/abs/1705.06777. arXiv:1705.06777 [gr-qc, physics:physics].
- [5] Zvi Bern, John Joseph M. Carrasco, and Henrik Johansson. Perturbative Quantum Gravity as a Double Copy of Gauge Theory. *Physical Review Letters*, 105(6):061602, August 2010. ISSN 0031-9007, 1079-7114. doi: 10.1103/PhysRevLett.105.061602.
 URL https://link.aps.org/doi/10.1103/PhysRevLett.105.061602.
- [6] Z. Bern, J. J. M. Carrasco, L. J. Dixon, H. Johansson, and R. Roiban. Simplifying multiloop integrands and ultraviolet divergences of gauge theory and gravity amplitudes. *Physical Review D*, 85(10):105014, May 2012. ISSN 1550-7998, 1550-2368. doi: 10.1103/PhysRevD.85.105014. URL https://link.aps.org/doi/10. 1103/PhysRevD.85.105014.
- [7] Alessandra Buonanno, Mohammed Khalil, Donal O'Connell, Radu Roiban, Mikhail P. Solon, and Mao Zeng. Snowmass White Paper: Gravitational Waves and Scattering Amplitudes, April 2022. URL http://arxiv.org/abs/2204.05194. arXiv:2204.05194 [gr-qc, physics:hep-th].

- [8] David A. Kosower, Ricardo Monteiro, and Donal O'Connell. The SAGEX Review on Scattering Amplitudes, Chapter 14: Classical Gravity from Scattering Amplitudes, October 2022. URL http://arxiv.org/abs/2203.13025. arXiv:2203.13025 [hepth].
- Clifford Cheung and Grant N. Remmen. Twofold Symmetries of the Pure Gravity Action. Journal of High Energy Physics, 2017(1):104, January 2017. ISSN 1029-8479. doi: 10.1007/JHEP01(2017)104. URL http://arxiv.org/abs/1612.03927. arXiv:1612.03927 [gr-qc, physics:hep-th].
- [10] John Joseph M. Carrasco. Gauge and Gravity Amplitude Relations. In Journeys Through the Precision Frontier: Amplitudes for Colliders, pages 477–557, November 2015. doi: 10.1142/9789814678766_0011. URL http://arxiv.org/abs/1506.00974. arXiv:1506.00974 [gr-qc, physics:hep-ph, physics:hep-th, physics:math-ph].
- Bryce S. DeWitt. Quantum Theory of Gravity. III. Applications of the Covariant Theory. *Physical Review*, 162(5):1239-1256, October 1967. ISSN 0031-899X. doi: 10. 1103/PhysRev.162.1239. URL https://link.aps.org/doi/10.1103/PhysRev.162. 1239.
- [12] Z. Bern, J. J. M. Carrasco, and H. Johansson. New Relations for Gauge-Theory Amplitudes. *Physical Review D*, 78(8):085011, October 2008. ISSN 1550-7998, 1550-2368. doi: 10.1103/PhysRevD.78.085011. URL http://arxiv.org/abs/0805.3993. arXiv:0805.3993 [gr-qc, physics:hep-ph, physics:hep-th].
- [13] Chris D. White. The double copy: gravity from gluons. Contemporary Physics, 59(2):109–125, April 2018. ISSN 0010-7514, 1366-5812. doi: 10.1080/00107514. 2017.1415725. URL http://arxiv.org/abs/1708.07056. arXiv:1708.07056 [gr-qc, physics:hep-ph, physics:hep-th, physics:physics].
- [14] Tim Adamo, John Joseph M. Carrasco, Mariana Carrillo-González, Marco Chiodaroli, Henriette Elvang, Henrik Johansson, Donal O'Connell, Radu Roiban, and Oliver Schlotterer. Snowmass White Paper: the Double Copy and its Applications, April 2022. URL http://arxiv.org/abs/2204.06547. arXiv:2204.06547 [gr-qc, physics:hep-ph, physics:hep-th, physics:math-ph].
- [15] Zvi Bern, John Joseph Carrasco, Marco Chiodaroli, Henrik Johansson, and Radu Roiban. The Duality Between Color and Kinematics and its Applications, September 2019. URL http://arxiv.org/abs/1909.01358. arXiv:1909.01358 [gr-qc, physics:hep-ph, physics:hep-th].
- [16] Chih-Hao Fu, Pierre Vanhove, and Yihong Wang. A Vertex Operator Algebra Construction of the Colour-Kinematics Dual numerator. *Journal of High Energy Physics*, 2018(9):141, September 2018. ISSN 1029-8479. doi: 10.1007/JHEP09(2018)141. URL http://arxiv.org/abs/1806.09584. arXiv:1806.09584 [hep-ph, physics:hep-th].
- [17] Andreas Brandhuber, Gang Chen, Henrik Johansson, Gabriele Travaglini, and Con-

gkao Wen. Kinematic Hopf Algebra for Bern-Carrasco-Johansson Numerators in Heavy-Mass Effective Field Theory and Yang-Mills Theory. *Physical Review Letters*, 128(12):121601, March 2022. ISSN 0031-9007, 1079-7114. doi: 10.1103/PhysRevLett. 128.121601. URL https://link.aps.org/doi/10.1103/PhysRevLett.128.121601.

- [18] Clifford Cheung and James Mangan. Covariant Color-Kinematics Duality. Journal of High Energy Physics, 2021(11):69, November 2021. ISSN 1029-8479. doi: 10.1007/ JHEP11(2021)069. URL http://arxiv.org/abs/2108.02276. arXiv:2108.02276 [hep-ph, physics:hep-th].
- [19] Yang Li, Diederik Roest, and Tonnis Ter Veldhuis. Hybrid Goldstone Modes from the Double Copy Bootstrap. *Physical Review Letters*, 132(21):211602, May 2024. ISSN 0031-9007, 1079-7114. doi: 10.1103/PhysRevLett.132.211602. URL https: //link.aps.org/doi/10.1103/PhysRevLett.132.211602.
- [20] Karol Kampf, Jiri Novotny, and Petr Vasko. Extended DBI and its generalizations from graded soft theorems. *Journal of High Energy Physics*, 2021(10):101, October 2021. ISSN 1029-8479. doi: 10.1007/JHEP10(2021)101. URL http://arxiv.org/ abs/2107.04587. arXiv:2107.04587 [hep-th].
- [21] H. Kawai, D. C. Lewellen, and S. H. H. Tye. A relation between tree amplitudes of closed and open strings. *Nuclear Physics B*, 269(1):1–23, May 1986. ISSN 0550-3213. doi: 10.1016/0550-3213(86)90362-7. URL https://www.sciencedirect.com/ science/article/pii/0550321386903627.
- [22] Clifford Cheung, Chia-Hsien Shen, and Congkao Wen. Unifying Relations for Scattering Amplitudes. Journal of High Energy Physics, 2018(2):95, February 2018. ISSN 1029-8479. doi: 10.1007/JHEP02(2018)095. URL http://arxiv.org/abs/1705.03025. arXiv:1705.03025 [hep-th].
- C. N. Yang and R. L. Mills. Conservation of Isotopic Spin and Isotopic Gauge Invariance. *Physical Review*, 96(1):191-195, October 1954. ISSN 0031-899X. doi: 10.1103/PhysRev.96.191. URL https://link.aps.org/doi/10.1103/PhysRev.96.191.
- [24] A. Zee. Quantum field theory in a nutshell. 2003. ISBN 978-0-691-14034-6.
- J. C. Ward. An Identity in Quantum Electrodynamics. *Physical Review*, 78(2): 182-182, April 1950. ISSN 0031-899X. doi: 10.1103/PhysRev.78.182. URL https://link.aps.org/doi/10.1103/PhysRev.78.182.
- [26] Vittorio Del Duca, Lance Dixon, and Fabio Maltoni. New Color Decompositions for Gauge Amplitudes at Tree and Loop Level. Nuclear Physics B, 571(1-2):51-70, April 2000. ISSN 05503213. doi: 10.1016/S0550-3213(99)00809-3. URL http://arxiv. org/abs/hep-ph/9910563. arXiv:hep-ph/9910563.
- [27] Wolfram Weise. Yukawa's Pion, Low-Energy QCD and Nuclear Chiral Dynamics. Progress of Theoretical Physics Supplement, 170:161–184, 2007. ISSN 0375-9687. doi:

10.1143/PTPS.170.161. URL http://arxiv.org/abs/0704.1992. arXiv:0704.1992 [nucl-th].

- [28] Ian Low. Adlers zero and effective Lagrangians for nonlinearly realized symmetry. *Physical Review D*, 91(10):105017, May 2015. ISSN 1550-7998, 1550-2368. doi: 10. 1103/PhysRevD.91.105017. URL https://link.aps.org/doi/10.1103/PhysRevD. 91.105017.
- [29] Jeremiah A. Cronin. Phenomenological Model of Strong and Weak Interactions in Chiral U (3) U (3). *Physical Review*, 161(5):1483-1494, September 1967. ISSN 0031-899X. doi: 10.1103/PhysRev.161.1483. URL https://link.aps.org/doi/10. 1103/PhysRev.161.1483.
- [30] Johan Bijnens, Karol Kampf, and Mattias Sjö. Higher-order tree-level amplitudes in the nonlinear sigma model. *Journal of High Energy Physics*, 2019(11):74, November 2019. ISSN 1029-8479. doi: 10.1007/JHEP11(2019)074. URL http://arxiv.org/ abs/1909.13684. arXiv:1909.13684 [hep-th].
- [31] Clifford Cheung, Karol Kampf, Jiri Novotny, Chia-Hsien Shen, and Jaroslav Trnka. A Periodic Table of Effective Field Theories. *Journal of High Energy Physics*, 2017 (2):20, February 2017. ISSN 1029-8479. doi: 10.1007/JHEP02(2017)020. URL http://arxiv.org/abs/1611.03137. arXiv:1611.03137 [gr-qc, physics:hep-ph, physics:hep-th].
- [32] Stephen L. Adler. Consistency Conditions on the Strong Interactions Implied by a Partially Conserved Axial-Vector Current. *Physical Review*, 137(4B):B1022-B1033, February 1965. ISSN 0031-899X. doi: 10.1103/PhysRev.137.B1022. URL https: //link.aps.org/doi/10.1103/PhysRev.137.B1022.
- [33] Laurentiu Rodina. Uniqueness from gauge invariance and the Adler zero. Journal of High Energy Physics, 2019(9):84, September 2019. ISSN 1029-8479. doi: 10.1007/ JHEP09(2019)084. URL http://arxiv.org/abs/1612.06342. arXiv:1612.06342
 [hep-th].
- [34] Leonard Susskind and Graham Frye. Algebraic Aspects of Pionic Duality Diagrams. *Physical Review D*, 1(6):1682–1686, March 1970. ISSN 0556-2821. doi: 10.1103/ PhysRevD.1.1682. URL https://link.aps.org/doi/10.1103/PhysRevD.1.1682.
- [35] Freddy Cachazo, Peter Cha, and Sebastian Mizera. Extensions of Theories from Soft Limits, June 2016. URL http://arxiv.org/abs/1604.03893. arXiv:1604.03893 [hep-th].
- [36] Henriette Elvang and Yu-tin Huang. Scattering Amplitudes in Gauge Theory and Gravity. Cambridge University Press, 1 edition, January 2015. ISBN 978-1-107-06925-1 978-1-107-70662-0. doi: 10.1017/CBO9781107706620. URL https://www. cambridge.org/core/product/identifier/9781107706620/type/book.

- [37] Ruth Britto, Freddy Cachazo, and Bo Feng. New Recursion Relations for Tree Amplitudes of Gluons. Nuclear Physics B, 715(1-2):499-522, May 2005. ISSN 05503213. doi: 10.1016/j.nuclphysb.2005.02.030. URL http://arxiv.org/abs/hep-th/0412308. arXiv:hep-th/0412308.
- [38] Ruth Britto, Freddy Cachazo, Bo Feng, and Edward Witten. Direct Proof Of Tree-Level Recursion Relation In Yang-Mills Theory. *Physical Review Letters*, 94(18): 181602, May 2005. ISSN 0031-9007, 1079-7114. doi: 10.1103/PhysRevLett.94.181602. URL http://arxiv.org/abs/hep-th/0501052. arXiv:hep-th/0501052.
- [39] Freddy Cachazo, Nick Early, and Bruno Umbert. Smoothly Splitting Amplitudes and Semi-Locality. Journal of High Energy Physics, 2022(8):252, August 2022. ISSN 1029-8479. doi: 10.1007/JHEP08(2022)252. URL http://arxiv.org/abs/2112.14191. arXiv:2112.14191 [hep-th].
- [40] Clifford Cheung. TASI Lectures on Scattering Amplitudes, August 2017. URL http: //arxiv.org/abs/1708.03872. arXiv:1708.03872 [hep-ph, physics:hep-th].
- [41] Steven Weinberg. Infrared Photons and Gravitons. *Physical Review*, 140(2B): B516-B524, October 1965. doi: 10.1103/PhysRev.140.B516. URL https://link. aps.org/doi/10.1103/PhysRev.140.B516. Publisher: American Physical Society.
- [42] Steven Weinberg. Photons and Gravitons in \$S\$-Matrix Theory: Derivation of Charge Conservation and Equality of Gravitational and Inertial Mass. *Physical Review*, 135 (4B):B1049-B1056, August 1964. doi: 10.1103/PhysRev.135.B1049. URL https://link.aps.org/doi/10.1103/PhysRev.135.B1049. Publisher: American Physical Society.
- [43] Eduardo Casali. Soft sub-leading divergences in Yang-Mills amplitudes. Journal of High Energy Physics, 2014(8):77, August 2014. ISSN 1029-8479. doi: 10.1007/ JHEP08(2014)077. URL http://arxiv.org/abs/1404.5551. arXiv:1404.5551 [hepth].
- [44] Burkhard U. W. Schwab and Anastasia Volovich. Subleading Soft Theorem in Arbitrary Dimensions from Scattering Equations. *Phys. Rev. Lett.*, 113(10):101601, 2014. doi: 10.1103/PhysRevLett.113.101601. __eprint: 1404.7749.
- [45] Freddy Cachazo and Andrew Strominger. Evidence for a New Soft Graviton Theorem, June 2014. URL http://arxiv.org/abs/1404.4091. arXiv:1404.4091 [hep-th].
- [46] LHCb collaboration. Observation of overlapping spin-1 and spin-3 \$\bar{D}^0 K^-\$ resonances at mass \$2.86 {\rm GeV}/c^2\$. *Physical Review Letters*, 113(16):162001, October 2014. ISSN 0031-9007, 1079-7114. doi: 10.1103/PhysRevLett.113.162001. URL http://arxiv.org/abs/1407.7574. arXiv:1407.7574 [hep-ex].
- [47] Clifford Cheung, Karol Kampf, Jiri Novotny, Chia-Hsien Shen, and Jaroslav Trnka. On-Shell Recursion Relations for Effective Field Theories. *Physical Review Letters*, 116(4):041601, January 2016. ISSN 0031-9007, 1079-7114. doi: 10.1103/PhysRevLett.

116.041601. URL http://arxiv.org/abs/1509.03309. arXiv:1509.03309 [hep-ph, physics:hep-th].

- [48] C. D. White. Exact solutions for the biadjoint scalar field. *Physics Letters B*, 763: 365-369, December 2016. ISSN 0370-2693. doi: 10.1016/j.physletb.2016.10.052. URL https://www.sciencedirect.com/science/article/pii/S0370269316306232.
- [49] Freddy Cachazo, Nick Early, and Yong Zhang. Color-Dressed Generalized Biadjoint Scalar Amplitudes: Local Planarity. Symmetry, Integrability and Geometry: Methods and Applications, February 2024. ISSN 18150659. doi: 10.3842/SIGMA.2024.016. URL http://arxiv.org/abs/2212.11243. arXiv:2212.11243 [hep-th].
- [50] Freddy Cachazo, Song He, and Ellis Ye Yuan. Scattering of Massless Particles: Scalars, Gluons and Gravitons. Journal of High Energy Physics, 2014(7):33, July 2014. ISSN 1029-8479. doi: 10.1007/JHEP07(2014)033. URL http://arxiv.org/abs/1309.0885. arXiv:1309.0885 [hep-th].
- [51] John Joseph M. Carrasco, Matthew Lewandowski, and Nicolas H. Pavao. Color-Dual Fates of F 3, R 3, and N = 4 Supergravity. *Physical Review Letters*, 131(5):051601, August 2023. ISSN 0031-9007, 1079-7114. doi: 10.1103/PhysRevLett.131.051601. URL https://link.aps.org/doi/10.1103/PhysRevLett.131.051601.
- [52] Johannes Broedel and Lance J. Dixon. Color-kinematics duality and double-copy construction for amplitudes from higher-dimension operators, October 2012. URL http://arxiv.org/abs/1208.0876. arXiv:1208.0876 [hep-th].
- [53] Huan-Hang Chi, Henriette Elvang, Aidan Herderschee, Callum R. T. Jones, and Shruti Paranjape. Generalizations of the Double-Copy: the KLT Bootstrap. *Journal* of High Energy Physics, 2022(3):77, March 2022. ISSN 1029-8479. doi: 10.1007/ JHEP03(2022)077. URL http://arxiv.org/abs/2106.12600. arXiv:2106.12600 [hep-ph, physics:hep-th].
- [54] Yu-tin Huang, Oliver Schlotterer, and Congkao Wen. Universality in string interactions. Journal of High Energy Physics, 2016(9):155, September 2016. ISSN 1029-8479. doi: 10.1007/JHEP09(2016)155. URL https://doi.org/10.1007/JHEP09(2016)155.
- [55] Thales Azevedo, Marco Chiodaroli, Henrik Johansson, and Oliver Schlotterer. Heterotic and bosonic string amplitudes via field theory. *Journal of High Energy Physics*, 2018(10):12, October 2018. ISSN 1029-8479. doi: 10.1007/JHEP10(2018)012. URL http://arxiv.org/abs/1803.05452. arXiv:1803.05452 [hep-th].
- [56] Henrik Johansson and Josh Nohle. Conformal Gravity from Gauge Theory, July 2017. URL http://arxiv.org/abs/1707.02965. arXiv:1707.02965 [gr-qc, physics:hep-th].
- [57] John Joseph M. Carrasco and Laurentiu Rodina. UV considerations on scattering amplitudes in a web of theories. *Physical Review D*, 100(12):125007, December 2019.

ISSN 2470-0010, 2470-0029. doi: 10.1103/PhysRevD.100.125007. URL https://link.aps.org/doi/10.1103/PhysRevD.100.125007.

- [58] Carlos R. Mafra, Oliver Schlotterer, and Stephan Stieberger. Complete N-Point Superstring Disk Amplitude II. Amplitude and Hypergeometric Function Structure, June 2011. URL http://arxiv.org/abs/1106.2646. arXiv:1106.2646 [hep-th].
- [59] John Joseph M. Carrasco, Carlos R. Mafra, and Oliver Schlotterer. Abelian Z-theory: NLSM amplitudes and alpha'-corrections from the open string. *Journal of High Energy Physics*, 2017(6):93, June 2017. ISSN 1029-8479. doi: 10.1007/JHEP06(2017)093. URL http://arxiv.org/abs/1608.02569. arXiv:1608.02569 [hep-th, physics:mathph].
- [60] Dijs de Neeling, Diederik Roest, and Sam Veldmeijer. Flavour-kinematics duality for Goldstone modes. Journal of High Energy Physics, 2022(10):66, October 2022. ISSN 1029-8479. doi: 10.1007/JHEP10(2022)066. URL http://arxiv.org/abs/2204.11629. arXiv:2204.11629 [hep-th].
- [61] Yang Li, Diederik Roest, and Tonnis ter Veldhuis. Hidden Zeros in Scaffolded General Relativity and Exceptional Field Theories, April 2024. URL http://arxiv.org/abs/ 2403.12939. arXiv:2403.12939 [hep-th].
- [62] Yang Li, Diederik Roest, and Tonnis ter Veldhuis. Hybrid Goldstone Modes from the Double Copy Bootstrap, July 2023. URL http://arxiv.org/abs/2307.13418. arXiv:2307.13418 [hep-th].
- [63] David A. Cox, John Little, and Donal OShea. Invariant Theory of Finite Groups. In David A. Cox, John Little, and Donal OShea, editors, *Ideals, Varieties, and Algorithms: An Introduction to Computational Algebraic Geometry and Commutative Algebra*, pages 345–383. Springer International Publishing, Cham, 2015. ISBN 978-3-319-16721-3. doi: 10.1007/978-3-319-16721-3_7. URL https://doi.org/10.1007/978-3-319-16721-3_7.
- [64] Richard P. Stanley. Invariants of finite groups and their applications to combinatorics. Bulletin of the American Mathematical Society, 1(3):475-511, 1979. ISSN 0273-0979, 1088-9485. doi: 10.1090/S0273-0979-1979-14597-X. URL https://www.ams.org/ bull/1979-01-03/S0273-0979-1979-14597-X/.
- [65] D. J. Benson. Polynomial Invariants of Finite Groups. London Mathematical Society Lecture Note Series. Cambridge University Press, Cambridge, 1993. ISBN 978-0-521-45886-3. doi: 10.1017/CBO9780511565809. URL https: //www.cambridge.org/core/books/polynomial-invariants-of-finite-groups/ 56EEC4701288E90F83F10DDCC2CE89E9.
- [66] M. (Morton). Hamermesh. Group theory and its application to physical problems. Dover books on physics and chemistry. Dover Publications, New York, 1989. ISBN 978-0-486-66181-0. Section: xv, 509 pages ; 22 cm.

- [67] Taro V. Brown, Karol Kampf, Umut Oktem, Shruti Paranjape, and Jaroslav Trnka. Scalar BCJ Bootstrap, May 2023. URL http://arxiv.org/abs/2305.05688. arXiv:2305.05688 [hep-th].
- [68] Glenn Barnich and Pierre-Henry Lambert. Einstein-Yang-Mills theory: I. Asymptotic symmetries. *Physical Review D*, 88(10):103006, November 2013. ISSN 1550-7998, 1550-2368. doi: 10.1103/PhysRevD.88.103006. URL http://arxiv.org/abs/1310.2698. arXiv:1310.2698 [gr-qc, physics:hep-th].
- [69] Ian Low and Zhewei Yin. The Infrared Structure of Nambu-Goldstone Bosons. Journal of High Energy Physics, 2018(10):78, October 2018. ISSN 1029-8479. doi: 10.1007/ JHEP10(2018)078. URL http://arxiv.org/abs/1804.08629. arXiv:1804.08629 [hep-ph, physics:hep-th].
- [70] Nima Arkani-Hamed and Carolina Figueiredo. Circles and Triangles, the NLSM and Tr(\$\Phi^3\$), March 2024. URL http://arxiv.org/abs/2403.04826. arXiv:2403.04826 [hep-ph, physics:hep-th].
- [71] Nima Arkani-Hamed, Qu Cao, Jin Dong, Carolina Figueiredo, and Song He. Hidden zeros for particle/string amplitudes and the unity of colored scalars, pions and gluons, December 2023. URL http://arxiv.org/abs/2312.16282. arXiv:2312.16282 [hepph, physics:hep-th].
- [72] Nima Arkani-Hamed, Qu Cao, Jin Dong, Carolina Figueiredo, and Song He. NLSM \$\subset\$ Tr\$(\phi^3)\$, April 2024. URL http://arxiv.org/abs/2401.05483. arXiv:2401.05483 [hep-ph, physics:hep-th].
- [73] John Joseph M. Carrasco, Laurentiu Rodina, Zanpeng Yin, and Suna Zekioglu. Simple encoding of higher derivative gauge and gravity counterterms. *Physical Review Letters*, 125(25):251602, December 2020. ISSN 0031-9007, 1079-7114. doi: 10.1103/PhysRevLett.125.251602. URL http://arxiv.org/abs/1910.12850. arXiv:1910.12850 [hep-ph, physics:hep-th].
- [74] Yoann Sohnle and Oliver Schlotterer. String corrections to the double copy of Einstein Yang-Mills.
- [75] Kang Zhou and Chang Hu. Towards tree Yang-Mills and Yang-Mills-scalar amplitudes with higher-derivative interactions, June 2024. URL http://arxiv.org/abs/2406.
 03034. arXiv:2406.03034 [hep-th].
- [76] Quentin Bonnefoy, Gauthier Durieux, and Jasper Roosmale Nepveu. Higher-derivative relations between scalars and gluons, October 2023. URL http://arxiv.org/abs/ 2310.13041. arXiv:2310.13041 [hep-ph, physics:hep-th].
- [77] Song He, Linghui Hou, Jintian Tian, and Yong Zhang. Kinematic numerators from the worldsheet: cubic trees from labelled trees. *Journal of High Energy Physics*, 2021(8):118, August 2021. ISSN 1029-8479. doi: 10.1007/JHEP08(2021)118. URL http://arxiv.org/abs/2103.15810. arXiv:2103.15810 [hep-th].

- [78] Xiangrui Gao, Song He, and Yong Zhang. Labelled tree graphs, Feynman diagrams and disk integrals. *Journal of High Energy Physics*, 2017(11):144, November 2017. ISSN 1029-8479. doi: 10.1007/JHEP11(2017)144. URL http://arxiv.org/abs/1708.08701. arXiv:1708.08701 [hep-th].
- [79] Freddy Cachazo, Song He, and Ellis Ye Yuan. Scattering Equations and Matrices: From Einstein To Yang-Mills, DBI and NLSM. *Journal of High Energy Physics*, 2015(7):149, July 2015. ISSN 1029-8479. doi: 10.1007/JHEP07(2015)149. URL http://arxiv.org/abs/1412.3479. arXiv:1412.3479 [hep-th].
- [80] Marco Chiodaroli, Murat Gunaydin, Henrik Johansson, and Radu Roiban. Explicit Formulae for Yang-Mills-Einstein Amplitudes from the Double Copy. *Journal of High Energy Physics*, 2017(7):2, July 2017. ISSN 1029-8479. doi: 10.1007/JHEP07(2017) 002. URL http://arxiv.org/abs/1703.00421. arXiv:1703.00421 [hep-th].
- [81] Kang Zhou and Bo Feng. Note on differential operators, CHY integrands, and unifying relations for amplitudes. *Journal of High Energy Physics*, 2018(9):160, September 2018. ISSN 1029-8479. doi: 10.1007/JHEP09(2018)160. URL http://arxiv.org/ abs/1808.06835. arXiv:1808.06835 [hep-th].
- [82] Henriette Elvang and Yu-tin Huang. Scattering Amplitudes, April 2014. URL http: //arxiv.org/abs/1308.1697. arXiv:1308.1697 [hep-ph, physics:hep-th].
- [83] Fang-Stars Wei and Kang Zhou. Tree and \$1\$-loop fundamental BCJ relations from soft theorems. The European Physical Journal C, 83(7):616, July 2023. ISSN 1434-6052. doi: 10.1140/epjc/s10052-023-11698-2. URL http://arxiv.org/abs/2305.04620. arXiv:2305.04620 [hep-th].
- [84] Zvi Bern. Perturbative Quantum Gravity and its Relation to Gauge Theory. Living Reviews in Relativity, 5(1):5, December 2002. ISSN 2367-3613, 1433-8351. doi: 10.12942/lrr-2002-5. URL http://arxiv.org/abs/gr-qc/0206071. arXiv:grqc/0206071.
- [85] Johannes Broedel, Oliver Schlotterer, and Stephan Stieberger. Polylogarithms, Multiple Zeta Values and Superstring Amplitudes. *Fortschritte der Physik*, 61(9):812–870, September 2013. ISSN 0015-8208, 1521-3978. doi: 10.1002/prop.201300019. URL http://arxiv.org/abs/1304.7267. arXiv:1304.7267 [hep-ph, physics:hep-th].
- [86] Clifford Cheung, Grant N. Remmen, Chia-Hsien Shen, and Congkao Wen. Pions as Gluons in Higher Dimensions. *Journal of High Energy Physics*, 2018(4):129, April 2018. ISSN 1029-8479. doi: 10.1007/JHEP04(2018)129. URL http://arxiv.org/ abs/1709.04932. arXiv:1709.04932 [gr-qc, physics:hep-ph, physics:hep-th].
- [87] Jin Dong, Xiang Li, and Fan Zhu. Pions from higher-dimensional gluons: general realizations and stringy models, April 2024. URL http://arxiv.org/abs/2404.
 11648. arXiv:2404.11648 [hep-th].

- [88] Clifford Cheung and Chia-Hsien Shen. Symmetry for Flavor-Kinematics Duality from an Action. *Physical Review Letters*, 118(12):121601, March 2017. ISSN 0031-9007, 1079-7114. doi: 10.1103/PhysRevLett.118.121601. URL https://link.aps.org/ doi/10.1103/PhysRevLett.118.121601.
- [89] Nima Arkani-Hamed, Qu Cao, Jin Dong, Carolina Figueiredo, and Song He. Scalar-Scaffolded Gluons and the Combinatorial Origins of Yang-Mills Theory, December 2023. URL http://arxiv.org/abs/2401.00041. arXiv:2401.00041 [hep-ph, physics:hep-th].
Appendix A

Diagrammatic calculation of double color ordered BAS amplitudes

A convenient method proposed by Cachazo, He and Yuan [82] to calculate the double color use diagrams with the partial orderings from which we can read off the propagators s_{ij} that make up the partial amplitudes. The 5-point partial amplitude $A_{BAS}(12345|14235)$ serves as a good example to demonstrate this method [83].

We construct a disk with the first ordering (12345) depicted counter-clockwise on it's boundary. Next we draw aloop of lines from point to point following the second ordering (14235), as one would with a 'connect the dots' drawing. The result is depicted in figure Figure A.1.



Figure A.1: Diagrams for $A_5(1, 2, 3, 4, 5|1, 4, 2, 3, 5)$ and $A_5(1, 2, 3, 4, 5|1, 2, 4, 3, 5)$

This diagram is the only such diagram that can contribute to this partial amplitudes.

Different diagrams, such as figure Figure A.1 correspond to different partial amplitudes. To calculate the amplitude, we consider the borders that are shared by both the outer disk and the internal lines. For figure Figure A.1b the borders $\{1,5\}$ and $\{2,3\}$ indicate the exchange channels s_{15} and s_{23} . These shared borders are also present in the ordering of A_{BAS} . The partial amplitude is then the product of these two channels

$$A_{BAS}(12345|14235) = \frac{1}{s_{23}} \frac{1}{s_{51}},\tag{A.1}$$

up to an overall sign. Similarly, for the diagram in Figure A.1b, the amplitude $A_{BAS}(12345|12435)$ shares the boundaries {3,4} and {5,1,2}. These indicate the channel $1/s_{34}$ and $1/s_{512} = 1/(s_{12} + s_{51})$, which gives us (up to an overall sign)

$$A_{BAS}(12345|12435) = \frac{1}{s_{34}} \left(\frac{1}{s_{12}} + \frac{1}{s_{51}}\right).$$
(A.2)

In the diagrammatic approach, the sign is determined through the following rules

- For polygons with an odd number of vertices, a positive sign is assigned when their orientation aligns with that of the disk, and a negative sign when it does not.
- Polygons with an even number of vertices consistently receive a negative sign.
- Every intersection point results in the addition of a negative sign.

It should be noted that different assignment of signs can be used, as long as we stay consistent in applying the same rules to all amplitudes. This way, the relative signs are conserved. The resulting double-ordered amplitudes are then given by

$$A_{BAS}(12345|12435) = -\frac{1}{s_{23}}\frac{1}{s_{51}},\tag{A.3}$$

$$A_{BAS}(12345|14235) = -\frac{1}{s_{34}} \left(\frac{1}{s_{12}} + \frac{1}{s_{51}}\right).$$
(A.4)

Appendix B

KLT double copy

We have extensively discussed the BCJ double copy, but there is another formulation of the double copy, namely the Kawai-Lewellen-Tye (KLT) formulae [21]. Similar to the BCJ formulation, it allows us to construct tree-level gravity amplitudes \mathcal{M}^{tree} by taking the (KLT) product of purely-adjoint gauge theory amplitudes. For 3- and 4-point they explicitly give the graviton amplitudes in the form of

$$\mathcal{M}_{3}^{tree}(1,2,3) = iA_{3}^{tree}(1,2,3)\tilde{A}_{3}^{tree}(1,2,3), \tag{B.1}$$

$$\mathcal{M}_{4}^{tree}(1,2,3,4) = -is_{12}A_{4}^{tree}(1,2,3,4)\tilde{A}_{4}^{tree}(1,2,3,4), \tag{B.2}$$

where the amplitudes A^{tree} and \tilde{A}^{tree} are the colour-ordered partial amplitudes discussed in subsection 1.1.4. Through the double copy, there exists a mapping between the on-shell gluon polarisation vectors e^i_{μ} and the polarisations of the fields corresponding to the resuting amplitude $e^i_{\mu}e^j_{\nu}$. Namely, these can be decomposed into three combinations representing a graviton polarisation $e^h_{\mu\nu}$, a B-field polarisation $e^B_{\mu\nu}$ and the dilaton polarisation $e^{\phi}_{\mu\nu}$. Respectively these are the symmetric-traceless, the antisymmetric and the scalar mode. An action that describes these three fields is

$$S = \int d^{D}x \sqrt{-g} \left[-\frac{1}{2}R + \frac{1}{2(D-2)} \partial^{\mu}\phi \partial_{\mu}\phi + \frac{1}{6} e^{-4\phi/(D-2)} H^{\lambda\mu\nu} H_{\lambda\mu\nu} \right],$$
(B.3)

where $H_{\lambda\mu\nu}$ is field strength of the two-index antisymmetri tensor $B_{\mu\nu}$ that describes the B-field. This action can be truncated to Einstein gravity by requiring \mathbb{Z}_2 symmetries. If we wish to describe graviton amplitudes as a product of gluon amplitudes through the KLT product instead of from the action, we will have to pick the gluon polarisations such that they are in the symmetric-traceless combination.

At *m*-point, the KLT product between gauge theory amplitudes is given by

$$\mathcal{M}_{m}^{tree} = -i \sum_{\sigma, \rho \in S_{m-3}} A_{m}^{tree}(1, \sigma, m-1, m) S[\sigma|\rho] \tilde{A}_{m}^{tree}(1, \rho, m, m-1).$$
(B.4)

The constituent tree-level amplitudes are colour-ordered with leg 1, leg m and leg (m-1) fixed. This fixing then leaves (m-3)! permutations of external legs (specified by σ and ρ) that are summed over. The permutation also specify the elements of the 'KLT kernel' $S[\sigma|\rho]$ that are the amplitudes are multiplied with. These elements are polynomials of products of p_i 's, which produce the factors of s_{ij} seen in (B.1).

The KLT kernel solves several problems that would occur in a naive product of gauge theory amplitudes. Firstly it cancels the double poles that occur, which would make for an unphysical amplitude. Secondly it provides the 'missing' poles that have to be included due to the gravity amplitude not being colour-ordered. Consider the 4 point example. The colour-stripped gluon tree amplitude $A_4[1234]$ has simple poles in s_{12} and s_{14} , but none in s_{13} as these legs are not adjecent in the colour-ordering. Naively squaring results in $A_4[1234]^2$ which has double poles in $s_{12} = 0$ and $s_{14} = 0$. However, the 4-graviton tree amplitude $M_4(1234)$ has a pole in all three channels s_{12} , s_{13} and s_{14} (the s, t, and u channel). The double copy can be written as the KLT copy in the following way:

$$M_4(1234) = A_4[1234]S_4[1234]A_4[1234].$$
(B.5)

The KLT kernel $S_4[1234|1234]$ can be calculated to be

$$S_4[1234|1234] = -\frac{s_{12}s_{14}}{s_{13}},\tag{B.6}$$

which provides exactly the cancellation of the double poles in s_{12} and s_{14} and introduces the new pole in s_{13} .

Originally the KLT relations were derived in the field of string theory as relations between colour-stripped disk amplitudes of open-strings. In these relations, there is an explicit dependence on the string tension α' . In the low energy limit, where α' goes to zero, the string-theory KLT kernel reduces to the field theory KLT kernel described. Interestingly the string tension α' can be used to formulate a pertubative expansion of higher-order corrections to the field theories. In the following sections we will explore constructing such higher-order order corrections by starting from field theory amplitudes instead of string theory amplitudes.

Appendix C

Z-Theory

It was found that the amplitudes of the open-superstring can be written as the double copy of color-ordered YM amplitudes together with certain functions F^{σ} [58, 84]

$$A(1,...,N) = \sum_{\sigma \in S_{N-3}} A_{YM}(1, 2_{\sigma}, ..., (N-2)_{\sigma}, N-1, N) F^{\sigma}_{(1,...,N)}(\alpha')$$
(C.1)

The F functions are later re-named as Z functions Z_P that give rise to the name Z-theory. The function is an integral over the boundary of a disk worldsheet¹ with two orderings Qand P. The first denotes the ordering in the integrand and is responsible for dictating which color-ordering of subamplitude is being computed. The latter the defines the ordering of the integration domain. Explicitly the functions are denoted as

$$Z_P(q_1, \dots, q_n) \equiv (\alpha')^{n-3} \int_{D(P)} \frac{dz_1 dz_2 \cdots dz_n}{\operatorname{vol}(SL(2, \mathbb{R}))} \frac{\prod_{i < j}^n |z_{ij}|^{\alpha' p_i p_j}}{z_{i_1 i_2} z_{i_2 i_3} \cdots z_{i_n i_2}},$$
 (C.2)

where $z_{ij} = z_i - z_j$ give rise to the propagators as seen in the denominator. The domain $D = \{-\infty < z_1 < z_2 < \cdots < z_n < +\infty\}$. The functions behave like scalar amplitudes, and exhibit a double ordering. The Z-function obey KK-relations similar to (1.21) and BCJ relation similar to (1.26) along the ordering Q. Additionally they obey the string-theory-monodromy relations along integration domain P

$$0 = \sum_{j=2}^{n-1} \exp\left[i\pi\alpha'(k_{p_1} \cdot k_{p_2p_3\dots p_j})\right] Z_{p_2p_3\dots p_jp_1p_{j+1}\dots p_n}(Q),$$
(C.3)

which leaves a basis of (n-3)! integration domains.

 $^{^{1}}$ In string theory, a worldsheet is the two-dimensional surface traced out by a string as it propagates through spacetime. This concept generalizes the idea of a particle's worldline to higher dimensions, providing a framework for analyzing the dynamics and interactions of strings.

After applying all these relations we are left with the *Q*-ordered functions that simply obey field-theory relations. [59] conjectures that the resulting functions are the color-ordered scattering amplitude for *some* EFT.

The fact that Z-functions obey the Kleiss-Kuijf (KK) and Bern-Carrasco-Johansson (BCJ) relations suggests they correspond to scattering amplitudes for a conjectural, yet unidentified, doubly-colored theory, hereby referred to as Z-theory. Consequently, the Z-functions will be referred to as Z-amplitudes.

It is also shown [85] that the $\alpha' \to 0$ limit of Z-functions aligns with the inverse of the KLT matrix

$$\lim_{\alpha \to 0} Z_P(Q) = m[P|Q] \tag{C.4}$$

This is later demonstrated to correspond to the (doubly-partial) tree-level amplitudes of Born-Infeld-Skyrme (BAS) theory.

$$M_{\phi^3} = \sum_{\sigma,\rho \in S_{n-1}} \operatorname{Tr} \left(T_1 T_{\sigma(2)} \dots T_{\sigma(n)} \right) \operatorname{Tr} \left(\tilde{T}_1 \tilde{T}_{\rho(2)} \dots \tilde{T}_{\rho(n)} \right) m[1, \sigma(2, \dots, n) | 1, \rho(2, \dots, n)]$$
(C.5)

It is intriguing that even though string theory does not include a BAS, the tree-level amplitudes of this theory are nevertheless contained within string tree-level amplitudes.

Z-amplitudes serve as double-copy factors for open-superstring scattering amplitudes, containing all orders in α' and obeying field-theory scattering amplitude relations. The lowenergy α' expansion should be identifiable as the scattering amplitudes of known theories.

NLSM from Abelian Z-theory

It was then shown that it is possible to replace Yang-Mills factors of Abelian open-string amplitudes by gauge-theory colour factors. This replacement gives rise to Abelian Zamplitudes. In the field theory limit the leading order α' contributions then give rise to NLSM amplitudes

$$A_{NLSM}(1,2,\ldots,n) = \lim_{\alpha' \to 0} (\alpha')^{2-n} \sum_{\sigma \in S_{n-1}} Z_{1\sigma(2,3,\ldots,n)}(1,2,\ldots,n).$$
(C.6)

Unlike the amplitudes in BAS ar theories, NLSM amplitudes come from specific nonzero orders $(\alpha')^{n-2}$. These amplitudes then denote the color-ordered NLSM amplitudes. Explicitly at n = 4 and n = 6 for example they give rise to the correct NLSM amplitudes.

$$A_{NLSM}(1,2,3,4) = \pi^2(s_{12} + s_{23}) \tag{C.7}$$

$$A_{NLSM}(1,2,\ldots,6) = \pi^2 \left[s_{12} - \frac{1}{2} \frac{(s_{12} + s_{23})(s_{45} + s_{56})}{s_{123}} + cyclic(1,2,3,4,5,6) \right]$$
(C.8)

Besides the leading order surviving contributions in α' there are also the subleading terms. These correspond to the 'stringy' higher-derivative corrections to NLSM amplitudes.

Appendix D

Transmutation operators

D.1 Kinematics

The initial amplitude A which we will be transforming is defined with the on-shell conditions $p_i p_i = p_i e_i = e_i e_i 0$ and conserves momentum $\sum_i p_i = 0$. The on-shell conditions are already preserved by defining the amplitude to strictly be a function of the kinematic data $e_i e_j$, $p_i e_j$ and $p_i p_j$. The momentum conservation can be enforced by defining a *Total* momentum operator \mathcal{P}_v :

$$P_v \equiv \sum_i p_i v = (p_1 + p_2 + \dots + p_n)v_i,$$
 (D.1)

where the sum runs over all external legs i and v denotes any momentum or polarisation vector. The inclusion of the somewhat arbitrary vector v is necessary for the operator to be Lorentz invariant. This total momentum operator annihilates the amplitude due to the presence of a momentum-conserving delta-function $\delta^{(D)}(p_1 + \ldots + p_n)$. Momentum conservation of the transmuted amplitude then has to satisfy:

$$P_v \cdot (\mathcal{T} \cdot A) = 0 = \mathcal{T} \cdot (P_v \cdot A) \tag{D.2}$$

The requirement for momentum conservation can then be cast into the following requirement:

$$[P_v, \mathcal{T}] = 0. \tag{D.3}$$

This is not satisfied for every possible differential operator which contains derivatives with respect to $p_i p_j$ and $p_i e_j$, therefore the condition constraints the spectrum of possible operators. The total momentum operator satisfies the condition $[P_v, P_w] = 0$.

D.2 Gauge invariance

The condition of gauge invariance is translated into a differential operator through the Ward identity. Recall that for a given scattering amplitude with an external gauge boson of momentum k, the amplitude can be written as $A(k) = e_{\mu}(k)A^{\mu}$. The ward identity then gives rise to a vanishing amplitude when replacing the polarisation vector e_{μ} by the corresponding momentum vector: $k_{\mu}A^{\mu}(k) = 0$. The differential Ward operator can therefore be defined as:

$$W_i \equiv \sum_{v} p_i v \partial_{v e_i}.$$
 (D.4)

The summation over v runs over all external momentum and polarisation vectors in the amplitude. The polarisation of leg i appears in the amplitude in the Lorentz invariants $e_i p_j$ and $e_i e_j$. What this operator essentially does is replace every $e_i \rightarrow p_i$ wherever it shows up in the amplitude, turning $e_i p_j \rightarrow p_i p_j$ and $e_i e_j \rightarrow p_i e_j$. To demonstrate this operator we explicitly denote the Ward operator for a three point amplitude:

$$W_1 = p_1(p_1\partial_{p_1e_1} + p_2\partial_{p_2e_1} + p_3\partial_{p_3e_1} + e_1\partial_{e_1e_1} + e_2\partial_{e_2e_1} + e_3\partial_{e_3e_1})$$
(D.5)

Any gauge invariant amplitude then satisfies the Ward identity and is therefore annihilated when the Ward operator acts on it:

$$W_i \cdot A = 0. \tag{D.6}$$

Requiring that gauge invariance be preserved for a transmuted amplitude then implies a vanishing result when the commutator $[\mathcal{T}, W_i]$ acts on an amplitude:

$$W_i(\mathcal{T} \cdot A) = \mathcal{T}(W_i \cdot A) = 0 \tag{D.7}$$

In the case of W_i , we can check that the operator itself preserves the gauge invariance of the amplitude as seen by calculating

$$[W_i, W_j] = \sum_{v, w} [p_i v \partial_{we_i}, p_j w \partial_{we_j}] = 0.$$
(D.8)

Moreover, it can be shown to satisfy our previous requirement of momentum conservation:

$$[W_i, P_v] \cdot A = \sum_{j,w} [p_i w \partial_{we_i}, p_j v] \cdot A = \delta_{ve_i} P_{p_i} \cdot A = 0.$$
(D.9)

D.2.1 Basis for Transmutation Operators

Starting from the most general form that a transmutation can take we can apply our condition for momentum conservation and gauge invariance in order to constrain the general form to form basic elements for the transmutation operators. The operators take the form of first order differential equations which act on the objects $e_i e_j$, $e_i p_j$ and $p_i p_j$. The general form can be written as:

$$\mathcal{T} \equiv \sum_{i,j} A_{ij} \partial_{p_i p_j} + B_{ij} \partial_{p_i e_j} + C_{ij} \partial_{e_i e_j}$$
(D.10)

Here, A_{ij} , B_{ij} and C_{ij} are general functions of external kinematic data and for later notation it is useful to choose $A_{ii} = B_{ii} = C_{ii} = 0$. We then impose the constraints of momentum conservation and gauge invariants upon this ansatz. First, momentum conservation implies

$$[T, P_v] = \sum_{i,j,k} [A_{ij}\partial_{p_i p_j} + B_{ij}\partial_{p_i e_j}, p_k v] = 0, \qquad (D.11)$$

where we can choose either v = p or v = e respectively to optain:

$$\sum_{i} A_{ij} + A_{ji} = \sum_{i} B_{ij} = 0 \tag{D.12}$$

This in turn implies that the rows and colums of A_{ij} sum to zero, the columns of B_{ij} sum to zero and C_{ij} remains unconstrained. In terms of transmutation operators this leaves a set of commuting operators which will form the basic building blocks for more complicated operators. The operators are denoted as

$$T_{ij} \equiv \partial_{e_i e_j} \tag{D.13}$$

$$I_{ijk} \equiv \partial_{p_i e_j} - \partial_{p_k e_j} \tag{D.14}$$

$$T_{ijkl} \equiv \partial_{p_i p_j} - \partial_{p_k p_j} + \partial_{p_k p_l} - \partial_{p_i p_l} \tag{D.15}$$

These operators have the following symmetry properties:

$$T_{ij} = T_{ji} \tag{D.16}$$

$$I_{ijk} = -I_{kji} \tag{D.17}$$

$$T_{ijkl} = -T_{kjil} = T_{klij} = -T_{ilkj} \tag{D.18}$$

Next we turn our attention to the constraint of gauge invariance. We can see that T_{ij} is intrinsically gauge inviariant, as $[T_{ij}, W_k] = \sum_v [\partial_{e_i e_j}, p_i v \partial_{v e_i}] = 0.$

Furthermore, we can see that I_{ijk} is not instrinsically gauge invariant. However, this does not pose a problem for as it can be shown to be *effectively* gauge invariant. The commutator is $[I_{ijk}, W_l] = \delta_{il}T_{ij} - \delta_{kl}T_{jk}$, which can be combined with supplemental operators allowing the vanishing of the amplitude.

The third operator strips of pairs of polarisation vectors and will therefore give rise to an object with double poles in momenta. The object generated after transmutation is therefore not a physical scattering amplitude. Regardless, this operator does appear in the calculation of the subleading soft factor of Born-Infeld theory.

In the next sections, we will see how these operators are the building blocks to connecting the amplitudes of various theories in a unified web.

D.3 Dimensional Reduction: Pions from higher dimensional gluons

The processes of the trace operator $T[\alpha]$ and the Longitudinal operator $T\mathcal{L}$ also have a more physical interpretation, namely a dimensional reduction [86]. Here we will discuss the variation of a dimensional reduction that is identical to the longitudinal operator.

In this mechanism, massless gluons are constructed to live in a (2d + 1)-dimensional space which will be reduced to a *d*-dimensional subspace. This construction implies that the momentum vectors \mathcal{P}_i^M are (2d + 1)-dimensional and are denoted with capital letters as Lorentz indices:

$$\mathcal{P}_{i}^{M} = (p_{i}^{\mu}, 0, 0) \tag{D.19}$$

The first and third entry of this vector are both d-dimensional and have Greek indices, while the middle entry is 1 dimensional. The external polarizations can also be seen to live in a (2d + 1)-dimensional space. We will pick the external polarizations to be

$$\mathcal{E}_{1}^{M} = \mathcal{E}_{n}^{M} = (0, 1, 0) \qquad \qquad \mathcal{E}_{i}^{M} = (p_{i}^{\mu}, 0, ip_{i}^{\mu}) \qquad \qquad i \neq 1, n.$$
(D.20)

Here we have set leg 1 and n to be polarized in their own 1-dimensional space while the other legs live in the d-dimensional subspaces. The reasoning for this specific choice of polarizations apparant as we compute the Lorentz invariant kinematic products of these vectors. The products of higher dimensional momenta gives rise to the product of d-dimensional momenta

$$\mathcal{P}_i \mathcal{P}_j = p_i p_j \quad . \tag{D.21}$$

Furthermore, the product of \mathcal{E}_1^M and \mathcal{E}_n^M will return 1 (in the 1-dimensional subspace).

$$\mathcal{E}_1 \mathcal{E}_n = 1. \tag{D.22}$$

Moreover, by construction these two polarizations are orthogonal to the polarizations of the other external legs as seen in (D.3). The other internal polarization products vanish:

$$\mathcal{E}_i \mathcal{E}_j = 0. \tag{D.23}$$

Finally the product of momenta and polarizations in (2d+1)-dimensions returns the product of d-dimensional momenta for $j \neq 1, n$

$$\mathcal{P}_i \mathcal{E}_j = p_i p_j \quad , \tag{D.24}$$

while the products $\mathcal{P}_{\lambda}\mathcal{E}_1$ and $\mathcal{P}_i\mathcal{E}_n$ vanish. This choice of external kinematics works identically to a differential operator which, at the level of the amplitude, acts as a longitudinal operator that replaces products of $e_i p_j$ by $p_i p_j$ and sets the pair of polarizations $e_1 e_n \to 1$:

$$\overline{T}_{1n}\overline{\mathcal{L}} = \frac{\partial}{\partial(e_1e_n)} \prod_{i=2}^{n-1} \left(\sum_{j \neq i} p_i p \frac{\partial}{\partial(e_ip_j)} \right)$$
(D.25)

This version of the longitudinal operator is formulated slightly different from the $T\mathcal{L}$ operator in (3.43). However, as shown in [87] the two dimensional reductions

Dim Red 1
$$e_a \cdot p_b \to 0$$
, $e_a \cdot e_b \to -p_a \cdot p_b$, $e_a, e_b \notin \{e_1, e_n\}$ (D.26)

Dim. Red. 2
$$e_a \cdot e_b \to 0, \qquad e_a \cdot p_b \to p_a \cdot p_b$$
 (D.27)

yield the same results at the level of the amplitude.

This higher dimensional interpretation has also been shown at the level of the action. The Lagrangian for YM in 2d + 1 dimension with Feynman gauge is simplified to be

$$\mathcal{L}_{\rm YM} = Tr\left(-\frac{1}{2}\partial_M \mathcal{A}_N \partial^M \mathcal{A}^N + i\sqrt{2}\partial_M \mathcal{A}_N \left[\mathcal{A}^M, \mathcal{A}^N\right] + \frac{1}{2}[\mathcal{A}_M, \mathcal{A}_N] \left[\mathcal{A}^M, \mathcal{A}^N\right]\right) \quad (D.28)$$

The gluon fields transform under the adjoint representation $\mathcal{A}_M = \mathcal{A}_M^a T^a$. The field can be split into three component field

$$\mathcal{A}_M = \mathcal{X}_M + \mathcal{Y}_M + \mathcal{Z}_M, \tag{D.29}$$

where these (2d + 1)-dimensional component fields are split into a (d, 1, d)-dimensional component fields through the parametrisation

$$\mathcal{X}_{M} = \frac{1}{\sqrt{2}} (X_{\mu}, 0, -iX_{\mu})$$

$$\mathcal{Y}_{M} = (0, Y, 0)$$

$$\mathcal{Z}_{M} = \frac{1}{\sqrt{2}} (X_{\mu}, 0, +iX_{\mu})$$

(D.30)

Here X_{μ} and Z_{μ} are *d*-dimensional vector fields and Y is a scalar field. In this parametrization the following inner product relations are established

$$\mathcal{A}_{M}\mathcal{A}^{M} = X_{\mu}Z^{\mu} + Z_{\mu}X^{\mu} + Y^{2}$$
$$\mathcal{X}_{M}\mathcal{X}^{M} = \mathcal{Z}_{M}\mathcal{Z}^{M} = 0$$
$$\mathcal{X}_{M}\mathcal{Y}^{M} = \mathcal{Z}_{M}\mathcal{Y}^{M} = 0$$
(D.31)

It can then be shown through weight counting of the interactions in the Lagrangian and the tranverse properties of the fields that the Lagrangian in (D.28) truncates to a Lagrangian that describes the NGSb of the NLSM through as the fields X_{μ} and Y_{μ}

$$\mathcal{L}_{\text{NLSM}} = Tr\left(X_{\mu}\Box Z^{\mu} + \frac{1}{2}Y\Box Y + i\left(X_{\mu\nu}\left[Z^{\mu}, Z^{\nu}\right] + Z^{\mu}[Y, \partial_{\mu}Y]\right)\right)$$
(D.32)

Where the field strength tensor for X_{μ} is defined as $X_{\mu\nu} = \partial_{\mu}X_{\nu} - \partial_{\nu}X_{\mu}$. This gives rise to pion amplitudes

$$A(\pi_1, \pi_2, \dots, \pi_{n-1}, \pi_n) = A(Y_1, Z_2, \dots, Z_{n-1}, Y_n),$$
(D.33)

in agreement with the findings of [88]. [86] Further shows that a similar dimensional reduction can be applied to gravitons to derive the action and amplitudes of Born-Infeld and Special Galileon theory. Last but not least the higher-dimensional perspective the double copy is manifest at the level of the action. This can be done by dropping flavor indices of the fields and doubling the kinematic structures, e.g. $X_{\mu\bar{\mu}}$.

The connection between higher-dimensional gluons and scalars was also demonstrated in [89], where it was shown that the scattering amplitudes of gluons are intricately embedded within 'stringy' deformations of pure scalar amplitudes. This paper introduces a new approach to formulating Yang-Mills scattering amplitudes in any number of dimensions and loop orders, using combinatorial and geometric ideas in kinematic space. The authors show that by using a "scalar scaffolding" method (Figure D.1), where every pair of coloured scalars produces a gluon, the shifted amplitudes of $Tr[\phi^3]$ theory match those of gluons. This scaffolded method preserves essential properties of gluon amplitudes such as multilinearity, gauge invariance, and factorization, effectively connecting scalar and gluon amplitudes through a shift in kinematic variables. It was shown in [72] that deformation of the "stringy" $Tr[\phi^3]$ amplitudes secretly contain pion amplitudes, extending the connection between scalars, gluons and pions.



Figure D.1: A diagram depicting the 3-point gluon amplitude that is scaffolded from a 6-point scalar amplitude.

D.4 Gauge invariance of longitudinal operator

Let us consider the conditions of conservation of the on-shell kinematics and the gauge invariance of the longitudinal operators. For starters, these operators can be written as linear combinations of these operators:

$$L_i = \sum_{j \neq k} p_i p_j I_{jik} \qquad \text{and} \qquad L_{ij} = -p_i p_j T_{ij}. \qquad (D.34)$$

The (effective) vanishing of the commutation with the total momentum operator P then guarantee the conservation of the on-shell kinematics. Next, while L_{ij} is intrinsically gauge invariant due to it being a linear combination of the gauge invariant trace operator T_{ij} , this is not the case for L_i . For this operator we can argue again that it can be shown to be effectively gauge invariant if the commutator $[L_i, W_j] = -L_{ij}$ annihilates the amplitude. This can be shown by rewriting the Ward identity operator as

$$W_i = L_i - \Delta_i,$$
 where $\Delta_i = \sum_j p_i e_j \partial_{e_j e_i}.$ (D.35)

With this we can rewrite a product $L_i \cdot L_j$ by inserting $L_i = W_i + \Delta_i$ and commute this operator to the right:

$$L_i \cdot L_j = (W_i + \Delta_i)L_j \tag{D.36}$$

$$= [W_i, L_j] + L_j \cdot W_i + \Delta_i L_j \tag{D.37}$$

$$=L_{ij}+\cdots$$
(D.38)

To finalise the argument we note that all the terms which are contained in \cdots will annihilate the amplitude. The term with W_i will annihilate the amplitude by definition of the Ward identity operator. The term with Δ_i will annihilate the amplitude due to the multi-linearity of the amplitude in polarisations. This argument can be extended to any product of L_i operators; by keeping track of all the commutation relations it can be shown that the four-particle longitudinal operator can be written as

$$L_i \cdot L_j \cdot L_k \cdot L_l = L_{ij} \cdot L_{kl} + L_{ik} \cdot L_{jl} + L_{il} \cdot L_{jk} + \cdots, \qquad (D.39)$$

where it becomes visible that the product of L_i will result in a sum of products of L_{ij} where i, j are grouped into distinct pairs.

D.5 Transmutation of the subleading order soft factors

YMS and BAS subleading soft factor from Yang-Mills

Since the soft factor is a pertubative expansion, we can apply transmutation to the subleading factor aswell. The leadling+subleading factor is given by:

$$S_{textYM}^{(ijk)} = \frac{p_i e_j - p_j J_i e_j}{p_i p_j} - \frac{p_k e_j - p_j J_k e_j}{p_k p_j}$$
(D.40)

where we consider the soft particle j adjacent to i and k. For this discussion we only consider the subleading factor denoted by

$$S_{\text{YM},sl}^{(ijk)} = \frac{p_j J_k e_j}{p_k p_j} - \frac{p_j J_i e_j}{p_i p_j} = -\frac{p_j J_i e_j}{p_i p_j} + (i \leftrightarrow k),$$
(D.41)

where the notation $p_j J_i e_j = p_j^{\mu} (J_i)_{\mu\nu} e_j^{\nu}$ denotes the summation over the Lorentz indices. In fact, J_i denotes the total angular momentum of the particle *i*, which can be split into the orbital and spin contribution:

$$(J_i)_{\mu\nu} = p_{i[\mu} \frac{\partial}{\partial p_i^{\nu]}} + e_{i[\mu} \frac{\partial}{\partial e_i^{\nu]}} = \underbrace{\left(p_{i\mu} \frac{\partial}{\partial p_i^{\nu}} - p_{i\nu} \frac{\partial}{\partial p_i^{\mu}}\right)}_{L_{\mu\nu}} + \underbrace{\left(e_{i\mu} \frac{\partial}{\partial e_i^{\nu}} - e_{i\nu} \frac{\partial}{\partial e_i^{\mu}}\right)}_{\Sigma_{\mu\nu}} \tag{D.42}$$

Here it becomes clear that the angular momentum operator is in fact a differential operator with respect to p_i and e_i , which allows for nontrivial commutation with the transmutation operators.

$$(J_i)_{\mu\nu} = [p_{i\mu}(\partial_{p_i})_{\nu} - p_{i\nu}(\partial_{p_i})_{\mu}] + [e_{i\mu}(\partial_{e_i})_{\nu} - e_{i\nu}(\partial_{e_i})_{\mu}]$$
(D.43)

The soft factor for YMS theory can again be deduced from the commutation with the trace operator

$$S_{\rm YMS,sl}^{(ijk|jl)} = \left[T_{jl}, S_{\rm YM,sl}^{(ijk)}\right] = -\sum_{m \neq i,k} \delta_{lm} I_{ilk} + \left\{\frac{N_{ijkl}}{p_i p_j} - (i \leftrightarrow k)\right\},\tag{D.44}$$

with $N_{ijkl} = \sum_{m \neq i} (\delta_{il} p_j e_m - \delta_{lm} p_j e_i) T_{im} + \sum_{m \neq i,k} \delta_{il} p_j p_m I_{mik}$. The commutator with the insertion operator results in the subleading soft factor for BAS theory

$$S_{\text{BAS,sl}}^{(ijk|ljm)} = \left[T_{ljm}, S_{\text{YM,sl}}^{(ijk)}\right] = \left\{\frac{(\delta_{li} - \delta_{im})(1 + p_j\partial_{p_i})}{p_i p_j} - \frac{p_j e_i}{p_i p_j}I_{lim} - (i \leftrightarrow k)\right\} - T_{ilkm}.$$
(D.45)

NLSM subleading soft factor

In the argumentation of (3.72) we discussed how we could use that effectively $[T^{(n-1)}, S^{(j)}] = 0$, as the leading order only involved derivatives with respect to pairs which do not contain soft leg *j*. For the subleading order, the soft factor does contain terms proportional to adjacent legs. Explicitly the factor is given by

$$S_{YM,sl}^{(ijk)} = \frac{(p_j p_i)(\partial_{p_i} e_j) - (p_j p_i)(\partial_{p_i} e_j) + (p_j e_i)(\partial_{e_i} e_j) - (p_j e_i)(\partial_{e_i} e_j)}{p_i p_j} + (i \leftrightarrow k), \quad (D.46)$$

which contains derivatives with respect to the soft leg. When considering the commutator with the longitudinal factor $L_{ij} = -p_i p_j \partial_{e_i e_j}$, the argumentation therefore has to be altered to incorporate this. First, consider the transmutation of the non-soft legs $T^{(n-1)} \propto (-p_l p_m \partial_{e_l e_m})$ with $l, m \neq j$. If $T^{(n-1)}$ acts on the Lorentz invariants (pp), (pe) or $(e_m e_j)$ it results in a vanishing contribution

$$T^{(n-1)} \cdot S^{(ijk)}_{\rm YM,sl} = 0$$
 (D.47)

Part of the commutator therefore already vanishes, the other part $S_{YM,sl}^{(ijk)} \cdot T^{(n-1)}$ contains several derivative terms. Derivatives of ∂_{e_i} will vanish, derivatives in the term $\partial_{p_i}(p_l p_m \partial_{e_l e_m}) e_j$ will only be nonzero if l, m = i so only $p_m e_j \partial_{e_l e_m}$ and $p_l e_j \partial_{e_l e_i}$ survive. The only surviving terms from $\partial_{p_k}(p_l p_m \partial_{e_l e_m})$ are those with l, m = k, which are $p_m e_j \partial_{e_k e_m}$ and $p_l e_j \partial_{e_l e_k}$. What remains of this contribution to the commutator is

$$S_{YM,sl}^{(ijk)} \cdot T^{(n-1)} = \frac{(p_j p_i)(p_m e_j)\partial_{e_i e_m} - (p_j p_i)(p_m e_j)\partial_{e_i e_m}}{p_i p_j},$$
 (D.48)

which will be hit by another longitudinal operator $L \sim \partial_{(ee)}$, resulting in an effectively vanishing contribution to the commutator as it does not contain any pairs of (ee). Therefore

it is shown that $[T^{(n-1)}, S^{(j)}]$ effectively vanishes for products of L_{ij} operators, which is neccessary to obtain amplitudes for the NLSM. Therefore, to obtain the subleading soft factor of the NLSM, the commutator with the Trace operator can be considered instead, due to the fact that $L_{ij} = -p_i p_j T_{ij}$:

$$S_{\text{NLSM,sl}}^{(ijk)} = \left[-p_j p_l T_{jl}, S_{YM}^{(ijk)} \right]$$
(D.49)

$$= \sum_{m \neq i,k} p_j p_l \delta_{lm} I_{ilk} - p_j p_l \left\{ \frac{N_{ijkl}}{p_i p_j} - (i \leftrightarrow k) \right\}$$
(D.50)

$$S_{\text{NLSM,sl}}^{(ijk)} = \sum_{l \neq i,k} p_j p_l I_{ilk}.$$
 (D.51)

It is noted that this subleading soft factor contains an insertion operator, indicating that soft behaviour is dependent on a lower point BAS amplitude, which is in agreement with the proposal of Cachazo and collaborators in the same publication that argued on the mixed amplitudes of pions and biadjoint scalars [35].

Appendix E

Mathematica computations of amplitudes

E.1 6-point YMS amplitude

At $\mathcal{O}(\tau^0)$ we encounter the YMS amplitude that is generated by summing up the contributions of the GT-operator for different arrangements of 6 particles. An example of one such an ordering is given by the ordering (12)(34)(56) which results in the contribution :

$$4A_{\rm YMS}(\phi_{1}\phi_{2},\phi_{3}\phi_{4},\phi_{5}\phi_{6}) = \frac{s_{2,3}}{s_{1,2}s_{1,2,3}} + \frac{s_{4,5}s_{2,3}}{s_{1,2}s_{1,2,3}s_{1,2,3}s_{1,2,3,4}} + \frac{1}{s_{1,2,3}} + \frac{-s_{2,4}s_{3,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{1,2}s_{3,4}s_{3,4,5}} \\ + \frac{s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{1,2}s_{4,5}s_{3,4,5}} + \frac{-s_{1,4}s_{2,5} - s_{2,4}s_{2,5} - s_{2,4}s_{3,5} + s_{2,3}s_{4,5}}{s_{1,2}s_{3,4}s_{1,2,3,4}} + \frac{s_{4,5}}{s_{1,2}s_{3,4}s_{1,2,3,4}} \\ + \frac{-s_{2,4}s_{2,5} - s_{2,4}s_{3,5} + s_{2,3}s_{4,5}}{s_{3,4}s_{2,3,4}s_{1,2,3,4}} + \frac{1}{s_{2,3,4,5}} + \frac{s_{4,5}}{s_{2,3,4}s_{2,3,4,5}} + \frac{-s_{2,4}s_{2,5} - s_{2,4}s_{3,5} + s_{2,3}s_{4,5}}{s_{3,4}s_{2,3,4}s_{1,2,3,4}} \\ + \frac{-s_{2,4}s_{3,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{3,4}s_{2,3,4,5}} + \frac{s_{4,5}}{s_{4,5}s_{2,3,4,5}} + \frac{s_{4,5}s_{4,5} + s_{2,5}s_{4,5}}{s_{4,5}s_{4,5}s_{2,3,4,5}} \\ + \frac{-s_{2,4}s_{3,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{3,4}s_{3,4,5}s_{2,3,4,5}} + \frac{s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{4,5}s_{4,5}s_{2,3,4,5}} \\ + \frac{-s_{2,4}s_{3,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{3,4}s_{3,4,5}s_{2,3,4,5}} + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}s_{4,5}s_{4,5}} \\ + \frac{-s_{2,4}s_{3,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,5}}{s_{4,5}s_{4,5}s_{2,3,4,5}} + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}s_{4,5}s_{4,5}} \\ + \frac{-s_{2,4}s_{3,5} + s_{2,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}s_{4,5}s_{4,5}s_{4,5}} \\ + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}s_{4,5}} + \frac{s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}s_{4,5}s_{4,5}} \\ + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}} + \frac{s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}s_{4,5}} \\ + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}} + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}s_{4,5}} \\ + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}} + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}} \\ + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}} + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}} \\ + \frac{s_{4,5}s_{4,5} + s_{4,5}s_{4,5}}{s_{4,5}s_{4,5}} + \frac{s_{4,5}s_{4,5}}{$$

with a factor 1/4 due to the replacement of the momentum factor $p_i p_j$ that was present in numerator of the gluon amplitude to $p_i p_j \rightarrow s_{ij}/2$.

The complete amplitude is the summation over all 15 arrangements and is given by

$$\begin{split} A_{\rm YMS} & \frac{s_{1,5}s_{2,3} + s_{2,5}s_{2,3} + s_{3,5}s_{2,3} + \frac{-s_{1,5}s_{2,3} - s_{2,5}s_{2,3} - s_{3,5}s_{2,3} + \frac{-s_{1,5}s_{2,3} - s_{2,5}s_{2,3} - s_{3,5}s_{2,3} + \frac{s_{2,3}s_{4,5}s_{1,2,3} + s_{2,3}s_{4,5}s_{1,2,3} + s_{2,3}s_{4,5}s_{1,2,3} + s_{2,3}s_{4,5}s_{1,2,3} + s_{2,3}s_{4,5}s_{1,2,3} + \frac{s_{2,3}s_{4,5}s_{1,2,3} + s_{2,3}s_{4,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,4} + s_{2,3}s_{4,5} + s_{2,3}s_{4,5} + s_{2,3}s_{4,5} + s_{2,5}s_{4,4} + s_{2,5}s_{3,4} - s_{2,3}s_{4,5} - s_{2,4}s_{2,5} + s_{2,4}s_{3,5} + s_{2,4}s_{4,5} + \frac{s_{2,3}s_{4,5} + s_{2,5}s_{3,4} + s_{2,5}s_{3,4} + s_{2,5}s_{3,4} + s_{2,5}s_{3,4} + s_{2,5}s_{3,4} + s_{2,5}s_{3,4} + s_{4,5}s_{3,4} + s_{2,5}s_{3,4} + s_{4,5}s_{3,4} + s_{2,5}s_{3,4} + s_{4,5}s_{3,4} + s_{2,5}s_{3,4} + s_{4,5}s_{3,4} + \frac{s_{2,5}s_{3,4} + s_{4,5}s_{3,4} + s_{4,5}s_{3,4} + s_{2,5}s_{3,4} + s_{4,5}s_{3,4} + s_{4,5}s_{4,5} + s_{4,$$

From a quick inspection it is clear that there are quite a few terms that share the same propagator which could be collected. It is chosen to not simplify this calculation at this stage for two reasons. The first being that the *Simplify* function of mathematica will collect all terms and simplify them to 1 term with a new denominator that is common to all elements of the sum. This does not give us any meaningful insight. The second reason is that we will expand on this result in later computations, therefore it is not yet necessary to simplify at this stage.